Recent Developments in Chemistry and Biochemistry Research

Vol. 3

Edited by Dr. Osunsanmi Foluso Oluwagbemiga







Recent Developments in Chemistry and Biochemistry Research Vol. 3

India . United Kingdom

B P International

1.001.0007

Contents

About The Editor	í.
Preface	
Chapter 1	1-21
Chemistry of Depsides from Manglicolous lichen Dirinaria consimilis (Stirton) D.D. Awasthi along with their Pharmacological Insights Vinay Bharadwaj Tatipamula and Girija Sastry Vedula	
Chapter 2	22.28
Optimizing Bio-Oil Yield: A Study on Trichosanthes Tricuspidata Seed Extraction	22-30
V. Mansa Devi, M. Dharmendira Kumar, R. Sridhar and C. Jaya Kumar	
Chapter 3	39-55
Kinetic Profiling of Phenobarbital Derivative Inhibition Levels on Serum Acetylthiocholine Esterase Activity N. Zaizafoon	
Chapter 4	56-73
Evaluation of Antimicrobial Efficiency of Bipyrimidines Using	
Zeolite as a Green Catalyst in Biphasic System	
Manisha S. Aswale and Raksha P. Dhankar	
Chapter 5	74-104
Production of Polyhydroxyalkanoates (PHAs) Using Raw Sewage and Cheese Whey	
Young-Cheol Chang	
Chapter 6	105-119
Preparation of Novel Dihydrophthalazinyl Phenylamide Derivatives and their Anti-Microbial Evaluation S. N. Murthy Boddapati, Jaganmohana Rao Saketi, A. Emmanuel Kola and Giri Prasad Gorumutchu	
Chapter 7	120-146
Determination of FAM72 Family Proteins as Poor Prognostic Markers in Clear Cell Renal Carcinoma	
the over the state and the state of the	147.16
Chapter 8 Toxicity of Organophosphorus Flame Retardants: Implications in Environmental Persistence Bob A. Howell	147-154

Post ISBN 978-81-973053-4-4, eBook ISBN 978-81-979067 4 Evaluation of Antimicrobial Efficiency of Bipyrimidines Using Zeolite as a Green

Manisha S. Aswale * and Raksha P. Dhankar *

DOI: 10.9734/bpi/rdcbr/v3/8227E

Peer Review History: This chapter was reviewed by following the Advanced Open Peer Review policy. This chapter was thoroughly theory. This chapter was reviewed by following the Advanced Open Peer Review policy. This chapter was thoroughly theory. Peer Review History: This chapter was reviewed by following the Advanced Open Heer Herview poech, in the chapter was thoroughly the book of the peer reviewers reviewed the manuscript of the peer review of the manuscript of final publication. Peer review of the manuscript the Book Editor approved the manuscript for final publication. Peer review of the manuscript the Book Editor approved the manuscript for final publication. Peer review of the manuscript the Book Editor approved the manuscript for final publication. Peer review of the manuscript the Book Editor approved the manuscript for final publication. Peer review of the manuscript the Book Editor approved the manuscript for final publication. Peer review of the manuscript the Book Editor approved the manuscript for final publication. This chapter was reviewed detorial policy, a minimum or two peer reviewed internance the menuacing and the prevent pre

ABSTRACT

In the present study, we have designed and synthesized a new series of In the present study, we have a screened them for their series of bipyrimidines by simple condensation and screened them for their in vito bipyrimidines by simple conserver has depicted the efforts towards the antimicrobial activities. Current research has depicted the efforts towards the antimicrobial activities. Guinant tems in the synthesis of a variety of heterocycles, utilization of biphasic reaction systems in the synthesis of a variety of heterocycles. utilization of biphasic reaction systemace of the monophasic solvent system. A This shows the remainance english condensation process has been developed to system. A simple, green and efficient catalytic condensation process has been developed to a simple of 2-amino-6-substituted developed to 2-amino-6-substituted- 4.6-diphenyl-3:4.4 5 tetrahydro[4,5'-bipyrimidine]-2'(1H)-one (3a- j) hybrids. The catalytic route was tetrahydro[4,5-bipyniniumere the presence of NaY zeolite in an organic-aqueous investigated efficiently in the presence of NaY zeolite in an organic-aqueous (dichloromethane-water) solvent system. In this method, biphasic solvent systems were explored for suitable applicability where catalyst exhibits remarkable reactivity. The synthesized scaffolds of bipyrimidines were studied as antimicrobial agents. The investigation of antimicrobial screening data revealed that among 10 compounds screened, compounds 3d, 3e, 3f and 3i demonstrated very good activity as compared to standard drugs and the remaining compounds showed good to moderate inhibition activities.

Synthesized compounds were screened for their in vitro antibacterial activity against Staphylococcus aureus, E. coli and Pseudomonas aurgenosa and also antifungal activity against the opportunistic pathogens Candida albicans and Aspergillus niger.

Keywords: Bipyrimidines; zeolite; dichloromethane-water biphasic system. antimicrobial activity.

^{*} Department of Chemistry, Centre for Higher Learning and Research, Sardar Patel Muhavidralevi Ganjward, Chandrapur-442402, India.

[&]quot;Corresponding author: E-mail: rakshadharikar@rediffmail.com.

Recent Developments in Chemistry and Biochemistry Research Vol. 3 Evaluation of Antimicrobial Efficiency of Bipyrimidines Using Zeolite as a Green Catalyst in Biphasic System

1. INTRODUCTION

Green synthesis of zeolites can be classified into four main categories: (1) synthetic methods that do not require a template or use a recyclable, inexpensive, or renewable template; (2) synthesis of zeolites that use sustainable silicon or aluminum sources; (3) solvent-free methods: and (4) facile synthesis methods, e.g., microwave heating to shorten the crystallization time, solid phase (or quasisolid phase) synthesis to promote product yield, or continuous-flow synthesis to achieve high productive efficiency [1,2]. Bipyrimidines hybrids play an essential role in several biological processes and have considerable chemical and pharmacological importance [3]. A large number of bipyrimidine derivatives are reported to be antimicrobial [4], anticancer [5], antiviral [6], anti-inflammatory [7], antifungal [8], analgesic [9], anticonvulsant [10], antioxidant [11], antitubercular, antimalarial [12] and antileishmanial [13].

Zeolite X is a highly versatile molecular sieve with a pore size of 7.4 A from the faujasite family of zeolites whose, three-dimensional pore structure and solid acidity make it useful as a catalyst in the synthesis of analogues of several *N*neterocycles[14,15,3]. Because of their unique properties, which include controlled variability, thermal stability, reusability, and environmental friendliness, zeolites are the most important catalysts in green chemistry. The efforts to use a biphasic reaction system in the synthesis of various heterocycles have been documented in current research. This shows the remarkable importance of the monophasic solvent system. The biphasic solvent system allows easy separation and reusability of the reactive aqueous phase containingspent homogeneous or heterogeneous catalysts [3]. The recent work from various research groups indicated that biphasic reaction systems showed significant advantages in protecting the products from further degradation by extracting the products produced from the monophasic solvent, simplifying the separation steps to achieve the final products, minimizing the side reactions and increasing the overall yield [16,17,3].

We opted for an approach for the development of an easy, efficient, green and clean method for the synthesis of 2-amino- 6-substituted-4,6-diphenyl-3',4,4',5tetrahydro[4,5'-bipyri- midine]-2' (1H)-one analogues (3a-j). The rarity of reports in the catalytic synthesis of bipyrimidines in a biphasic solvent system, the work aims to study the beneficial approach in yields of bipyrimidines in DCM-water optimized phase using zeolite as a catalyst [3]. In the present study, we have designed and synthesized a new series of bipyrimidines by simple condensation and screened them for their in vitro antimicrobial activities where 3d, 3e, 3f and 3i were found to exhibit excellent potent activity as antibacterial and antifungal agents. Other compounds possessed moderate to low activity.

2. EXPERIMENTAL

Bipyrimidines were synthesized by using analytical grade substituted chalcone and guanidine hydrochloride (S.D. Fine Chemicals, 98 %). Zeolite, dichloromethane, ethyl acetate and n-hexane were obtained from Qualigen India Ltd. Mumbai.

Recent Developments in Chemisky and Diochemisky Research 25, Second Developments in Chemisky and Diobas Catalysis (1997) Recent Developments in Course Jointe as a Green Callery Research view of Antonio Callery Strengton Branch view Callery Strengton Branch view Strengton Bra

Melting points were determined by open capillary method and are uncorrected As Melting points distilled and dried prior to use [3]. TLC was performed on titles As Melting points were determined by open causion, TLC was performed uncorrected solvents were distilled and dried prior to use [3]. TLC was performed on tile of solvents were exposed to iodine vapours for visualization. A mixture get G and the spots were exposed to iodine vapours for visualization. A mixture of G and the spots were exposed to iodine vapours for visualization. A mixture of and 13C March 13 G and the spots were exposed to ionine value as an eluent. 1H NMR and 13C of n. hexane and ethyl acetate (7.3) was used as an eluent. 1H NMR and 13C of n. hexane and ethyl acetate (7.3) on a Brucker AC 400 (MHz) instrument, Ch. Mar. becare and ethyl acetate (7.3) was used as AC 400 (MHz) instrument and 13C M/lis spectra were recorded in CDCIs on a Brucker AC 400 (MHz) instrument, Chemical spectra were recorded in com using TMS as the internal standard, IR spectra incar spectra were recorded in CDCIs on a brocket internal standard. IR spectra were shifts are reported in ppm using TMS as the internal standard. IR spectra were shifts are reported in ppm using TMS as the internal standard were using KBr discs and were spectra were reported in ppm using TMO as the transfer using KBr discs and were obtained on a Perkin Elmer 1800 spectrophotometer using KBr discs and mass obtained on a Perkin Elmer 1800 spectra gas chromatograph coupled with QP5050 spectra were measured with Shimadzu gas chromatograph coupled with QP5050 spectra were measured with Shimauco get analysis was carried out with Qp5050 Spectrometer at 1-1.5 eV. CHN elemental analysis was carried out with Perkin

In vitro Antibacterial and antifungal activities: All the series of synthesized In vitro Antibacterial and antiturigation of their efficacy against the clinically synthesized compounds were evaluated for their efficacy against the clinically isolated isolated for their efficacy against the clinically isolated solution. compounds were evaluated for them (ATCC 25922). Pseudomonas aeruginonas aeruginonaeruginonaeruginonas aeruginonas aeruginonas (ATCC 85327) (Gram-negative bacteria) Staphylococcus aureus (ATCC 29213) (ATCC 85327) and Aspendida albicans (ATCC 102310) and Aspendit (ATCC 85327) (Gram-negative bacteria), Candida albicans (ATCC 102310) and Aspergillus night (Gram-positive bacteria), Candida albicans (ATCC 102310) and Aspergillus night (Gram-positive bacteria), Canoua autorial activities of the compounds night (ATCC 439). The preliminary antimicrobial activities of the compounds 3a-j were (ATCC 439). The presiminary antition [18, 19,3]. The compounds to be screened tested using the cup-plates method [18, 19,3]. The compounds to be screened tested using the cup-plates method pro-trations viz. 12.5, 25, 50 and 100µg/mL were dissolved in DMSO at different concentrations viz. 12.5, 25, 50 and 100µg/mL were dissolved in DMSO at different 24 h, the control was similarly maintained with The plates were incubated at 37 °C for 24 h, the control was similarly maintained with The plates were incubated at or of the bitton of bacterial and fungal growth were 1 mL of DMSO and the zones of inhibition of bacterial and fungal growth were 1 mL of DMSO and the zoneson to conazole were used as the standarddrugs measured inmm. Ampicillin and ketoconazole were used as the standarddrugs. The inoculated plates were incubated at 37 °C for 24 hin the case of bacteria and The inoculated plates were incubated at 37 °C for 24 hin the case of bacteria and The inoculated plates were incounted inhibition was compared with the standard 48 h in the case of fungus. The zoneof inhibition was compared with the standard

The minimum inhibitory concentrations (MIC) of compounds were tested using the microdilution susceptibility method [16]. The chemical stock solutions of all the compounds and reference drugs were prepared by dissolving 1000 µg in 5 mL DMSO. A series of dilutions were prepared as 100, 50, 25 and 12.5 µg/mL. The solutions with no turbidity were considered as MIC for tested compounds [3].

General procedure for the synthesis of 5-cinnamoyl-6-methyl- 4-phenyl-3,4dihydropyrimidin-2(1H)-one (2a): The synthesis of chalcones was carried out via Claisen-Schmidt condensation. A mixture of 5-acetyl-6-methyl-4-phenyl-3,4dihydro- pyrimidine-2(1H)-one (1 mmol, 0.214 g) and benzaldehyde (1 mmol, 0.106 g) was dissolved in 10 mL of ethanol in 250 mL round bottom equipped with a magnetic stirrer. Then 20 mL NaOH solution (8 g in 20 mL H2O) was added dropwiseto the reaction mixture with vigorous stirring for 0.5 h at roomtemperature. The reaction mixture was kept overnight. The reaction mixture was neutralized by adding dilute HCI wherebythe precipitation occurred. The product was filtered and recrystallized by ethanol. The physicochemical analysis of the synthesized compounds is shown in Table 1 [3].

General procedure for the synthesis of 2-amino-6-methyl-4,6-diphenyl-3',4,4',5tetrahydro-[4,5'-bipyrimidine]- 2' (1H)-one (3a): In a 50 mL of round bottom flask 5cinnamoyl-6-methyl-4-phenyl-3,4-dihydropyrimidin-2(1H)-one (2a) (5 mmol, 1.65g) and zeolite (30 mol %, 0.573 g) were thoroughlymixed in dichloromethane and were stirred for few minutes. Guanidine hydrochloride (10 mmol, 0.95 g) dissolved in water

Recent Developments in Chemistry and Blochemistry Research Vol. 3 Evaluation of Antimicrobial Efficiency of Bayrimidmes Using Zeolits as a Green Caralystic Biohamic System

was poured into a round bottom flask of reaction mixture. The solution was now stimed in organic-water phases as reaction media at 60 °C for 0.5 h (Scheme-I) [3]. The extent of the reaction was monitored by TLC. After the completion of the reaction, the product was extracted from biphasic solvents as a solid material by filtration. The solvents were separated by a separating funnel and water was evaporated to get the zeolite for the next run. The product (3a) obtained was subjected to recrystallization by ethanol. The physico-chemical analysis of the synthesized compounds are shown in Table 2 [3].

Compd.	Rı	R1	Yield(%)	Time (min)	m.p. ("C)
2a	-H	-H	83	90	160
2b	-H	-NO2	79	120	215
20	-H	-OCH3	85	105	220
2d	-OCH ₃	-OCH ₃	80	110	200
2e	-CI	-H	82	105	173
21	-H	-CI	82	105	168
20	-NO2	-H	85	120	210
2h	-OCH3	-NO2	86	115	220
21	-CI	-OCH ₃	87	135	208
21	-NO2	-OCH3	84	130	217

Table 1. Synthesized of Substituted 5-CinnamoyI-6-MethyI-4-PhenyI-3,4 Dihydropyrimidin-2(1H)-One (2a-j)



Scheme-L



			1000		and the second se	
Compd.	R1	Ra	Ra	Yield(%)	Time(min)	m.p.(*C)
3a	-H	-H	-H	89	30	198
36	-H	-H	-NO2	86	34	195
3c	-H	-H	-OCH ₃	85	36	190
3d	-OCH ₃	-H	-OCH3	84	30	222
30	-CI	-H	-H	87	45	202
31	-H	-H	-CI	83	34	188
30	-NO2	-H	-H	87	48	201
3h	-OCH3	-H	-NO2	86	56	265
3i	-CI	-H	-OCH3	85	47	226
3	-NO2	-H	-OCHs	86	45	233

Recent Developments in Chemistry and Blochemistry Research Ver a Compact of Reportations Using Zeolite as a Gran Constant in Super-Recent Developments in Chemistry and Discontinistry Research Use Evaluation of Antimicrobial Efficiency of Bayromidines Lising Zeolite as a Grean Catalyst in B

[4.5 bipyrimidine]. 2 (1H)-one (3a) [3]: Yellow-colored solid, (400 MHz, CDCl₂, ö ppm); 6.9-7 8 (KBr, vel. cm⁻¹): 3286, 1612,1454, 1074, ¹H NMR (400 MHz, CDCl₂, ö ppm); 6.9-7 8 (m.10H 2.0 (s. 3H, -NH), 8.7 (s. 1H, -NH), 3.9 (s. 2H, -NH₂), 2.4 (s. 3H, -10H cm⁻¹): 3286, 1612, 1454, 1074, ⁻H NMR (400 H, 3.9 (s. 2H, -NHs), 2.4 (s. 3H, 10H, arom.), 8.6 (s. 1H, -NH), 8.7 (s. 1H, -NH), 3.9 (s. 2H, -NHs), 2.4 (s. 3H, 10H, arom.), 8.6 (s. 1H, -NH), 8.7 (s. 1H, -CH₂), 3.32-3.34 (t. 1H, -CH), 5.6 (s. 2H, -CH₂), 5.6 (s. 2 arom.), 8.6 (s. 1H, -NH), 8.7 (s. 1H, -CH₂), 3.32-3.34 (l. 1H, -CH), 5.6 (s. 3H, -CH₂), 2.68-2.70 (d. J = 10.68 Hz, 2H, -CH₂), 3.32-3.34 (l. 1H, -CH), 5.6 (s. 2H, -CH₂), 2.68-2.70 (d. J = 10.68 Hz, 2H, -CH₂), 3.32-3.34 (l. 1H, 4, 120.9, 127.1, 128.7 Hz), 2.68-2.70 (d. J = 10.68 Hz, 2H, -CH₂), 3.32-3.34 (l. 1H, -CH), 5.6 (s. 2H, -CH₂), 3.32-3.34 (l. 1H, 4, 120.9, 127.1, 128.7 Hz), 5.6 (s. 2H, -CH₂), 3.32-3.34 (l. 1H, 4, 120.9, 127.1, 128.7 Hz), 5.6 (s. 2H, -CH₂), 3.32-3.34 (l. 1H, 4, 120.9, 127.1, 128.7 Hz), 5.6 (s. 2H, -CH₂), 5.6 (s. 2H, ¹⁰C NMR (CDCb, 5 ppm): 15.2, 35.6, 46 (70 eV): m/z =359.17 [M*], Anal Calci 139.9, 141.8, 148.5, 153.6, 163.7, MS (70 eV): m/z =359.17 [M*], Anal Calci 139.9, 141.8, 148.5, 153.6, 163.7, MS (70 eV): m/z =359.17 [M*], Anal Calci Calci 139.9, 141.8, 149.5, 153.6, 103.7, 103.7, 103.7, 103.1, 10

2-Amino-6-methyl-6-(4-nitrophenyl-4-phenyl-3',4,4',5-tetrahydro[4,5',

2-Amino-6-methyl-6-(4-nitroprietry, Light yellow coloured solid, yield, 86 %, m.p. 195 bipyrimidine]-2 (1H)-one (3b) [3]: Light yellow coloured solid, yield, 86 %, m.p. 195 bipyrimidine]-2 (1H)-one (30) (31, -53, 1072, 'H NMR (400 MHz, CDCI), 5, m.p. 195 °C, IR (KBr, vmst, cmr'): 3285, 1611, 1450, 1072, 'H NMR (400 MHz, CDCI), 5, m.p. 195 (8, 2H, -NH), 8, 3 (s, 1H, -NH), 3, 9 (s, 2H, -NH), 5, 0 (s, 2H, -NH), 5, *C, IR (KBr, v_{max}, cm⁻¹): 3285, 1611, 1406, 83 (s. 1H, -NH), 3.9 (s. 2H, -NH₂), 6 ppm), 6.8-7.2 (m, 9H, arom.), 8.1 (s. 1H, -NH), 8.3 (s. 1H, -NH), 3.9 (s. 2H, -NH₂), 2.3 (s. 6.8-7.2 (m, 9H, arom.), 8.1 (s. 1H, -CH₂), 3.32-3.34 (t. 1H, -CH), 5.6 (s. 2H, -NH₂), 2.3 (s. 2H, -NH₂), 3.32-3.34 (t. 1H, -CH), 5.6 (s. 2H, -NH₂), 2.3 (s. 2H, -NH₂), 3.32-3.34 (s. 2H, 6.8-7.2 (m, 9H, arom.), 5.1 (s, 4H, 2H, -CH₂), 3.32-3.34 (t, 1H, -CH), 2.3 (s, 3H, -CH₀), 2.58-2.60 (d, J = 10.68 Hz, 2H, -CH₂), 3.32-3.34 (t, 1H, -CH), 5.6 (s, 2H, 3H, -CH₀), 2.58-2.60 (d, J = 10.68 Hz, 2H, -CH₂), 3.32-3.34 (t, 1H, -CH), 5.6 (s, 2H, 3H, -CH₀), 2.58-2.60 (d, J = 10.68 Hz, 2H, -CH₂), 3.32-3.34 (t, 1H, -CH), 5.6 (s, 2H, 3H, -CH₀), 2.58-2.60 (d, J = 10.68 Hz, 2H, -CH₂), 3.32-3.34 (t, 1H, -CH), 5.6 (s, 2H, 3H, -CH₀), 2.58-2.60 (d, J = 10.68 Hz, 2H, -CH₂), 3.32-3.34 (t, 1H, -CH), 5.6 (s, 2H, 3H, -CH₀), 2.58-2.60 (d, J = 10.68 Hz, 2H, -CH₂), 3.32-3.34 (t, 1H, -CH), 5.6 (s, 2H, 3H, -CH₀), 2.58-2.60 (d, J = 10.68 Hz, 2H, -CH₂), 3.32-3.34 (t, 1H, -CH), 5.6 (s, 2H, 3H, -CH₀), 2.58-2.60 (d, J = 10.68 Hz, 2H, -CH₂), 3.32-3.34 (t, 1H, -CH), 5.6 (s, 2H, 2H, -CH₀), 2.58-2.60 (d, J = 10.68 Hz, 2H, -CH₂), 3.32-3.34 (t, 1H, -CH), 5.6 (s, 2H, 2H, -CH₀), 2.58-2.60 (d, J = 10.68 Hz, 2H, -CH₂), 3.32-3.34 (t, 1H, -CH), 5.6 (s, 2H, 2H, -CH₀), 3.32-3.34 (t, 1H, -CH), 5.6 (s, 2H, 2H, -CH₀), 3.32-3.34 (t, 1H, -CH), 5.6 (s, 2H, 2H, -CH₀), 3.32-3.34 (t, 1H, -CH), 5.6 (s, 2H, 2H, -CH₀), 3.32-3.34 (t, 1H, -CH), 5.6 (s, 2H, 2H, -CH₀), 3.32-3.34 (t, 1H, -CH), 5.6 (s, 2H, 2H, -CH₀), 3.32-3.34 (t, 1H, -CH), 5.6 (s, 2H, 2H, -CH₀), 3.32-3.34 (t, 1H, -CH), 5.6 (s, 2H, 2H, -CH), 5.6 (s, 3H, -CH₃), 2.58-2.60 (d, J = 10.00112, 24 -CH). ¹⁵C NMR (CDCI₃, 8 ppm): 14.6, 35.0, 40.3, 56.4, 58.7, 115.1, 123.2, -CH). ¹⁵C NMR (CDCI₃, 8 145.5, 149.5, 154.3, 164.7, MS (70 eV). -CH). ¹⁵C NMR (CDCb, e ppin, 141.5, 149.5, 154.3, 164.7, MS (70 eV), 123.2, 125.2, 127.9, 128.8, 141.7, 141.8, 145.5, 149.5, 154.3, 164.7, MS (70 eV), 123.2, 125.2, 127.9, 128.8, 141.7, 141.8, 145.5, 149.5, 154.3, 164.7, MS (70 eV), 123.2, 125.2, 127.9, 128.8, 141.7, 141.8, 145.5, 149.5, 154.3, 164.7, MS (70 eV), 123.2, 125.2, 127.9, 128.8, 141.7, 141.8, 145.5, 149.5, 154.3, 164.7, MS (70 eV), 123.2, 125.2, 127.9, 128.8, 141.7, 141.8, 145.5, 149.5, 154.3, 164.7, MS (70 eV), 123.2, 125.2, 127.9, 128.8, 141.7, 141.8, 145.5, 149.5, 154.3, 164.7, MS (70 eV), 123.2, 125.2, 127.9, 128.8, 141.7, 141.8, 145.5, 149.5, 154.3, 164.7, MS (70 eV), 123.2, 125.2, 127.9, 128.8, 141.7, 141.8, 145.5, 149.5, 154.3, 164.7, 164.7, 165.2, 127.2, 1 125.2, 127.9, 128.8, 141.7, 141.0, forC21H20NeO3, m.w. 404.14; C, 62.37 (63.65); m/2 = 404 [M*]. Anal. Calcd. (found) % forC21H20NeO3, m.w. 404.14; C, 62.37 (63.65);

2-Amino-6-(4-methoxyphenyl)-6-methyl-4-phenyl-3',4,4',5-tetrahydro(4,5',

2-Amino-6-(4-methoxypheny) [3]: Dark brown coloured solid, yield: 85 % m.p. 190 °C. IR (KBr, Omax. cm⁻¹): 3282, 1613, 1452, 1072, ¹H NMR (400 MHz, CDCl3, 6 ppm): 6.9-7,4 (m, 9H, arom.), 8.5 (s. 1H, -NH), 3.9 (s. 2H, - NH2), 2.5 (s. 3H, -CH3), 2.83-2.85 (d, J = 10.68 Hz, 2H, - CH2), 3.22-3.24 (t, 1H, -CH), 5.4 (s, 2H, -CH), 3.2 (s, 3H, - OCH3), 13C NMR (CDCl3, 8 ppm): 14.6, 35.1, 40.3, 56.7, 58.8, 113.1, 115.0, 124.7, 126.1, 128.5, 133.6, 141.3, 149.8, 152.3, 163.1, 165.1, MS (70 eV): m/z 389.19 [M*]. Anal. Calcd. (found) % for C22H23N5O2, m.w. 389.14: C, 67.85 (67.83); H, 5.95 (5.93); N, 17.98 (17.96); O, 8.22 (8.20).

2-Amino-4',6-bis(4-methoxyphenyl)-6-methyl-3',4,4',5-tetrahydro[4,5'-

bipyrimidine]-2'(1H)-one (3d) [3]: Yellow coloured solid, yield: 84 %, m.p. 222 °C. IR (KBr, vmax, cm⁻¹):3285, 1612, 1453, 1070. ¹H NMR (400 MHz, CDCl3, 8 ppm): 6.9-7.9 (m, 8H, arom.), 8.6 (s, 1H, -NH), 4.02 (s, 2H, -NH2), 2.4 (3H, -CH3), 2.50-2.53 (d, J = 10.68 Hz, 2H, -CH2), 3.32- 3.34 (t, 1H, -CH), 5.5 (s, 2H, -CH), 3.6 (s, 3H, -OCH3). 13CNMR (CDCI3, 6 ppm): 14.1, 35.1, 39.8, 55.9, 58.8, 114.2, 115.0, 124.6, 126.7, 128.9, 133.2, 139.9, 141.8, 149.5, 152.8, 159.9, 162.4, 164.1. MS (70 eV): m/z 419.19 [M*]. Anal. Calcd. (found) % for C23H25N5O3, m.w. 419.18: C. 65.85 (65.82); H.6.01 (6.01); N. 16.70 (16.68); 0, 11.44 (11.42).

2-Amino-4-(4-chlorophenyl)-6-methyl-6-phenyl-3',4,4',5-tetrahydro[4,5'bipyrimidine]-2'(1H)-one (3e) [3]: Pale yellow coloured solid, yield: 87 %, m.p.

202 °C. IR (KBr, vmax, cm⁻¹): 3281, 1610, 1452, 1073. ¹H NMR (400 MHz, CDCl3, 8 ppm): 7.3-7.9 (m, 9H, arom.), 8.6 (s, 1H, -NH), 3.9 (s, 2H, - NH2), 2.1 (s, 3H, -CH3).

Recent Developments in Chemistry and Biochemistry Research Vol. 3 Evaluation of Antimicrobial Efficiency of Bipyrimidines Using Zeolitz As a Green Cetalyst in Biphasic

System

2.60-2.63 (d, J = 10.68 Hz, 2H, -CH2), 3.42-3.44 (t, 1H, -CH), 5.6 (s, 2H, -CH), ¹⁷C NMR (CDCI3, 8 ppm): 15.2, 35.8, 40.9, 58.6, 114.4, 123.5, 127.1, 128.3, 129.8 131.3. 139.9. 141.8. 149.5. 153.6. 163.7. MS (70 eV) m/z = 393.14 [M*]. Anal. Calcd. (found) % for C21H20N5Ocl. m.w. 393.12; C, 64.04 (64.02); H, 5.12 (5.10); N. 17.75 (17.75); O.4.06 (4.03); Cl. 9.00 (9.90).

2-Amino-6-(4-chlorophenyl)-6-methyl-4-phenyl-3:,4,4:,5-tetrahydro[4,5-

bipyrimidine]-2 (1H)-one (3f) [3]: Light yellow coloured solid, yield: 83 %; m.p. 188 ec. IR (KBr, vmax, cm⁻¹): 3283, 1612, 1454, 1073. ¹H NMR (400 MHz, CDCI3, 5 ppm): 7.4-7.8 (m. 9H, arom.), 8.7 (s. 1H, -NH), 3.7(s. 2H, - NH2), 2.1 (s. 3H, -CH3), 2.40-2 43 (d, J = 10.68 Hz, 2H, -CH2), 3.32-3.34 (t, 1H, -CH), 5.7 (s, 2H, -CH), ¹³G NMR (CDCl3, 8 ppm): 15.2, 35.8, 40.9, 58.6, 114.4, 123.6, 127.1, 128.7, 131.0, 135.9, 137.6. 141.8. 149.5. 153.6. 163.7. MS (70 eV): m/z = 393.14 [M+] Anal. Calcd. (found) % for C21H20N5Ocl. m.w.393.14: C. 64.04 (64.02); H. 5.12 (5.10); N. 17.78 (17.75): O.4.06 (4.03): CI, 5.12 (5.10).

2-Amino-6-methyl-4-(4-nitrophenyl)-6-phenyl-3',4,4',5-tetrahydro[4,5'bipyrimidine]-2 (1H)-one (3g) [3]: Yellow coloured solid, yield: 87 %, m.p. 201 *C, IR (KBr, vmax, cm⁻¹): 3284,1610,1453,1072. 'H NMR (400 MHz, CDCI3, 8ppm); 7.5-7.9 (m.9H, arom.). 8.6 (s, 1H, -NH). 8.1 (s, 1H, -NH), 3.9 (s, 2H, -NH2), 2.2 (3H, -CH3), 2.47-2.50 (d, J = 10.68 Hz, 2H, -CH2), 3.34-3.32 (t, 1H, -CH), 5.6 (s, 2H, -CH). 13C NMR (75 MHz, CDCl3, 8ppm): 15.2, 35.8, 40.9, 58.6, 114.4, 123.6, 123.9. 127.1, 128.7, 131.0, 139.9, 146.2, 147.9, 149.5, 153.6, 163.7, MS (70 eV): m/z 404.16 [M*], Anal. Calcd. (found) % for C21H20N6O3, m.w. 404.16; C, 62.37 (62.35); H. 4.98 (4.97); N. 20.78 (20.75); O. 11.87 (11.85).

2.Amino-4-(4-methoxyphenyl)-6-methyl-6-(4-nitro-phenyl)-3',4,4',5-

tetrahydro [4,5'-bipyrimidine]-2'(1H)-one (3h) [3]: Dark yellow solid, yield: 86 %, m.p. 265 °C. IR (KBr, vmax, cm⁻¹): 3286, 1613, 1453, 1073. ¹H NMR (400 MHz, CDCl3. 5 ppm): 6.7-7.8 (m, 8H, arom.). 8.6 (s. 1H, -NH), 8.2 (s. 1H, - NH), 4.0 (s. 2H, -NH2), 2.2 (s. 3H, -CH3), 2.50-2.53 (d. J = 10.68 Hz, 2H, -CH2), 3.42-3.44 (t. 1H. -CH), 5.6 (s, 2H, -CH), 3.7 (s, 3H, -OCH3), 13C NMR (CDCI3, 8 ppm): 15.2, 35.8. 40.9. 58.6. 113.1. 114.4. 123.4. 125.5. 127.1. 128.7. 135.7. 146.9. 149.5. 153.6. 158.5. 163.7. MS (70 eV): m/z 434.45 [M*].

Anal. Calcd. (found) % for C22H22N6O4, 434.42; C, 60.87(60.85); H, 5.10 (4.98); N. 19.34 (19.32); O. 14.73 (14.70).

2-Amino-6-(4-chlorophenyl)-6-methyl-4'-(4-methoxy-phenyl)-6',4,4',

Stetrahydro[4,5'-bipyrimidine]-2'(1H)-one(3i) [3]: Yellow coloured solid, yield: 85 %, m.p. 226 °C, IR (KBr, vmax, cm⁻¹); 3286, 1614, 1453, 1074. ¹H NMR (400 MHz.CDCl3. 5 ppm): 7.0-7.9 (m, 8H, arom.), 8.6 (s, 1H, -NH), 8.0(s, 1H, -NH), 3.9 (s, 2H, -NH2), 2.3 (s, 3H, -CH3), 2.50-2.53(d, J = 10.68 Hz, 2H, -CH2), 3.32-3.34 (t, 1H, -CH), 5.4 (s,2H, -CH), 3.7 (s, 3H, -OCH3). 13C NMR (CDCi3, 6 ppm): 15.2.35.8, 40.9, 56.2, 58.6, 113.9, 114.4, 115.8, 123.7, 124.9, 127.1, 128.7, 133.1, 136.5, 138.9, 149.5, 153.6, 158.4, 163.7, MS (70 eV): m/z 423.18 [M1]. Anal.

Recent Developments in Chemistry and Blochemistry Research Ver Money of Buynmithes Using Zeone as a Green Catalyn in Buynmithes 1 Recent Developments in Commany Formers in Commany Research Va Evaluator of Antimicrobial Efficiency of Bigyrmidines Using Zeolde as a Graen Calabrit III Bigs 1

Caled. (found) % forC22H22N5O2CI. m.w. 423.18; C. 65.17 (65.15); H. 5.72 (5 73) CI, 8.36 (8 00); N. 17.27 (17.29); O. 8.84 (7.85).

2-Amino-4'-(4-methoxyphenyl)-6'-methyl-6'-(4-nitrophenyl)-3',4,4,5-2-Amino-4'-(4-methoxypheric) (1H)-one (3j) [3]: Light yellow solid, yield 85 to tetrahydro[4,5'-bipyrimidine]- 2 (1H)-one (3j) [3]: Light yellow solid, yield 85 to tetrahydro[4,5'-bipyrimidine]- 2 (1H)-one (3j) [3]: Light yellow solid, yield 85 to tetrahydro[4,5'-bipyrimidine]- 2 (1H)-one (3j) [3]: Light yellow solid, yield 85 to tetrahydro[4,5'-bipyrimidine]- 2 (1H)-one (3j) [3]: Light yellow solid, yield 85 to tetrahydro[4,5'-bipyrimidine]- 2 (1H)-one (3j) [3]: Light yellow solid, yield 85 to tetrahydro[4,5'-bipyrimidine]- 2 (1H)-one (3j) [3]: Light yellow solid, yield 85 to tetrahydro[4,5'-bipyrimidine]- 2 (1H)-one (3j) [3]: Light yellow solid, yield 85 to tetrahydro[4,5'-bipyrimidine]- 2 (1H)-one (3j) [3]: Light yellow solid, yield 85 to tetrahydro[4,5'-bipyrimidine]- 2 (1H)-one (3j) [3]: Light yellow solid, yield 85 to tetrahydro[4,5'-bipyrimidine]- 2 (1H)-one (3j) [3]: Light yellow solid, yield 85 to tetrahydro[4,5'-bipyrimidine]- 2 (1H)-one (3j) [3]: Light yellow solid, yield 85 to tetrahydro[4,5'-bipyrimidine]- 2 (1H)-one (3j) [3]: Light yellow solid, yield 85 to tetrahydro[4,5'-bipyrimidine]- 2 (1H)-one (3j) [3]: Light yellow solid, yield 85 to tetrahydro[4,5'-bipyrimidine]- 2 (1H)-one (3j) [3]: Light yellow solid, yield 85 to tetrahydro[4,5'-bipyrimidine]- 2 (1H)-one (3j) [3]: Light yellow solid, yield 85 to tetrahydro[4,5'-bipyrimidine]- 2 (1H)-one (3j) [3]: Light yellow solid, yield 85 to tetrahydro[4,5'-bipyrimidine]- 2 (1H)-one (3j) [3]: Light yellow solid, yield 85 to tetrahydro[4,5'-bipyrimidine]- 2 (1H)-one (3j) [3]: Light yellow solid, yield 85 to tetrahydro[4,5'-bipyrimidine]- 2 (1H)-one (3j) [3]: Light yellow solid, yield 85 to tetrahydro[4,5'-bipyrimidine]- 2 (1H)-one (3j) [3]: Light yellow solid, yield 85 to tetrahydro[4,5'-bipyrimidine]- 2 (1H)-one (3j) [3]: Light yellow solid, yield 85 to tetrahydro[4,5'-bipyrimidine]- 2 (1H)-one (3j) [3]: Light yellow solid, yield 85 to tetrahydro[4,5'-bipyrimidine]- 2 (1H)-one (3j) [3]: Light yellow solid, yellow solid, yellow solid, yellow solid, yell tetrahydro[4,5 -bipyrimiune), 3282, 1610, 1452, 1074, 'H NMR (400 B6 %) m p. 233 °C IR (KBr, vmax, cm '): 3282, 1610, 1452, 1074, 'H NMR (400 MHz, 400 MHz) m.p. 233 °C IR (KBr, Ymax, Grow), 8.5 (s. 1H, -NH), 8.5 (s. 1H, -NH), 4.0 (s. 2H, CDCI3, 8 ppm) 6.7-8.2 (m. 8H, arom.), 8.5 (s. 1H, -NH), 8.5 (s. 1H, -NH), 4.0 (s. 2H, CDCI3, 8 ppm) 6.7-8.2 (m. 8H, arom.), 8.5 (s. 1H, -NH), 8.5 (s. 1H, -NH), 4.0 (s. 2H, CDCI3, 8 ppm) 6.7-8.2 (m. 8H, arom.), 8.5 (s. 1H, -NH), 8.5 (s. 1H, -NH), 4.0 (s. 2H, CDCI3, 8 ppm) 6.7-8.2 (m. 8H, arom.), 8.5 (s. 1H, -NH), 8.5 (s. 1H, -NH), 4.0 (s. 2H, CDCI3, 8 ppm) 6.7-8.2 (m. 8H, arom.), 8.5 (s. 1H, -NH), 8.5 (s. 1H, -NH), 4.0 (s. 2H, CDCI3, 8 ppm) 6.7-8.2 (m. 8H, arom.), 8.5 (s. 1H, -NH), 8.5 (s. 1H, -NH), 4.0 (s. 2H, CDCI3, 8 ppm) 6.7-8.2 (m. 8H, arom.), 8.5 (s. 1H, -NH), 8.5 (s. 1H, -NH), 4.0 (s. 2H, CDCI3, 8 ppm) 6.7-8.2 (m. 8H, arom.), 8.5 (s. 1H, -NH), 8.5 (s. 1H, -NH), 8.5 (s. 1H, -NH), 4.0 (s. 2H, CDCI3, 8 ppm) 6.7-8.2 (m. 8H, arom.), 8.5 (s. 1H, -NH), 8.5 (s. 1H, -N -NH2), 2.2 (s. 3H. -CH3), 2.334, -OCH3), ¹³C NMR (75 MHz, CDCI3, 5 ppm), 152, -CH), 5.5(s, 2H, -CH), 3.7 (s. 3H, -OCH3), ¹³C NMR (75 MHz, CDCI3, 5 ppm), 152, -CH), 5.5(s, 2H, -CH), 5.1 (s, 114.4, 115.8, 124.9.127.1, 128.7, 133.9, 135.7, 149.5, 35.8, 40.9, 56.2, 58.6, 113.9, 114.4, 115.8, 124.9.127.1, 128.7, 133.9, 135.7, 149.5, 35.8, 40.9, 56.2, 58.6, 113.9, 114.4, 115.8, 124.9, 127.1, 128.7, 133.9, 135.7, 149.5, 124.9, 127.1, 128.7, 139.5, 124.9, 127.1, 128.7, 139.5, 124.9, 127.1, 128.7, 139.5, 124.9, 127.1, 128.7, 139.5, 124.9, 127.1, 128.7, 139.5, 124.9, 127.1, 128.7, 139.5, 124.9, 127.1, 128.7, 139.5, 135.7, 149.5, 124.9, 127.1, 128.7, 139.5, 124.9, 127.1, 128.7, 139.5, 124.9, 127.1, 128.7, 139.5, 124.9, 127.1, 128.7, 139.5, 124.9, 127.1, 128.7, 139.5, 124.9, 127.1, 128.7, 139.5, 124.9, 127.1, 128.7, 139.5, 124.9, 127.1, 128.7, 149.5, 124.9, 127.1, 128.7, 149.5, 124.9, 127.1, 128.7, 149.5, 124.9, 127.1, 128.7, 149.5, 124.9, 127.1, 128.7, 12 35.8, 40.9, 55.2, 56.0, 113.5, 114.9, 11/2 423.16 [M*]. Anal. Calod. (found) 149.5, 153.6, 163.7, 165.2, MS (70 eV): m/z 423.16 [M*]. Anal. Calod. (found) 149.5, 153.6, 163.7, 165.2, 102.16, 0 52.34 (62.35); H 5.23 (5.25); N 16.52 (100.16); for 153.6, 163.7, 165.2, M6 (16 C, 62.34 (62.36); H, 5.23 (5.25); N, 16.52 (16.53); C 22H22N5O2CI.m.w. 423.16; C, 62.34 (62.36); H, 5.23 (5.25); N, 16.52 (16.53); C 1

3. RESULTS AND DISCUSSION

In a model reaction of 5-cinnamoyl-6-methyl-4-phenyl- 3,4-dihydropyrimidin-2(1H)-one (2a) and guanidine hydrochloride in biphasic system of solvents of DCM and water weresubjected to catalytic reaction in presence of NaY zeolite forthe 2-amino-6-methyl-4,6-diphenyl-3',4,4',5bipyrimidine]-2' (1H)-one (3a) at 60 °C for 30min stirring [3]. Initially, our efforts focused on delineating a simple, green condensation reaction of 5-cinnamoyl-6methyl-4-phenyl-3,4-dihydropyrimidin-2(1H)-one (2a) and guanidine hydrochloridein presence of NaY zeolite to get 2-amino-6-methyl- 4.6diphenyl-3',4,4',5tetrahydro-[4,5'-bipyrimidine]-2' (1H)- one (3a). We proceeded with an organic solvent of dichloro-methane which came with only 12 % yield. Further, we worked on the reaction with other aprotic solvents as n-hexane toluene, cyclohexane, ethyl acetate and THF, results were unsuccessful. Looking into the disadvantages of the monophasic solvent system, we focused our efforts on the organic-aqueous biphasic system to get better yields of product [20-22,3]. The biphasic solvents play a crucial role in determining the amountof product in catalysis. The first step towards determining the biphasic organic solvents that are immiscible with water, resultsin the minimum number of toxic side products and intermediates with a clean environmental approach [23] (Table-3) [3]. All other permutations generated by varying parameters, such as concentration of zeolite, reaction time for condensation, reaction temperature; and the ratio of the organic-aqueous system led to lower yields. The extensive optimization for the various biphasic systems (Table-3), and DCM-water 1:1 ratio (Table-4) in 30 mol % of catalyst(Table-5) furnished astonishing results with 89 % in yield. The possible explanation for such an amazing behaviour of organic-aqueous reaction media of DCM-water system is due to characteristic properties of DCM as compared to other solvents i.e.high polarity, high dielectric constant, high dipole moment and low boiling point of dichloromethane [3].

Table 4 shows that the organic solvent and water ratio had a significant impact on the product yield. To understand the effect of the organic phase on the yield of the product, the distribution of different ratios of the biphasic system was carried out during the reaction. Initially, only in the monophasic system gave we could Recent Developments in Chemistry and Blochemistry Research Vol. 3 Evaluation of Antimicrobial Efficiency of Bipyrimiones Using Zeolite as a Green Catalyst in Biohasic System

obtain 12 % yield which was very low. However, using the different ratio of the DCMwater system with zeolite, the yield obtained was dramatically enhanced [3]. When the proportion ratio reached 1:1, it was observed that the reaction yield was greatest. Further increasing the organic solvent ratio resulted in a decrease in yield. It is interesting to note that the amount of product was reduced when the reaction was carried in pureorganic phase and when the reaction mixture was diluted withaqueous phase the yield was augmented which indicated thatwater is essential for this reaction [3].

Entry No.	Solvent system	Time (min)	Yield (%)	Ratio Organic:Aqueous
1	Toluene-water	60	30	7.3
2	DCM-water	30	45	7.3
1	Cyclohexane-water	70	25	7.3
1	n-Hexane-water	66	23	7:3
5	Ethylacetate-water	80	22	7.3
6	DMA-water	75	30	7:3

fable 3. Optimization of variou	s organic-aqueous	solvent system	ns
---------------------------------	-------------------	----------------	----

S. No.	Organic solvent (DCM) (mL)	Water (mL)	Temp. (°C)	Yield (%)	Time (min)
1	7	4	60	50	45
2	7	5	60	56	42
3	7	6	60	75	40
4	7	7	60	89	30

Table. 4 Optimization of ratio of DCM-water system

Table 5.	Optimization o	fcatalyst	concentrations for	synthesis of	bipyrimidines
(ED20393)	1949/1941 - 1941		(3a-j)	71202312331	99923002399.0240

S. No.	Catalyst conc. (mmol %)	Reaction time (min)	Yield (%)
1	10	120	32
2	12.5	115	35
3	16	100	45
4	20	90	47
5	22.5	75	56
6	25	50	87
7	30	30	89

The purity of the compound was confirmed by a single spotin TLC. In IR spectra of compound 2a carbonyl absorption at 1722-1690 cm⁻¹ and -NH absorption at 3225 cm⁻¹ have been observed. The compound showed a mass ion peak of 100 % intensity corresponding to its molecular weight in mass spectrafurther confirming the structure [3]. Furthermore, ¹H NMR spectraof compound 2a, a singlet appeared at δ 2.12 ppm owing to three protons of -CH3. A doublet at δ 6.93 to δ 6.88 ppm is assigned for one proton of the -CO-CH= group of chalcone with (J = 17.8 Hz). The peak of doublet at δ 7.46 to δ 7.40 ppm represents one proton of =CH-Ar with (J =

Symmetry

Recent Developments in Chemistry and Biochemistry Research kol Communication of Buylimidines Using Zeoire as a Green Catalyst – Brank a Recent Developments in University and Constity Research Vol Evaluation of Antimicrobial Efficiency of Burylimidines Lising Zeolite as a Green Catalyst = 8-04-0 00-00-0

17.12 Hz). A singlet for one proton of -NH appears at 5.8.32 ppm and singlet of 17.12 Hz). A singlet for one proton of -CH of the Biginelli ring corresponds at 5.50 ppm. Asinglet for one proton of -CH of the Biginelli ring corresponds at 5.50 ppm. Asinglet for one proton of -CH or the bighten of the multiplet peaks at 6 7.6 to 5 6.9 proton of -NH appears at 6 8.23 ppm. The multiplet peaks at 6 7.6 to 5 6.9 proton of -NH appears at in other protons indicating the presence of two phenyl in 6.9 corresponds to 10 aromatic protons indicating the presence of two phenyl in 6.9 corresponds to 10 aromatic protons indicating the presence of two phenyl in 6.9 corresponds to 10 aromatic protons indicating the presence of two phenyl in 6.9 corresponds to 10 aromatic protons indicating the presence of two phenyl in 6.9 corresponds to 10 aromatic protons indicating the presence of two phenyl in 6.9 corresponds to 10 aromatic protons indicating the presence of two phenyl in 6.9 corresponds to 10 aromatic protons indicating the presence of two phenyl in 6.9 corresponds to 10 aromatic protons indicating the presence of two phenyl in 6.9 corresponds to 10 aromatic protons indicating the presence of two phenyl in 6.9 corresponds to 10 aromatic protons indicating the presence of two phenyl in 6.9 corresponds to 10 aromatic protons indicating the presence of two phenyl in 6.9 corresponds to 10 aromatic protons indicating the presence of two phenyl in 6.9 corresponds to 10 aromatic protons indicating the presence of two phenyl in 6.9 corresponds to 10 aromatic protons indicating the presence of two phenyl in 6.9 corresponds to 10 aromatic protons indicating the presence of two phenyl in 6.9 corresponds to 10 aromatic protons indicating the presence of two phenyl in 6.9 corresponds to 10 aromatic protons indicating the presence of two phenyl in 6.9 corresponds to 10 aromatic protons indicating the presence of two phenyl in 6.9 corresponds to 10 aromatic protons indicating the presence of two phenyl in 6.9 corresponds to 10 aromatic protons indicating the presence of two phenyl in 6.9 corresponds to 10 aromatic protons indicating the presence of two phenyl in 6.9 corresponds to 10 aromatic phenyl corresponds to 10 aromatic protons N_2O_2 which exhibited molecular ion at m_{V_2} = The product 2a was analyzed for $C_{20}H_{10}N_2O_2$ which exhibited molecular ion at m_{V_2}

The spectra of the hybrid of synthesized compounds (3a-j) were analyzed by ¹H NMR. IR and mass spectra. The product 3a was analyzed for C21H21N3O which

3.1 Antimicrobial Evaluation

All the synthesized compounds were screened as potent antibacterial and antifungal scaffolds3a-j. The microbial study was carried out against two Gramnegative bacterial strains namely E. coli (ATCC 25922) and P. aeruginonasa (ATCC 8532) and one Gram-positive bacterialstrain named S. aureus (ATCC 29213) and against two fungalstrains namely as C. albicans (ATCC 10231) and A. niger (ATCC 439). The screening was performed with standard drugs such as ampicillin for antibacterial activity and ketoconazole for antifungal activity [3]. The microbial activities of bipyrimidine derivatives were further evaluated for the minimum inhibition concentration (MIC). The inhibitions of microbial growth were used to demonstrate the therapeutic efficacy of hybrid scaffolds. The activity data is illustrated in Tables 6-8. The results of microbial analysis of the tested compounds revealed that thesehybrids have shown moderate to good antibacterial efficacy against selected bacteria strains (E. coli, P. aeruginosa and S. aureus) [3]. On the basis of the zone of inhibition test against test bacterium, E. coli, compounds 3i (R1 = CI, R3 = OCH3), 3e (R1 = CI, R3 = H) and 3f (R1 = H, R3 = CI) were found to have verygood activity; and compounds 3d (R1 = OCH3, R3 = OCH3), 3c(R1 = H. R3 = OCH3 and 3a (R1 = H, R3 = H) possessed good activity, while compound 3i (R1 = CI, R3 = OCH3), 3h (R1 = OCH3, R3 = NO2), 3g (R1 = NO2, R3 = H), 3b (R1 = H, R3 = NO2) showed moderate activity when compared with the standard drug ampicillin. In the case of P. aeruginosa, compounds 3i and3f very good activity and compounds 3a, 3c, 3d, 3e and 3j exhibited good activity while compounds 3b, 3g and 3h resulted in moderate activity as compared to the standard drug ampicillin. For S. aureus, compounds 3e, 3f and 3i showedvery good activity, compounds 3a, 3c, 3d and 3j possessed good activity while compounds 3b, 3g and 3h revealed moderate activity in comparison to the standard drug ampicillin [3].

For fungal strains, the screenings of synthesized compounds have revealed very good activity to moderate activity. For Candida albicans, compounds 3f and 3i showed very goodactivity and compounds 3c and 3e possessed good activity while compounds 3a, 3b, 3d, 3g, 3h and 3i exhibited moderate activity as compared to the standard drug ketoconazole. In the case of A. niger, compounds 3e and 3i revealed very good activity and compounds 3a, 3c, 3d, 3f and 3g possessed good activitywhile compounds 3b, 3h and 3j displayed a moderate zoneof inhibition [3].

nd Biochemistry Research Vol. 3	Green Catalyst in Biphasic System	
Recent Developments in Chemistry a	tion of Antimicrobial Efficiency of Bipyrimidines Using Zeolife as a	
	Evalua	

Table 6. inhibition zones (mm) of synthesized bipyrimidines (3a-j) againstbacteria and fungi by the disk diffusion method

		10 - Farmer		Microbial	at other a second	1110000		The second s	Fur	legn	000000000000000000000000000000000000000
Compound	μ.	Scherichie ATCC 259	a coli (22)	P. aeruginonas (ATCC 8532)	Staphy (ATC	lococcus C 29213)	aureus	Candida (ATCC	albicans (10231)	Aspergi (ATC(llus niger C 439)
	50 µg	25 µg	12.5 µg	100 µg	100 μg	50 µg	25 µg	100 μg	50 µg	100 μg	50 µg
3a	11	12	10	11	12	12	10	10	07	10	07
3b	01	E	60	1	11	60	1	10	08	60	8
30	12	12	10	12	12	12	10	12	60	E	60
3d	12	F	10	11	13	12	10	F	07	12	60
3e	14	12	10	12	4	14	F	13	60	4	60
3	4	12	11	13	14	13	÷	14	10	51	60
30	01	11	08	90	10	80	1	H	80	10	50
35	0	1	60	07	1	10	ţ	H	20	60	99
31	15	12	11	13	15	14	12	15	Ħ	15	10
3	10	10	69	11	12	12	1	11		08	90
Ampicillin	18	15	12	16	32	29	24	4	ī	ĩ	r
Ketoconazole	1	1	1	1	1	1	1	20	12	20	12

.

65

Evaluation of Antimicrobial Efficiency of Bipyrimidines Using Zeolde as a Green Catalyst in Biphase System

Table 7. Minimum inhibitory concentrations (µG/ML) of synthesized 3a-J against and Gram-positive and Gram-Negative

Ľ.	1	1				-	man cou	gunoniesa	ATCC	Charle				
	~	200	100	Ann	4.5.6		8532			udese	ylac	OCCUS.	ATC	199
1	0			UIU	2	2	50	100	Ann	100	10	reus		
	• •	:	Ŧ	3	1	5			MICH	5.21	25	20	100	
	•	+	Ŧ	100		ŧ	\$:	25					
	:	;	+++	200	1	I	;	:		•	•	:	;	
	ŧ	;	:	9.5	r.	:	;	;	88	E	+	;	;	
	:	ŧ	4	3	ł,	:	;		Q a	ī.	•	:	:	
	:	:		93	1	:	;		33	1	+	:		
	•	+		24	ĩ	:	;		82	1	+	\$;	
	+	+	: :	001	i	1	+	1	5	ŗ	;	:	;	
	‡	:	-	00.2	,	1	+	4	80	,	+		;	
-1	+	+	;	09	1	\$	\$	1	100	1		+	;	
1	+++	:	1	nn,			+	:	1	•	:	;	:	
man	growth,	No onhibity	Will Interthing	10 40	#	:	:	:	nne	1		:	‡	
			Sallana III	MULTIN . 1	THEORY (+) >	V active /	07-40	1		;	ŧ			L

99

-	15
1	2
5	12
~	150
4	400
75	12
÷.,	- 24
*	. 46
e.	12
2	- 124
Ξ	100
œ.	-
-	12
1.	-
75	100
÷	27
÷.	10
÷	-
£.	73.
25	ы.
2	- 22
5	42
1	10
*	10
ĩ.	4
8	
2	10
2	-
4	17
	-92
E.	-5
Ξ.	- 62
π.	50
5	74
ч.	100
-	÷.
4	25
*	5
÷	-
τ.	10
Ξ.	
z.	5
۰.	*
0	2
*	se.
Ξ.	*
6	6
6	20
_	1
Ξ.	**
2	8
σ.	25
ō.	8
ΩĽ.	1
-	
	32
	=
	1
	-
	2
	ъ.
	÷.
	×.
	ъ.
	2
	-
	-0
	-
	4
	5
	5
	4
	12
	2
	-81
	14
	UL.

Entry		Candi	da albicans A	VTCC (10231		8	Aspergillus n	iger -	A.	rcc (439)
2	12.5	25	50	100	App MIC	12.5	25	50	100	App MIC
2		+	:	‡	50	1	+	+	\$	100
2	1	1	:	:	20	1	3	\$	‡	20
2	4	:	:	:	25	1	;	‡	:	25
P	,	•	:	:	50	1	t	+	:	100
e	1	;	;	:	25	1	\$;	;	25
*	1	\$	\$;	25	1	;	:	:	25
pg	1	Ţ	1	;	100	1	1	•	:	100
5		,	+	;	100	ţ	ţ	•	:	00
10	1	‡	‡	;	25	1	:	:	;	25
8	1	1	+	:	100	h	1	1	:	100
Ketoconazole	ŧ	‡	ŧ	‡		ŧ	+++	:	:	:

Table 8. Minimum inhibitory concentrations (µG/ML) of synthesized 3a-J against fungl

3 < - C

1.10.000

Burnert Developments in the long factor as a realist the long to the second state of t Constant of Red

3.2 Antibacterial Activity

For evaluation of microbial activity, as minimum inhibition concentration of For evaluation of microbial activity, as minimum inhibition concentrations as 12.5 or of For evaluation of microbial activity, as transmission concentrations as 12.5, 25 to be synthesized compounds was studied at different concentrations as 12.5, 25 to be compounds. 3c, 3e, 3f and 3i exhibited very good inhub the compounds activity produces the second activity good inhub the second activity good For evention compounds was studied at uncertained at and 3i exhibited very 900d into 3 to 30 million and 30 showed good activity with Mile and 30 million and 30 showed good activity with Mile and 30 million and 30 mi at MIC = 25 µg/mL Compounds 3b. 3g. 3h and 3j revealed moderate activity with MIC = 35 at MIC = 25 µg/mL. Compounds be and 3j revealed moderate activity with MIC is g/mL and the compounds 3b. 3g. 3h and 3j revealed moderate activity with MIC is M/C is M/C.

Against P. eerugenonasa, compounds 3d, 3f and 3i exhibited very good inhibition as Compounds 3b, 3c, and 3e compounds revealed good inhibition as Against P. aeruginonasa, compounds so, and 3e compounds revealed good inhibition as MIC= 25 up/mL. Compounds 3b, 3c, and 3e compounds revealed good inhibition and compounds 3a, 3g, 3h and 3j resulted MIC = 100 $MIC = 25 \mu g/mL$ Compounds 30, 90 and 3g, 3h and 3g resulted $MIC = 100 \mu g/mL$ with $MIC = 50 \mu g/mL$ and compounds 3a, 3g, 3h and 3g resulted $MIC = 100 \mu g/mL$ compounds $\mu g/mL$ compounds $MIC = 25 \mu g/mL$ compounds $\mu g/mL$ with MIC = 50 μ g/mL and composition good inhibitional MIC = 25 μ g/mL, compounds 3f and 3 exhibited very good inhibitional MIC = 25 μ g/mL, compounds 3f and 3 exhibited very good activity with MIC = 50 μ g/mL and compounds 3a Compounds 3f and 3i exhibited very getter with MIC = 50 µg/mL and compounds 3a and 3e compounds showedgood activity with MIC = 100 µg/mL and compounds 3c and 3e compounds showed your activity with MIC = 100 $\mu g/mL$ against g

3.3 Antifungal Activity

Minimum inhibition concentrations of all the derivatives of bipyrimidines for antifungal activity were studied and the results of compounds 3c, 3e, 3f and 3 exhibit very good inhibition by way of MIC = 25 µg/mL. Compounds 3a, 3b and 3d and 3d compounds revealed good inhibition withMIC = 50 µg/mL and compounds 3g. 3hard3 whimoderate inhibitionMIC = 100 µg/mL against C. albicans. In the case of A niger MIC = 25 µg/mL exhibits very good inhibition of compounds3c, 3e, 3f and 3i Compounds 3b showed good inhibition with MIC = 50 µg/mL and compounds 3a 3d, 3g, 3h and 3j revealed moderate inhibition with MIC = 100 µg/mL [3].

The results depicted in Tables 7 and 8 suggested that theelectron-withdrawing substituents revealed very significant minimum inhibition concentration whereas electron-donating substituents exhibited insignificant minimum inhibition concentration activity against microbial strains [3].

4. CONCLUSION

An expedited process catalyzed by zeolite has the merit of being an environmentally friendly simple operation, involving convenient workup, a short reaction time and resulting in goodto excellent yields. Substituted 2-amino-6-methyl-4.6-diphenyl- 3',4,4',5-tetrahydro-[4,5'-bipyrimidine]-2'(1H)-one synthesized by substituted chalcone and guanidine hydrochloride in biphasic system was confirmed by spectral characterization. The synthesized scaffolds of bipyrimidines were studied as anti-microbial agents [3]. The investigation of antimicrobial screeningdata revealed that among 10 compounds screened, compounds3d, 3e, 3f and 3i demonstrated very good activity as compared to standard drugs and remaining compounds showed good tomoderate inhibition activities.

Recent Developments in Chemistry and Biochemistry Research Vol. 3 Evaluation of Antimicrobial Efficiency of Biogramitiones Using Zeolite as a Green Catalyst in Biotrasic System

ACKNOWLEDGEMENTS

The authors are thankful to Sophisticated Analytical Instrumentation Facility (SAIF). Chandigarh, India for ¹H NMR, ¹³C NMR, IR, mass and CHN spectroscopic chanoigend TheMicrocare Laboratory and Tuberculosis Research Centre, Surat, India for antimicrobial activities.

COMPETING INTERESTS

Authors have declared that no competing interests exist.

REFERENCES

- Pan T, Wu Z, Yip AC. Advances in the green synthesis of microporous 1.1 and hierarchical zeolites: a short review. Catalysts. 2019;9(3):274.
- Hartmann M, Machoke AG, Schwieger, W, Catalytic test reactions for the 2evaluation of hierarchical zeolites. Chem. Soc. Rev. 2016;45:3313-3330.
- Aswale MS, Dhankar RP. Synthesis and Antimicrobial Evaluation of 3. Bipyrimidines in Efficient Biphasic System using Zeolite as Green
- Catalyst, Asian Journal of Organic & Medicinal Chemistry, 2018;3(4). 143-149.

Available:https://doi.org/10.14233/ajomc.2018.AJOMC-P140

B. Chandrasekaran, S.Cherukupalli, S. Karunanidhi, A. Kaiee, R.Reddy 4. Aleti N. Sayyad B. Kushwaha S. Reddy Merugu K.Mlisana R. Karpoormath, Design and synthesis of novel heterofused pyrimidine analogues as Journal of Molecular Structure. effective antimicrobial agents. 2019;1183:246.

Available:https://doi.org/10.1016/j.molstruc.2019.01.105

- Kumar AK, Bhagat A. Singh H, Singh T, Angrea A, Verma H, Khalilullah 5. M. Jaremko A. Emwase and P.Kumar, Medicinal chemistry perspective of pyrido[2,3-d] pyrimidines as anticancer agents, RSC Adv. 2023;13, 6872. Available:https://doi.org/10.1039/D3RA00056G
- Zhan W, Mao P, Yuan C, Zhang Y, Zhang T, Liu Y, Tian J, Xue W. Design, 6. synthesis and antiviral activities of chalcone derivatives containing pyrimidine, Journal of Saudi Chemica Society. 2023;27:101590. Available:https://doi.org/10.1016/j.jscs.2022.101590
- Rashid H, Martines M, Duarte A, Jorgea J, Rasool S, Muhammad R, 7. Ahmad N, Muhammad N Umar, Research developments in the syntheses, activities and structure-activity relationships of anti-inflammatory pyrimidines, RSC Adv., 2021;11:6060.
 - Available:https://doi.org/10.1039/D0RA10657G
- Wu W, Lan W, Wu C, Fei Q. Synthesis and Antifungal Activity of Pyrimidine 8. Derivatives Containing an Amide Molety, Frontiers in Chern, 2021;9:695628.
- Available:https://doi.org/10.3389/fchem.2021 Patil SB. Recent medicinal approaches of novel pyrimidine analogs: A 9. review, Hellyon, 2023;9:e16773.
 - Available:https://doi.org/10.1016/j

TOTTLY

Recent Developments in Chemistry and Biochemistry Research is a Green Celaint in Base ve Becant Developments in Chemistry and Discremistry Research Evaluation of Antimicrobial Efficiency of Bipprimitines Using Zeolina data Green Calager in Bipprimitines Discrete Calager in Bipprimitines Using Zeolina data Green Calager in Bipprimitines Using Zeolina data a Green Calager in Bipprimitines Bipprimitines Using Zeolina data a Green Calager in Bipprimitines Using Zeolina d

- Basha J. Goudgaon N. A comprehensive review on pyrimidine enaligy. Basha J. Goudgaon N. A comprehension biological potential, dournal and biological potential, dournal of versable scaffold with medicinal and biological potential, dournal of versable scaffold with medicinal and biological potential, dournal of the scaffold with medicinal and biological potential, dournal of the scaffold with medicinal and biological potential, dournal of the scaffold with medicinal and biological potential, dournal of the scaffold with medicinal and biological potential, dournal of the scaffold with medicinal and biological potential, dournal of the scaffold with medicinal and biological potential, dournal of the scaffold with medicinal and biological potential, dournal of the scaffold with medicinal and biological potential, dournal of the scaffold with medicinal and biological potential, dournal of the scaffold with medicinal and biological potential, dournal of the scaffold with medicinal and biological potential, dournal of the scaffold with medicinal and biological potential, dournal of the scaffold with medicinal and biological potential, dournal of the scaffold with medicinal and biological potential, dournal of the scaffold with medicinal and biological potential, dournal of the scaffold with medicinal and biological potential and biological potential, dournal of the scaffold with medicinal and biological potential and biological potential, dournal of the scaffold with medicinal and biological potential and biological potential and biological potential, dournal of the scaffold with medicinal and biological potential and biological potential and biological potential, dournal of the scaffold with medicinal and biological potential and biological to.
- Available https://doi.org/10.10.roy.mas.Deligiannidou G, Patsilinakos A, Myriagkou M, Papakonstantinou E, Deligiannidou G, Patsilinakos A, Myriagkou C, Eleni Pontiki, Athanassopoulos C, Novel Pyrimia A, Available M. Papakonstantinou E. Compoulos C. Novel Pyrimidine Kontogiorgis C. Eleni Pontiki, Athanassopoulos C. Novel Pyrimidine Kontogiorgis S. Antioxidant and Anticancer Agents: Design, Synthesis 11. Kontogiorgis C, Eleni Pontiki, Kunancer Agents: Design, Synthioline Derivatives as Antioxidant and Anticancer Agents: Design, Synthesis and Modeling Studies, Molecules, 2023;28(9):3913. Molecular Modeling Studies, Molecules, 2023;28(9):3913. Available https://doi.org/10.3390%2Fmolecules28093913
- Available https://doi.org/10.33500/Castagnolo D, Roh J, Korabecny J Finger V, Kufa M, Soukup O, Castagnolo D, Roh J, Korabecny J Finger V. Kufa M. Soukup G. Buropean Journal of Pyrimidine derivatives with antitubercular activity. European Journal of Journal of 2023;246:114946 12. Available.https://doi.org/10.1016/j.ejmech.2022.114946
- Available https://doi.org/10.1616/j.c. Singh K, Kaur T, Pyrimidine-based antimalarials: Design strategies and MedChemComm. 2016;7(5):749. 13. antiplasmodial effects, MedChemComm, 2016;7(5):749. Available:https://doi.org/10.1039/C6MD00084C
- Available:https://doi.org/10.10/ Mohan Kandepi VVK, Narender N. Synthesis of N-Heterocyclic Mohan Kandepi VVK, Molecular Sieve Catalysts: An Arecyclic 14. Mohan Kandepi VVN, Molecular Sieve Catalysts: An Approach Compounds Over Zeolite Catal Sci Technol. 2012;2:471 towards Green Chemistry, Catal. Sci. Technol. 2012;2:471, Available https://doi.org/10.1039/C2CY00162D
- Available https://doi.org/ Ni Y, Sun A, Wu X, Hai G, Hu J, Li T, Li G. The Preparation of Nano-Sized 15 NLY, Sun A, Wu A, that G, and Its Application in the Aromati- sized H[Zn, Al]ZSM-5 Zeolite and Its Application in the Aromati- zation of Methanol, Micropor, Mesopor, Mater, 2011;143;435. Available:https://doi.org/10.1016/j.micromeso.2011.03.029.
- Available https://doi.org/ Bushuev MB, Pishchur DP, Logvinenko VA, Gatilov YV, Korolkov IV, Shundrina 16. Bushuev MB, Pishthar LY, Logalov VP, A Mononuclear Iron(II) Complex IK, Nikolaenkovac EB, Krivopalov VP, A Mononuclear Iron(II) Complex Cooperativity, Kinetics and Activation Energy of the Solvent-Dependent Spin Available:https://doi.org/10.1039/C5DT03750F
- Palucki M, Pyrimidines, Tetrahedron Organic Chemistry Series, Chap. 17. 2007; 26(11):475-509.
 - https://doi.org/10.1016/S1460-1567(07)80060-2.
- Collins CH, Lyne PM, Grange JM, Falkinham JO III, Collins and Lyne's 18. Microbiological Methods, Arnold, a Member of the Hodder Headline Group:
- Yao J, Moellering R, eds. P. Murray E. Baron, M. Pfaller F. Tenover and R. 19. Yolken: Antibacterial Agents, In: Manual of Clinical Microbiology, ASM, Washington DC. 1995:1281-1290.
- 20. Wang TF, Nolte MW, Shanks BH, Catalytic Dehydration of C6 Carbohydrates for the Production of Hydroxymethylfurfural (HMF) as a Versatile Platform Chemical, Green Chem. 2014;16:548. Available:https://doi.org/10.1039/C3GC41365A
- 21, Tian G, Tong X, Cheng Y, Xue S. Tin-Catalyzed Efficient Conversion of Carbohydrates for the Production of 5-Hydroxymethyl- furfural in the Presence of Quaternary Ammonium Salts, Carbohydr, Res. 2013;370-33. Available:https://doi.org/10.1016/j.carres.2013.01.012

Recent Developments in Chemistry and Biochemistry Research Vol. 3 Evaluation of Antimicrobial Efficiency of Bipyrimidines Using Zeotte as a Green Catalyst in Biphasic System

Crossley S, Faria J, Shen M, Resasco DE, Solid Nanoparticles that Catalyze Biofuel Upgrade Reactions at the Water/Oil Interface, Science, 22. 2010:327:68 Available:https://doi.org/10.1126/science.1180769

Csihony S. Mika LT. Vlad G. Barta K. Mehnert CP. Horvath IT. Oxidative Carbonylation of Methanol to Dimethyl Carbonate by Chlorine- Free 23. Homogeneous and Immobilized 2,2-Bipyrimidine Modified Copper Catalyst, Coll. Czech. Chem. Commun., 2007;72:1094. Available:https://doi.org/10.1135/cccc20071094

आदिवासी पारधी (हिरापूर)

डॉ. शरयू म. पोतनुरवार

आदिवासी पारधी (हिरापूर)

डॉ. शरयू मनिष पोतनुरवार (बिडवाई)



सीवली पब्लिकेशन्स, नागपूर (महाराष्ट्र)

आदिवासी पारधी (हिरापूर) ADIWASI PARADHI (HIRAPUR)

डॉ. शरयू मनिष पोतनुवार (बिडवाई)

Dr. Sharyu Manish Potnurwar (Bidwai)

प्रकाशक	शशी भोवते,
	सीवली पब्लिकेशन्स, नागपूर
	३-१, रोड क्र. ४, द्वारकापूरी, नागपूर
	मोबा. ९८८१७१२१४९
पृष्ठ संख्या	60
प्रथमावृत्ती	१५ जुलै, २०२४
मुखपृष्ठ	शशी भोवते
अक्षरांकन	सीवली ग्राफिक्स, नागपूर
	मो. ७७२१८०९२५०
मुद्रक	वंश क्रिएशन, नागपूर
	मो. ९५९५७२७६१४
मूल्य	₹ १२५/-
ISBN	978-81-942663-5-8

अनुक्रमणिका

٠	आदिवासी संकल्पना	9
•	आदिवासी जमातीचे लोकसंख्यात्मक विश्लेषण	88
•	आदिवासींवर विविध प्रभाव	१३
•	स्वातंत्रोत्तर भारतातील आदिवासींची परिस्थिती	१६
•	आदिवासीचे कलाकौशल्ये	२०
•	क्रांतीकारी आदिवासी	55
	आदिवासी कल्याण कार्यक्रमात शासनाची भूमिका	३३
•	अनुसूचित जाती व जमातींसाठीचे कायदे	३६
•	आदिवासींसाठी विविध विकास योजना	36
	आदिवासी पारधी जमात	85
•	पारधी जमातीच्या चालीरिती	४७
•	आदिवासी पारधी (विशेष ग्राम-हिरापूर)	46
•	आदिवासी पारधी यांच्या सर्वांगिण	60
	विकासाकरिता आवश्यक सुविधा	
•	संदर्भ	ξe



डॉ.शरयू मनीष पोतनूरवार, सहयोगी प्राध्यापक, एम.ए.अर्थशास्त्र, बी.एड.,पी.एच.डी., सरदार पटेल महाविद्यालय चंद्रपूर येथे सन 2000 पासून अर्थशास्त्र विभाग प्रमुख म्हणून कार्यरत आहेत. राष्ट्रीय, आंतरराष्ट्रीय, अशा विविध स्तरावर 52 शोध निबंध प्रकाशित झालेले आहेत.





DIGITAL TWIN FOR SMART MANUFACTURING

Edited by Rajesh Kumar Dhanaraj Ali Kashif Bashir Rajasekar Vani Balamurugan Balusamy Pooja Malik



Second the Present

Academic Press is an imprint of Elsevier 125 London Wall, London EC2Y 5AS; United Kingdom 525 B Street, Suite 1650, San Diego, CA 92101, United States 50 Hampshire Street, 5th Floor, Cambridge, MA 02139, United States The Boulevard, Langford Lane, Kidlington, Oxford OX5 1GB, United Kingdom

Copyright @ 2023 Elsevier Inc. All rights reserved.

No part of this publication may be reproduced or transmitted in any form or by any means, electronic or mechanical, including photocopying, recording, or any information storage and retrieval system, without permission in writing from the publisher. Details on how to seek permission, further information about the Publisher's permissions policies and our arrangements with organizations such as the Copyright Clearance Center and the Copyright Licensing Agency, can be found at our website: www.elsevier.com/permissions.

This book and the individual contributions contained in it are protected under copyright by the Publisher (other than as may be noted herein).

MATLAB³⁶ is a trademark of The MathWorks, Inc. and is used with permission. The MathWorks does not warrant the accuracy of the text or exercises in this book. This book's use or discussion of MATLAB³⁰ software or related products does not constitute endorsement or sponsorship by The MathWorks of a particular pedagogical approach or particular use of the MATLAB³¹ software.

Notices

Knowledge and best practice in this field are constantly changing. As new research and experience broaden our understanding, changes in research methods, professional practices, or medical treatment may become necessary.

Practitioners and researchers must always rely on their own experience and knowledge in evaluating and using any information, methods, compounds, or experiments described herein. In using such information or methods they should be mindful of their own safety and the safety of others, including parties for whom they have a professional responsibility.

To the fullest extent of the law, neither the Publisher nor the authors, contributors, or editors, assume any liability for any injury and/or damage to persons or property as a matter of products liability, negligence or otherwise, or from any use or operation of any methods, products, instructions, or ideas contained in the material herein.

ISBN: 978-0-323-99205-3

For Information on all Academic Press publications visit our website at https://www.elsevier.com/books-and-journals

Publisher: Matthew Deans Acquisitions Editor: Sophie Harrison Editorial Project Manager: Rafael G. Trombaco Production Project Manager: Prasanna Kalyanaraman Cover Designer: Mark Rogers

Working together to grow libraries in developing countries

Typeset by MPS Limited, Chennai, India

CHAPTER 7

The convergence of digital twin, Internet of Things, and artificial intelligence: digital smart farming

Shipra Yadav¹, Amir Ahmad Dar², Pankaj Dhumane³ and

D. Kalaskar Keshao⁴

¹IICC, RTM Nagpur University, Nagpur, Maharashtra, India

²Department of Mathematical Sciences, Islamic University of Science and Technology, Pulwama, Jammu and Kashmir, India

Sardar Patel College, Chandrapur, Maharashtra, India

⁴Dr. Ambedkar College, Chandrapur, Maharashtra, India

7.1 Inroduction

Development of many countries depends heavily on agriculture, which is also essential to achieving Sustainable Development Goal 2 of "Zero Hunger" [1]. To fulfill the demands of a projected population of 10 billion people by 2050, the Food and Agriculture Organization estimates that agricultural output must increase by 40% between 2012 and 2050 [2]. The creative application of technologies like drones, applications, and machines coupled with social, organizational, and institutional shifts is one strategy for enhancing the production.

Around 70% of the freshwater consumed globally is used for agriculture [3]. This makes a strong case for the development of technologies like the Internet of Things (IoT) to increase the amount of food produced on farms while reducing the amount of water required in agriculture.

Irrigation systems utilize the majority of the freshwater that is present, and 40% of the freshwater used in developing nations is wasted due to leakages and over irrigation [4]. The availability of freshwater has become a global concern due to factors including climate change and population growth, specifically due to rainfall.

The regions facing scarcity [5] call for a different viewpoint on irrigation systems and their optimization to guarantee food security for the expanding population [6]. In agriculture, appropriate irrigation that is

Digital Turin For Smart Manufacturing	© 2023 Elsevier Inc.	
DOI: https://doi.org/10.1016/B978-0-323-99205-3.00007-9	All rights reserved.	135

managed by field sensors is crucial, as inadequate or excessive watering reduces crop output [7]. AI can improve the farming process in this context by collecting data on plant conditions and computing it with high performance and low cost [8], maintaining crop yield at normal standards, reducing water waste, and ultimately increasing the availability of potable water [6].

A digital twin model based on the IoT can be applied in farms to effectively recognize their current environment to capitalize on this worldwide concern. This means that a virtual representation of a farm should be able to behave depending on the systems' analyses and judgments as well as gather information from the farm. The fundamental creation of a digital twin for smart farming using the IoT to regulate an irrigation system based on a farmer's and/or AI choice is presented in this chapter. Section 2 of this chapter provides examples of IoT applications in agriculture, Section 3 introduces the idea of a digital twin in the context of agriculture, Section 4 details the creation of the digital twin, Section 5 describes the initial findings and analysis, and Section 6 concludes the chapter.

7.2 Internet of Things in agriculture

The majority of the literature on the development of IoT technologies in agriculture consists of exploratory research that demonstrate systems in small pilot projects [9]. The development of equipment and devices used on farms to gather information about the soil, crop quality, weather conditions, and other factors can be divided into two categories when it comes to the use of IoT in agriculture [10,11]. The second category includes the development of platforms that are used to store, organize, analyze, and visualize data to help in the decision-making [12,13].

The term "smart farming" appears while reading the literature on the application of information and communication technologies (ICT) in agriculture. Although the term "smart farming" has been in use for some time, there is still a need for a comprehensive definition of the term that encompasses the technology now employed in the agriculture industry. To integrate information and communication technologies in the cyberphysical farm management cycle, smart farming entails integrating them into machinery, equipment, and sensors [14,15].

Several technologies, including IoT, big data, AI, process management, etc., are perceived as being included in this notion. According to the literature, ICT usage in agriculture is a developing field that currently faces some challenges, but there are also many advantages associated with its use.

7.3 Digital twin smart farming

According to the research by [16], a digital twin model is one in which allows data transfer between a physical and digital entity automatically, as seen in Fig. 7.1. By utilizing technologies like big data, IoT, AI, etc., a digital twin is able to link information about the farm and business and is able to act depending on a choice made automatically by the system.

By extending the idea of smart farming, a digital twin for a smart farm or a digital smart farm is suggested. Building discrete services to understand the data of a specific system, such as an irrigation system, a seeding system, etc., and bringing them together in a cyberphysical system is how a digital smart farm is executed. This makes it possible to combine various systems and gives farmers a thorough understanding of how their crops are doing. It is possible to adjust the farm to changes in the environment, weather, markets, water limitations, etc., by employing a digital smart farm.

7.4 Digital smart farming system

This section, which is separated as follows, describes a digital smart farm based on two initiatives (The Sensing Change and SWAMP projects): The previous two projects are summarized in Section A, the system design



Figure 7.1 Digital twin concept [16],

is described in Section B together with its hardware and devices, and the system architecture is shown in Section C along with cloud services and other components.

7.4.1 Sensing change and smart water management platform programmers are related works

The Sensing Change project's approach was based on the idea that it could be applied to any type of agricultural activity and that it could make use of low-cost commercial tools and supplies [17]. Three key parts make up the system: the monitoring station, the smartphone app, and the cloud system (Fig. 7.2). This project created an information-gathering and information-analysis monitoring system for a farm. However, it was the farmers who made the decisions and took the necessary steps.

The SWAMP project, on the other hand, is creating a hands-on, IoTbased smart water management platform for precision irrigation in four locations across Brazil, Italy, and Spain [18]. The SWAMP platform can be designed and implemented in various ways, resulting in several SWAMP systems that are tailored to meet the needs and limitations of various settings, countries, climates, soils, and crops. This indicates that the decision-making process can be entirely handled by the farmer, by a machine, or by a combination of the two. According to Fig. 7.3, the SWAMP design is separated into five layers.



Figure 7.2 Sensing change system diagram [17].



Figure 7.3 Smart water management platform project architecture layers.

7.4.2 A system design

The system comprises a field-installed probe that measures ground temperature at 7 cm depth (DS18B20), soil moisture at depths of 7, 28, 50, and 72 cm, ambient light, and geographical position (CSMv1.2). The Raspberry Pi-3 module receives probe signals via the I2C bus (CSM v1.2 and BH1750), GPIO (DHT22), serial bus (Venus GPS), and One-Wire bus (DS18B20). a module for ADS1115 is also employed for the conversion of the CSMv1.2A/D signal. This data is read by the Raspberry module using a Python script, which displays numbers as percentages of soil moisture. The script then generates a payload including all probe data and delivers it to the Orion broker for subscription via the IoT agent. A prototype of the system utilized for laboratory testing is shown in Fig. 7.4. The final model of the probe that can be seen in the figure will be used in the field.

7.4.3 System architecture

Numerous equipment and systems, such as soil probes, weather stations, irrigation systems, seeders, harvesters, etc., are used on the farm. These tools and equipment are linked to the cloud through a gateway, which



Figure 7.4 System design.

communicates data to an IoT agent. The suggested system architecture is shown in Fig. 7.5, along with the services employed.

- A service called Fiware IoT Agent converts many communication protocols to the cloud-based standard.
- The term "fiware" Orion is a context broker that enables you to control every stage of the context information lifecycle, including updates, queries, registrations, and subscriptions.
- A document-based database called Mongo DB is used to store the most recent data in a complex structure.
- Draco is a generic enabler that is a different data persistence method for controlling context history.
- Time-series data are kept in MySQL, an open-source relational database management system.
- MySQL Grafana is an open-source analytics and monitoring tool for building data dashboards.

This digital environment is intended to visually enlighten the data gathered by IoT and to deliver data to the systems based on the decision-making



Figure 7.5 System architecture.

process carried out inside of it. Information can only currently be displayed visually on dashboards.

However, all of the other settings must be built in an integrated manner to properly construct the digital smart farm. The information gathered and analyzed in the cloud and represented digitally should be entered into the physical system via the cloud or by integrating programmable logic controllers into the machinery, equipment, and irrigation system.

7.5 First result and analysis

The system's initial testbed findings are shown in Fig. 7.6 in our laboratory. On the basis of the thermogavimetric data, the soil moisture sensor was calibrated [19]. This initial test shows that the probe can send data to the cloud and that it is possible to display that data in a dashboard that updates in real time. It is obvious that the air temperature and humidity have dropped abnormally, which suggests a hardware or communication issue that has to be further investigated.

Fig. 7.7 displays field data that was gathered using a nearby weather station. The Penman-Monteith equation can be used to extract the



Figure 7.6 Dashboard with probe data.



Figure 7.7 Field data collected.

reference evapotranspiration (ETo) using data from this meteorological station, including air temperature and humidity, wind velocity and direction, rainfall solar radiation, dew point, rain forecast, and rain chance [20].

The crop evapotranspiration is calculated by multiplying the ETo by the crop factor (Kc) for the specified crop. The ETo stands for the amount of water, in millimeter, that the reference crop uses [21].

The initial findings show that the technology is functional and capable of uploading data to the cloud for usage in spreadsheets and dashboards. Additionally, it is possible to display the water patterns in the soil and extract ETo using the nearby weather station. It is clear from Fig. 7.7 that during periods of drought, the soil moisture reduces and increases after each rainfall.

7.6 Conclusion

Farmers can connect various assets and systems utilizing the digital twin concept and IoT technology to have a better understanding of the various factors and elements that affect the farm's behavior, final yield production, and resource use. This essential component helps farmers make wiser decisions and minimize their influence on the environment's water, land, and soil resources. This chapter outlines the preliminary steps in creating a digital smart farm. However, numerous systems must work together to represent all of the farm's functions before a digital smart farm can be fully implemented.

According to this study, system architecture and cloud implementation are effective and may be utilized in the deployment of the subsequent steps, which include the creation of AI algorithms and other digital contexts. Once the complete system is operational, it will be feasible to comprehend how farms use resources and how that affects agricultural productivity. This promotes sustainable development and enhances food security for everyone on the planet.

References

- Goal 2: Zero Hunger, United Nations. http://www.un.org/sustainabledevelopment/ hunger/ [Online] (2020).
- [2] The State of Food and Agriculture: Leveraging Food Systems for Inclusive Rural Transformation. Food and Agriculture Organization of the United Nations. [Online]. Easily accessed at: http://www.fao.org/3/a-I7658e.pdf percent 0A percent 5Cn http://www.ncbi.nlm.nih.gov/pubmed/24897208, 2017, vol. 2, no. 7929.
- [3] AQUASTAT, 2016 http://www.fao.org/nr/water/aquastat/waterusage.
- [4] J. Panchard, S. Rao, T.V. Prabhakar, J. Hubaux, H. Jamadagni, Common sense net: a wireless sensor network for resource-poor agriculture in the semi-arid areas of developing countries, Inf. Technol. Int. Dev. 4 (2007) 51-67.

- [5] R. Gonzalez Perea, E. Cannacho Poyato, P. Montesinos, J.A. Rodriguez D'az, Optimization of water demand forecasting by artificial intelligence with short data sets, Biosyst. Eng. 177 (2019) 59-66.
- [6] M. Pulido-Velazquez, F.A. Ward, Water conservation in irrigation can improve water use, Proc. Natl Acad. Sci. U S Am. 105 (47) (2008) 18.
- [7] A.C. Charles, A.A. Namen, P.P.G.W. Rodrigues, Comparison of data mining methods applied to a surface meteorological station, RBRH 22 (0) (2017).
- [8] Proceedings of the International Conference on IoT in Social, Mobile, Analytics and Cloud, I-SMAC 2017, by S. Athani and C. H. Tejeshwar.
- [9] Internet of Things in agriculture, C. Verdouw, 9, volume 11, issue 035 of CAB reviews: perspectives in agriculture, veterinary science, nutrition, and natural resources [Online]. 2016. Available: http://www.cabi.org/cabreviews/review/20163379897.
- [10] R.R. Agale, Automated irrigation and crop security system in agriculture using Internet of Things, in: International Conference on Computing, Communication, Control and Automation (ICCUBEA), pp. 1-5, 2017.
- [11] K.L. Krishna, O. Silver, W.F. Malende, K. Anuradha, Internet of Things application for implementation of smart agriculture system, in: Proceedings of the International Conference on IoT in Social, Mobile, Analytics and Cloud, I-SMAC 2017, pp. 54–59, 2017.
- [12] S. Suakanto, V.J. Engel, M. Hutagalung, D. Angela, Sensor networks data gathering and job management for decision support of smart farming, International Conference on Information Technology, 2016 http://linkinghub.elsevier.com/retrieve/pii/ S2214317316301184.
- [13] P.P. Jayaraman, A. Yavari, D. Georgakopoulos, A. Morshed, A. Zaslavsky, Internet of Things platform for smart farming: experiences and lessons learned, Sensors 16 (11) (2016) 1–17.
- [14] D. Pivoto, P.D. Waquil, E. Talamini, C.P.S. Finocchio, V.F.D. Corte, Gd.V. Mores, Scientific development of smart agricultural technologies and their implementation in Brazil, Inf. Process. Agric. 5 (1) (2018) 21–32 [Online]. Available, Available from: https://linkinghub.elsevier.com/retrieve/pii/S2214317316301184.
- [15] S. Wolfert, L. Ge, C. Verdouw, M.J. Bogaardt, Big data in smart farming a review, Agric. Syst. 153 (2017) 69-80.
- [16] W. Kritzinger, M. Kamer, G. Traar, J. Henjes, W. Sihn, Digital twin in manufacturing: a categorical literature review and classification, in: IFAC-PapersOnLine, vol. 51, no. 11, pp. 1016–1022, 2018. [Online]. You can get it at: https://doi.org/ 10.1016/j.ifacol.2018.08.474.
- [17] R.F. Maia, A.L.H. Tran, Precision agriculture employing remote monitoring systems in Brazil, in: 2017 GHTC 2017 - IEEE Global Humanitarian Technology Conference, Proceedings, vol. 2017-Janua, pp. 1–6.
- [18] C. Kamienski, J. Soininen, M. Taumberger, S. Fernandes, A. Toscano, T.S. Cinotti, et al. Swamp: an iot-based smart water management platform for precision irrigation in agriculture, in: Global Internet of Things Summit (GIoTS), June 2018, pp. 1–6.
- [19] D.A. Robinson, C.S. Campbell, J.W. Hopmans, B.K. Hornbuckle, S.B. Jones, R. Knight, et al., Soil moisture measurement for ecological and hydrological watershed-scale observatories; a review, Vadose Zone J. 7 (1) (2008) 358–389.
- [20] R.G. Allen, L.S. Pereira, D. Raes, M. Smith, W. Ab.Guidelines for computing crop water requirements-FAO Irrigation and drainage document 56, Crop. Evapotransp., pp. 1–15, 1998.
- [21] J. Doorenbos, W. Pruitt, Guidelines for forecasting crop water requirements, FAO Irrig. Drain. Pap. 24 (1977) 144.

¹⁴⁴ Digital Twin For Smart Manufacturing
As per the new Semester-wise Syllabus of Gondwana University

MATHEMATICS

B.Sc. Semester V (CBCS)

DSE - III : Matrices and Theory of Equations
 DSE - IV : Special Relativity - I

- Dr. Ram Naresh B.S. Sisodiya
- Dr. Hariom Singh Tomar
- Dr. Vinod Kumar
- Vijay Soni
- Dr. Ajay Kumar Sharma Edited by
- Dr. Rishi Agrawal

Himalaya Publishing House ISO 9001:2015 CERTIFIED

As per the new Semester-wise Syllabus of Gondwana University

Mathematics

B.Sc. Semester V (CBCS)

DSE-III : Matrices and Theory of Equations DSE-IV : Special Relativity-I

Dr. Ram Naresh B.S. Sisodiya

Head, Dept. of Mathematics Sardar Patel Mahavidyalaya Chandrapur.

Dr. Vinod Kumar

Head, Dept. of Mathematics Chintamani College of Science Pombhurna, Dist. Chandrapur.

Dr. Hariom Singh Tomar

Head, Dept. of Mathematics Chintamani College of Arts & Science Gondpipari, Dist.-Chandrapur.

Vijay Soni

Head, Dept. of Mathematics D.B.Science College Gondia.

Dr. Ajay Kumar Sharma

Head, Dept. of Mathematics Mahatma Gandhi College of Science, Gadchandur, Dist.- Chandrapur

Edited by

Dr. Rishi Agrawal

Asso. Prof. and Head Dept. of Mathematics Hislop College Nagpur.



Himalaya Publishing House

ISO 9001 : 2015 CERTIFIED

AuHini

Annuar De part of the publication may be represented where in a relificative solution to the provident to any bern in by the moure declarate mechanics of plantation provide measuring and/or influences solitant the provident parameter of the publication

First Fidthan (9093

Published by	 Khin Khanan Printing for Ellimetryn Parlilladdig Elminn Pyl 1 Ed.; Ennichard Fre Diederies Many Engineer, Administ Able tele) Phintin 1633 3 ministra 3 diff Ellimetric Free 1633 3 diff (170 Phintin 1633 4 ministration freedo etc. Wintentin 37373 Languite Free
firench Offices Hew Delhi	- Thooja Aparhumula", A.H. Muran Lal Minni, Kuaan Huad, Iracza Daca, Huar Irada - Filebos
*******	 Planar 011 91970 019 91970 01970 01970 01000 010000 010000 010000 010000 Kundondal Chanalak Industrial Estatus, Ohnd Bond, Haujan - Afternik Planar, 0713 2721210, 2721210
Bangaluuu	Phillip 01 13, 201 Monthead, Sestindaparata Debud Halacaja Hardin, Dorgalara - Michae
	Photos Obt ATTANKET, Middle GETTORATET, BETTORATORA
Hyderabad	i No. 3.4. DOI: Dependent/, Desides Prophesendre Swancy Mathem, Kastegoria, Nydersbert, 566 037, Param. 646 27666641, 27556730
Ghennel	+ Dew Die Abis, Odd Die 2023, Octoord Elsen, Secondpopend Street, T. Heiper, Chemical Editer (
	Motole for BOARDA LO
Pune	Plakatur Apademint, Filal Flora, He. 627, Moldingana, Shurawaipani (Fisar Pratam Transa)
	Pum - 411 030 Phone 020 24406323, 24466333, Middle 60370576333
Lucknew	i - Thomseller 731, Maddagaan Gelony, Hoor NTE Gravinal School, Alignig, Locknews 226.627
	Phone 0532 4012353, Mobile 00307501540
Alumnilatiail	114, "SHAIL", IN Flore, Opp. Marthu Surbar House, C.O. Bond, Having Para.
	Alumentations - 380 000 Philam 070 26560126, Motale 05377086647
Gullack	E. Pid Hu St 755/4, Social 9, ODA Market Hagar, Gullack - 753 014, Onistia
	Medile (RE1387Ab007
Kolkala	3 106/4, Bolinghula Main Read, Boar II (Respital, Opp. 50) Back, Kelkata - 700.010
	Phone: 011 32449049, Mobile: 07439040301
DTP by	s 1091, Daypar (Proset)
Printed at	Enotarial Provident Chil, Hagpar, on behalf of FPH



ABOUT THE AUTHORS





By Ram Nameh Bill, Bloodba D presents, working B an Assister Perform and Head. Department of Mathematika shafe? And Mathematika Cam) Work. Chambergor (M13) - 642 402; He da bir Ph.D. In the year 2011 from 86 birth Antibackayalaya. Cam) Work, Chambergor (M13) - 642 402; He da bir Ph.D. In the year 2011 from 86 birth Antibackayalaya. Cam) Work, Chambergor (M13) - 642 402; He da bir Ph.D. In the year 2011 from 86 birth Antibackayalaya. Cam) Work, Chambergor (M13) - 642 402; He da bir Ph.D. In the year 2011 from 86 birth Antibackayalaya. Cam) Work, Chambergor (M13) - 642 402; He da bir Ph.D. In the year 2011 from 86 birth Antibackayalaya. Cam) Work, Chambergor (M13) - 642 402; He da bir Ph.D. In the year 2011 from 86 birth Antibackayalaya. Cam) Work, Chambergor (M13) - 642 402; He da bir Ph.D. In the year 2011 from 86 birth Antibackayalaya. Cam) Work, Chambergor (M13) - 642 402; He da bir Ph.D. In the year 2011 from 86 birth Antibackayalaya. Cam) Work, Chambergor (M13) - 642 402; He da bir Ph.D. In the year 2011 from 86 birth Antibackayalaya. Cam 80 birth Antibackayalaya. Cambergor (M13) - 640 birth Antibackayalaya. Cambergor (M13) - 640 birth Antibackayalaya. Cambergor (M13) - 640 birth Antibackayalaya. Cambergor (M13) from 2012 for presented of Mathematika, and Colego Colego. Cambergor (M13) from 2012 for presented of 2010 for a 2013 for presented of Mathematika, and Colego. Cambergor (M13) form 2012 for presented of Mathematika, from 0 for D.A. Index work and the 2010 for a 2013 for an employee for the Colego for

Dr. Viewel Kommer is Preservity Working at Post of Assistent Professor & Head Department of Mathematics at Chintermani College of Science. Portibutine, District-Chandrapar (MH) - 442918. He did his Pis.D. in 2013, From Dr. B. R. Ambeddar University, Agris (U.P.), India. He hist & year Teaching and research experience. His area (j special assion includes: Fluid Dynamics, Physiological Phild Dynamics, Operation Research, and Integral Equations He has Published. 12 Research Papers in Victors Journals and 12 Books, 3-cbooks, 4-Chapter in Books in Various National level Publications and One colard book in National Revel Publication



Vijay M. Bool (M.So., M.Bd.) is an Associate Professor and thead in Department of Mathematica, D.B. Scie College, Gonda with 29 years of teaching experience at U.G. Jevel. His interest lice in samplifying methamatica novice. "Students are the best outlets that a searcher moulds link responsible circuity.", is his conviction. "Theory relativity' is his form: He size has a knock in screeneplacing social wate through various HGOs.



Dr. Ajay Kamar Sharma is presently working as an Assistant Professor the and Head, Department of Mathematica, Mahama Gandhi College of Science, Cadchandar, Chandrapur (MH) - 442 908. He and int Ph.D. in the year 2017 from C.L. Juln (PG) College (Dr. B.R. Ambediar University), Efforated (U.P.). Trens, Dr. Sharma has 6 years of teaching experience and 12 years of research experience. The main areas of his research are Fluid Dynamics, Physiological Fluid Dynamics, and Operations Research. Dr. Sharma has published 12 research papers in various journals and proceedings, and 8 books in various National level publications. He attended and presented research papers in various national and international conferences.

Edited by



Dr. Rista K. Agrawai (M.Sc., NET, B.Ed., Ph.D.) is serving as Histop College, Nagsur as Professor and the Department of Mathematics with a teaching experience of 24 years at UC level. Sizec 2003, he has been worther a self-developed innovative teaching-learning method named 'GANEET DARSHAN'. With the same bits, he is a do on YouTube and having an Application on Play Store. Working for popularisation of Mathematics as a service to society with a strong belief that each student has his own mathematical laboratory on the form of his mind which a be opened by a teacher and let the student make proper use of it by thinking and visualizing mathematics area the teacher and the student make proper use of it by thinking and visualizing mathematics area the students make proper use of it by thinking and visualizing mathematics area the student with the second of the student's active and the student's association with Hindi Academy, Govt. of Delhi on International Teaching Diff of your Genber 5, 2021. them.

..... Jai Gancet

₹ 240/-

Contraction of the second seco



www.himpub.com

ISBN: 978-93-5840-105-9

PSM 0969



Ram Naresh Singh, Shailendra D. Deo and Ajay Kumar Sharma

for publication of a book chapter entitled, "A REVIEW OF THE INTERPRETATION OF DUALITY IN DIFFERENTIABLE AND NON-DIFFERENTIABLE MATHEMATICAL PROGRAMMING" in the book "Research Trends in Science and Technology Volume III (ISBN: 978-93-88901-74-1)" published by our publishing house in the month of July, 2023.



ISBN: 978-93-88901-74-1

RESEARCH TRENDS IN SCIENCE AND TECHNOLOGY VOLUME III

EDITORS DR. ALOK RANJAN SAHU MR. BASANT DESHWAL DR. NARAYAN GAONKAR DR. SANJAY SINGH



FIRST EDITION: JULY 2023

BHUMI PUBLISHING, INDIA

....

200

700 600 600

10

Research Trends in Science and Technology Volume III

(ISBN: 978-93-88901-74-1)

Editors

Dr. Alok Ranjan Sahu	

Department of Botany, Vikash Degree College, Bargarh, Odisha

Dr. Narayan Gaonkar

Department of Physics, University College of Science, Tumkur University, Tumkur, Karnataka

Mr. Basant Deshwal

Division of Nematology, Indian Agricultural Research Institute (IARI), New Delhi

Dr. Sanjay Singh

Department of Physics, Chintamani College of Arts and Science, Gondpipri, Maharashtra



July 2023

First Edition: July, 2023 ISBN: 978-93-88901-74-1



© Copyright reserved by the Editor

Publication, Distribution and Promotion Rights reserved by Bhumi Publishing, Nigave Khalasa, Kolhapur

Despite every effort, there may still be chances for some errors and omissions to have crept in inadvertently.

No part of this publication may be reproduced in any form or by any means, electronically, mechanically, by photocopying, recording or otherwise, without the prior permission of the publishers.

The views and results expressed in various articles are those of the authors and not of editors or publisher of the book.

Published by:

Bhumi Publishing,

Nigave Khalasa, Kolhapur 416207, Maharashtra, India

Website: www.bhumipublishing.com

E-mail: bhumipublishing@gmail.com

Book Available online at:

https://www.bhumipublishing.com/book/



PREFACE

In the ever-evolving landscape of science and technology, knowledge is a beacon that illuminates the path to progress. The pursuit of understanding and innovation has been the driving force behind the remarkable advancements that have shaped the world we live in today. As we embark on a new era, it becomes increasingly crucial to navigate through the diverse and dynamic currents of research to discern the trends that will define our future.

"Research Trends in Science and Technology" represents a collective effort to explore and elucidate the cutting-edge developments that are shaping the fields of science and technology. This book is an assembly of insightful chapters contributed by leading experts, researchers, and visionaries, all of whom share a common passion for unraveling the mysteries of the universe and harnessing the power of technology for the betterment of humanity.

In this volume, we delve into a broad spectrum of disciplines, ranging from fundamental sciences such as physics, chemistry, and biology to the transformative fields of artificial intelligence, nanotechnology, and biotechnology. By curating a diverse selection of research trends, we aim to showcase the interdisciplinary nature of modern scientific inquiry and the interconnectedness of technological breakthroughs.

We believe that knowledge should be shared and disseminated freely, fostering a collaborative spirit that transcends geographical and disciplinary boundaries. As such, "Research Trends in Science and Technology" serves as a platform for disseminating the latest discoveries, ideas, and perspectives that shape the course of human progress.

We extend our heartfelt gratitude to all the contributors who have dedicated their expertise and passion to enrich this compilation. Their invaluable insights and visionary outlooks have made this endeavor possible.

We hope that this book will inspire readers, whether they are students, researchers, policymakers, or curious minds, to embrace the spirit of inquiry and embark on their own explorations. By staying attuned to the latest research trends and leveraging collective knowledge, we can collectively chart a course towards a more sustainable, equitable, and innovative future.

Sr. No.	Book Chapter and Author(s)	Page No.		
1.	APPLICATIONS OF ARTIFICIAL INTELLIGENCE IN	1 - 12		
	BIOTECHNOLOGY			
	Bindu Rajaguru, Meenakshi Johri and Bhakti Gavhane			
2.	NUCLEOLUS METHOD IN A LIBERALIZED ELECTRICITY	13 - 32		
	MARKET			
	Murali Matcha, Siddheswar Kar and Neha Verma			
3.	SOCIAL WELFARE MAXIMIZATION IN A RESTRUCTURED	33 - 53		
	POWER SYSTEM USING BAT ALGORITHM			
	Murali Matcha, Siddheswar Kar and Neha Verma			
4.	REVIEW ON APPLICATION PROGRAMMING INTERFACES	54 - 63		
	FOR INTERNET PROTOCOL			
	Sumit Chopra, Lovedeep Kaur,			
	Anchal Nayyar and Gagandeep Singh			
5.	PROCESS MANAGEMENT IN OPERATING SYSTEM	64 - 71		
	Sumit Chopra, Diksha Rani, Rajesh Sharma and Simranjot Kaur			
6.	INTRODUCTION TO SENSORS	72 – 75		
	M. W. Bhade			
7.	PROTEIN ENGINEERING FOR PRODUCTION OF	76 - 83		
	INDUSTRIAL IMPORTANT ENZYMES			
	Kirti Yadav, Neeraj K. Aggarwal and Awtar Singh			
8.	BIOMATERIALS	84 - 102		
	D. R. Nagapure			
9.	A REVIEW OF THE INTERPRETATION OF DUALITY IN	103 - 108		
\checkmark	DIFFERENTIABLE AND NON-DIFFERENTIABLE			
	MATHEMATICAL PROGRAMMING			
	Ram Naresh Singh, Shailendra D. Deo and Ajay Kumar Sharma			
10.	CELLULAR COMMUNICATION IN BREAST CANCER	109 - 117		
	Juily R. Atitkar			

TABLE OF CONTENT

A REVIEW OF THE INTERPRETATION OF DUALITY IN DIFFERENTIABLE AND NON-DIFFERENTIABLE MATHEMATICAL PROGRAMMING

Ram Naresh Singh^{*1}, Shailendra D. Deo² and Ajay Kumar Sharma^{*2}

¹Department of Mathematics,

Sardar Patel Mahavidyalaya Chandrapur - 442402 (MS), India ²Mahatma Gandhi College of Science,

Gadchandur, Dist - Chandrapur-442908(MS), India

*Corresponding author E-mail: drrbs79@gmail.com, ajaykumarsharma101010@gmail.com

Abstract:

The existence of an ideal solution to either the primal problem or the dual problem ensures the existence of an ideal solution to the other because these two projects are linked by the duality principle. If the primary issue is constrained minimization (or maximisation), then the dual issue is a constrained maximisation (or minimization) problem. The duality effects have shown to be very helpful in enhancing numerical techniques for addressing particular categories of optimisation problems. Because it offers appropriate halting guidelines for primary and secondary problems, The presence of duality theory in nonlinear programming problems makes it easier to create numerical algorithms. The inclusion of duality theory in nonlinear programming issues aids in the development of numerical algorithms. If the original issue is the dual of the dual, then a nonlinear programming problem and its dual are said to be symmetric. This postulation's major objective is to take into account optimality and duality in a range of mathematical programming issues, with a particular emphasis on non-differentiable nonlinear programming and variational issues, such as blended sort symmetric and self duality. Nondifferentiable fractional minmax programming, continuous-time minmax programming, minimaxvariational problems, and continuous-time minmax programming.

Overview

In the late 1940s, mathematical programming achieved the status of a logical science unto itself, and ever since then, it has made enormous strides. It is currently regarded as one of the most dynamic and invigorating disciplines of contemporary mathematics, with numerous applications in a variety of contexts, including design, financial matters, and basic sciences. Mathematical programming problems frequently include finding the least-weight design of a structure that is bound to stress and deflection restrictions.

A mathematical programming problem has the following structure: (MP): Maximize or minimize f(x). Depending on $g_i(x) \le 0, i = 1, 2, 3, m$ $h_j(x) = 0, j = 1, 2, 3, k,$ $x \in X$

The function f and each of its f_j and h_j on the n-dimensional Euclidean space R_n and XR_n are real valued capabilities. This is referred to as the general mathematical programming issue. Fairness constraints are defined as $h_j(x) = 0, j = 1,2,3 \dots, k$, whereas inequality constraints are defined as $g_i(x) = 0, i = 1,2,3 \dots, m$. The integration of x X is a conceptual constraint. We define the previous issue as a differentiable programme if the objective and critical skills are differentiable. The aforementioned issue is referred to as a convex programming problem if X is a convex set and the objective and inequality constraints are both relative to capacity.

Duality in adaptable mathematical programming

If f: Rn R and hj: Rn R, where j = 1, 2,..., m, then think about the nonlinear programming problem:

Min f(x)(P) according to, $h_i(x) = 0, j = 1, 2, 3 m$.

The Lagrangian dual for issue (P) in the case of Rm is defined as

This is, M inf (u) plus Th(u) (LD)

If all of the functions f and $h_j(x) = 0, j = 1,2,3 \dots m$. are differentiable convex functions, then the problem (LD) is similar to the problem below:

Maxf(x) + Th(x) (WD)

Depending on Vf(x) plus Th(x) = 0, 0 and R_m , The Wolfe type of dual for the issue (P) is superior than everything else here. When Mangasaria gives an example, he suggests that some duality theorems might not hold true if the goal function or the constraint function is a summed up convex function. This inspired Mond and Weir to create a new dual for (P) as a result.

Maxf(x)+ (MWD)

Depending on

 $f(x) + \lambda Th(x) = 0, \lambda Th(x) \ge 0, \lambda \ge 0, \lambda \in R_m,$

Non-differentiable mathematical programming duality

The class of non-differentiable mathematical programming issues that Mond took into account was as follows:

Min f(x) + xTBx 1/2 (NP)

depending on $h_i(x) \le 0, j = 1, 2, 3 \dots m$.

In this case, B is a n X n positive semi definite (symmetric) matrix, and f and $h_j(x), j = 1,2,3...,m$. are twice differentiable functions from Rn to R. The functions f and hj (j = 1, 2,...,

m) are anticipated to be convex functions. They established a duality theorem between (*NP*) and the following problem.

(*ND*): [f(u) + y Th(u)] maxf(u) + y Th(u)] - uT according to,

 $yT + f(u) + h(u) + Bw = 0, wTBw \ 1, y \ge 0.$

A new dual program was introduced by Chandra, Craven, and Mond along the lines of Mond and Weir:

Maxf(u) - [uTf(u) + yTh(u)] (NWD)

depending on Bw = f(u) + yTh(u)

 $yTh(u) \ge 0, wTBw \le 1, y \ge 0.$

Additionally, for any possible solution to (NP) and (NWD), settled duality theorems by assuming that f(u) + wTBw is pseudoconvex and that yTh is quasiconvex.

Later, Mond and Schechter replaced the square root term with the standard term and, in addition, defined the non-diffemtiable nonlinear programming problems as follows:

(NP)1 Min ||Sx||p + f(x)

according to, $h_j(x) \le 0, j = 1, 2, 3 \dots m$.

Here, the convex functions f and h_j $(j = 1,2,3 \dots m)$ are twice differentiable from Rn to R. The problem is the dual for (NP)1:

Maxf(u) + [yTh(u) + uTS Tv (NP)]

according to $F(u) + Y Th(u) = ||v||, q \le 1, y \ge 0.$

where conjugate instances for p and q are.

Symmetric duality in distinctive mathematical programming

Take a look at a capacity f(x, y) that can be differentiated in the ranges Rn and Rm. The challenges that Dantzig et al. introduced are as follows:

$$Min(f(x, y) - yTy f(x, y) (SP))$$

Depending on

 $yf(x, y) \le 0$, $(x, y) \ge 0.Maxf(x, y) - xTf(x, y)$ (SD) Depending on Vxf(x, y) > 0, (x, y) > 0.

Additionally, the existence of a typical optimal solution to the primal (SP) and (SD) was shown when (i) f is convex in x for every y and concave in y for every x, and (ii) f, which is twice differentiable, has the property that its matrix of second partials for y is negative definite at (x_0, y_0) and also provided the information below regarding symmetric dual programming issues: Min(f(x, y) (MSP) yTy f(x, y))depending on $f(x, y) \le 0, x \le 0$. Maxf(x, y)(MSD)x Tf(x, y)depending on $xf(x, y) \ge 0, y \ge 0$. It should be noted that whereas in both (SP) and (SD) the constraints include x 0 and y 0, only x 0 is necessary in the primal and only y 0 is necessary in the dual.

Hypothetical duality in non-differentiable mathematical programming

Suppose f(x, y) =be a true esteemed continuously differentiable function in the intervals x Rn and y Rn. Chandra and Husain created the following set of symmetric dual nondifferentiable algorithms to demonstrate how duality is a result of the convexity-concavity assumption applied for the bit function f(x, y) =:

 $Min(f(x, y) = yTyf(x, y) + (xTBx)_{1/2}$

depending on $Cw - yf(x,y) \le 0$, $wTCw \le 1$, (x,y) > 0.

Maxf(x, y) - xTx, f(x, y) - (yT Cy)1/2 (ND)

depending on $f(x, y) - Bz \le 0$, $zTB z \le 1$, $(x, y) \ge 0$.

n x m and m x m positive semi-definite matrices, respectively, make up B and C.

Chandra, Craven, and Mond showed another combination of symmetric dual nondifferentiable algorithms by reducing the convexity criteria on the bit function f(x,y) to pseudoconvexity.

The issues in are as follows:

Min f(x, y) + (xTBx)1/2 - yTCz (PS)

depending on Cz = yf(x,y) = 0. When yT[y f(x,y) - Cz] > 0, $zTCz \le 1$, $x \ge 0$.

Max f(x, y) - (yT Cy)1/2 + xTBw (DS)

depending on $Bw + xf(x, y) \le 0$, If xT[xf(x, y) + Bw] is zero, wTB $w \le 1, y \ge 0$.

Programming with differentiable fractions

Assume that $f_{,-g}$ and h_{j} (J = 1, 2, and m) are real, esteemed, differentiable convex functions defined on the open convex set X Rn. Consider the convex-concave fractional programming issue at this point. (FP): Depending on

$$h_j \leq 0, (j = 1, 2, ..., m)$$

S = x X:hj(x) 0; j = 1, 2,..., m; g(x) > 0 for every x X; and if g(x) is not

Affine, then for all x X, $f(x) \leq 0$.

There are two prominent duality models for (FP), and these have been extensively discussed in the work. These are a result of Schaible, Jagannathgan, and Bector. The issue is the Bector's double [17] for (FP).

according to,

The accompanying proportionate problem (EPF) of FP is essentially represented by this Wolfe double:

EFP: Minimum according to,

In (BD), one needs that $f(u) + yTh(u) \ge 0$, if g isn't relative, in order to make the target function pseudoconvex and ensure that duality theorems are true.

Programming that is not differentiable in fractions

Mond thought about the following fundamental issue according to,

 $h(x) \le 0.,$

In this scenario, G(x) > 0 for every possible x, B and D are n x n symmetric positive semidefinite frameworks, f, g, and h are differentiable functions from Rn into R, Rm separately. Mond [9] developed suitable duality theorems under convexity assumptions on f, g, and h, described a twofold problem to the (NFP) issue, and proved necessary and sufficient conditions for the existence of an ideal solution.

After some time, Zhang and Mend found a number of conditions that had to exist in order for the optimal configuration of (NFP) to exist. Additionally, as an application of these optimality requirements, the first and second request duals indicated below were derived individually, and relevant duality theorems were also produced.

Diverse problems

A variation problem is a particular example of an optimum control problem where the state function is subordinate to the control function.

In terms of numbers, the following variational problem exists:

Depending on

The separation operator D is given by $y = Dx \Rightarrow x(t) = x(a) + (s)ds$ except at discontinuities, and C (I, Rn) is the space of persistently differentiable functions x: I Rn. Where f:I x Ru x R n Rm and g:I x Ru x Rn Rm are persistently differentiable functions as for each of their contentions. Valentine establishes the necessary circumstances for the presence of an external force (VP).

Final summary

But in the current context, the heightened competitiveness and customer expectations frequently require the greatest solutions, not just practical ones. It has been found that even shortening the time it takes to improve can help a corporation save money by enhancing the design process. In this manner, the optimisation theory regulates selecting the best option among a few alternatives in terms of given capacity with the minimum amount of resources that is theoretically feasible. Mathematical programming problems are a new class of problems as a result. The most effective approaches for solving problems are frequently applied in operations research and are known as mathematical programming techniques.

References

- S. K. Mishra (2010). Second Order Mixed Symmetric Duality in Non-Differentiable Multi-Objective Mathematical Programming, Journal of Applied Analysis, Volume 13, Issue 1, Pages 117–132
- Yadvendra Singh, S. K. Mishra, and K. K. Lal (2017). Optimality and duality for nonsmooth semi-infinite multiobjective programming with support functions, Yugoslav Journal of Operations Research 27: Number 2, 205-218.
- 3. S. K. Mishra & M. Jaiswal (2015), Optimality Conditions and Duality for Semi-Infinite Mathematical Programming Problem with Equilibrium Constraints, Numerical Functional Analysis and Optimization, 36:4, 460–480, DOI: 10.1080/01630563.2015.1013552,
- Shrivastav, Santosh Kumar (2012). Duality Gap in Nonlinear Programming, International Journal of Engineering Research & Technology (IJERT), Vol. 1 Issue 6, August - 2012, 1-13
- 5. Ivanov, V.I. (2013). Duality in nonlinear programming. Optim Lett 7, 1643–1658. https://doi.org/10.1007/s11590-012-0512-6
- 6. V. Jeyakumar and G.Y. Li (2014). Optimization, 63:5, 713–733, DOI: 10.1080/02331934.2012.690760
- Dimitris Bertsimas, Frans J. C. T. de Ruiter (2016) Duality in Two-Stage Adaptive Linear Optimization: Faster Computation and Stronger Bounds. INFORMS Journal on Computing 28(3):500-511.http://dx.doi.org/10.1287/ijoc.2016.0689

ISBN: 978-93-88901-74-1

RESEARCH TRENDS IN SCIENCE AND TECHNOLOGY WOLUME III

EDITORS DR. ALOK RANJAN SAHU MR. BASANT DESHWAL DR. NARAYAN GAONKAR DR. SANJAY SINGH



FIRST EDITION: JULY 2023

BHUMI PUBLISHING, INDIA

....

700 600 500

10





AIJR Abstracts

Page No-19

Abstracts of the International Conference on Recent Trends in Mathematics and Computer Science 2023

Editors

- V. Santiagu Theresal
- S. Arul Amirtha Raja
- Joseph Clement

Organized by Department of Mathematics and Computer Science, Auxilium College of Arts and Science for Women, Regunathapuram, Tamil Nadu, India



ICRTMCS-2023 19-21 October 2023



Series: AIJR Abstracts

Browse AIJR Abstracts Series at-

https://books.aijr.org/index.php/press/catalog/series/abstracts

Page No-21

V. Santiagu Theresal S. Arul Amirtha Raja Joseph Clement (Editors)

Abstracts of the International Conference on Recent Trends in Mathematics and Computer Science 2023 ICRTMCS-2023 (19-21 October 2023)

Organized by

Department of Mathematics and Computer Science, Auxilium College of Arts and Science for Women, Regunathapuram, Tamil Nadu, India

> *Published by* AIJR Publisher, Dhaurahra, Balrampur, India 271604



Abstracts of the International Conference on Recent Trends in Mathematics and Computer Science 2023 ICRTMCS-2023 (19-21 October 2023)

Editors

Sr. Dr. V. Santiagu Theresal Auxilium College of Arts and Science for Women Regunathapuram, India.

Dr. S.Arul Amirtha Raja St. Joseph's College of Engineering OMR, Chennai, India.

Dr. Joseph Clement Department of Mathematics School of Advanced Sciences VIT, Vellore, India.

Conference Organizer Departments of Mathematics and Computer Science, Auxilium College of Arts and Science for Women, Regunathapuram, Tamil Nadu, India

Conference Venue Hybrid (Online & Auxilium College of Arts and Science for Women, Regunathapuram, India)

ISBN: 978-81-965621-0-6 DOI: https://doi.org/10.21467/abstracts.158

Type Abstract Book Series AIJR Abstracts

Published 19 October 2023

Number of Pages 96

Copyeditor Augustine Gomes

Imprint AIJR Books

© 2023 Copyright held by the author(s) of the individual abstract. Abstracts are not considered as pre-publication and hence author(s) are free for subsequent publications. Abstracting is permitted with credit to the source. This is an open access book under Creative Commons Attribution-NonCommercial 4.0 International (CC BY-NC 4.0) license, which permits any non-commercial use, distribution, adaptation, and reproduction in any medium, as long as the original work is properly cited.

Published by



AIJR Publisher, Dhaurahra, Balrampur, India 271604

Disclaimer

This conference book of abstracts has been produced using author-supplied copy via conference organizer. Editing has been restricted to some formatting and style where appropriate. No responsibility is assumed by the publisher or conference organizer for any claims, instructions, methods, diagnosis, treatment, or policy contained in the abstracts. It is recommended that these shall be verified independently. Further the conference organizer & AIJR publisher both remain neutral concerning jurisdictional claims in published maps and institutional affiliation.

Page No-25

ICRTMCS-2023

Table of Contents

Disclaimeri
Forewordix
About ICRTMCS - 2023x
Message from the Secretaryxi
Message from the Principal xii
Message from the Convenerxiii
ICRTMCS-2023 Committeesxiv
Improving the Sensor Energy Consumption Using IWC Algorithm U. Chinnadurai and A. Vinayagam1
Detection of Pelvic Inflammatory Disease Using CNN Segmentation Algorithm S. Padmavathi
A Bi-Stage IG - PCA Dimensionality Reduction Approach for the Diagnosis of Pandemic Diseases from Clinical Attributes Preethi K. and Ramakrishnan. M
Comprehensive Analysis of Machine Learning Algorithms for Crop Disease Prediction V. Praba and K. Krishnaveni
Join Query Optimization Using Firefly Algorithm and Hybrid DQN-Firefly and DDQN-Firefly Approaches M. P. Karthikeyan and K. Krishnaveni
Ontology Based Knowledge Extraction on Diabetic Patients and Drugs from EHRS K. Pushpavath and K. L. Shunmuganathan
Future Adaptive and Reliable Vehicular Networks toward 6G Using Machine Learning Strategies Bharathi S. and Durgadevi P
A Survey of Detection and Classification of Rice Leaf Disease Using Image Processing R. Dhivya and N. Shanmugapriya
A Comparison of Image Denoising Techniques Using Traditional Filters and Deep Learning Amal Kumar, Piyush Kumar Singh, Jainath Yadav
Artificial Intelligence and Machine Learning Solutions for Controlling Pollution in Air Vansh Arora
Natural Image Generation Systems Using Generative Adversarial Networks in Captcha N. Sengamala Selvi
Cyber Security Facing Threats and Ways to Overcome These Problems by Using Two Fish Techniques D. Keerthi

Fruitful Wave Effect Supply Chain Management on Big Data by Improving Demand Indicator Forecasting Surya V
Security Threats in Cloud Computing-data Breaches M. Sowmiya Saliq14
Healthcare Image Fusion Based on the Dragon Cluster Method and Enhanced Vision Decomposition V. Sheela Angel Harish
Cyber Security to Prevent Cyber Threat Using Triple DES Algorithm <i>B. Roobashri</i>
Improving Security of Customer Personal Identifiable Information (PII) on IoT Enabled Mobile Applications R. Allirani
Analysis of Consumer Preference of Agricultural Products in e-Commerce Website
Pagidela Siva Krishna Chaitanya Reddy, Prahadeeswaran M., Pragathish V
Agricultural Commodity Market Analysis and Price Forecasting Pragathish V., Prahadeeswaran M., Sandhiya R19
Cloud Computing and Security Issues in the Cloud S. Renilda20
Speaker Identification in Emotional Environment: A Review
Debasis Mohanta and Jainath Yadav21
Spectrum Sensing Techniques in Cognitive Radio Networks: A Review D. Seema Dev Aksatha and R. Pugazendi
A Game Theory Method to Cyber-threat Information Sharing in Cloud Computing Technology L. M. Muthulakshmi
Technical Perspective: Prediction of Cyber Threats and Attacks Using Machine Learning Algorithms C. Bala Murugan and R. Sri Devi
Study the Flow of a Newtonian Fluid in a Cylinder with a Peripheral Layer Using Numerical Methods
Ram Naresh Singh B. S. Sisodiya25
Analyzing the Bounds of Transparency Order of Boolean Functions Mayasar Ahmad Dar and Deepmala Sharma26
Hall Effects on MHD Rotating Flow Past an Exponentially Accelerated Inclined Plate with Variable Temperature and Uniform Mass Diffusion Loganathan S., Thamizhsudar M., Muthucumaraswamy R.
A Mathematical Approach of General Theory of Relativity Dudheshwar Mahto
Continuous Maps in Ideal Topological Spaces
O. Uma Maheswari and A. Balraj29
Enhanced Semi-local Convergence Analysis for Higher-order Iterative Techniques Jinny Ann John and Jayakumar Jayaraman

Radio Mean and Radio Antipodal Mean Labelling of Circulant Graphs G(4K + 2,1,2) <i>R. Gomathy and T. Arputha Jose</i>	31
Markovian Jumping Parameters for Takagi-Sugeno Fuzzy Delayed Neural Networks T. Aparna and K. Asmiya Banu	32
On the μF-subgroups of Some Finite Abelian Groups Shailesh Singh, Vilas Kharat, Manish Agalave	33
On Neutrosophic Resolvable Sets and Functions in Neutrosophic Hyper Connected Spaces M. Deepa and D. Sasikala	34
The Analysis of the Consumer Preferences and Behavior of Electric Vehicles and Fuel Vehicles <i>M. Deepa, P. Nandhini Sri, M. Pooja, D. Reshmitha</i>	35
Exploring the Reliability of the System in Various Configurations with Different Components Nikhil Vikas Jaipurkar and Dr. Ram Naresh B. S. Sisodiya	36
Supra b [*] - Closed Sets in Supra Topological Spaces <i>C. R. Parvathy and S. Deepa</i>	37
Earth-shattering Role of Statistics in Computational Sciences Sonia Sharma, Manmeet Kaur, Jaspreet Kaur, Arshpreet Kaur	38
Co-isolated Locating Dominating Sets on Subdivision Graphs Meenal N	39
Nonlinear Fredholm Integral Equation via Fixed Point Theorems on Convex Contraction in Multiplicative Metri Spaces A Mary Priva Dharsini and P. Joselin Lavino	ic 10
Total Eccentric Domination Number of Some Special Types of Graphs J. John Flavia	11
M-Polynomial Approach to Compute Molecular Descriptors of Conjugated Borane Dendrimer Theertha Nair A., D. Antony Xavier, Annmaria Baby, Akhila S4	12
A Study on Covering Extended Distance Energy of Some Graphs Y. Yogalakshmi, U. Mary, S. Sreeja4	13
Effects of Reversible Chemical Reaction on Boundary Layer Flow R. Umadevi, D. Arivukkodi, K. Vijayalakshmi4	14
Computation of Topological Descriptors of Boron Nanosheet D. Antony Xavier, Akhila S., Annmaria Baby, Theertha Nair4	15
Cost Analysis on a Fuzzy Inventory Model with Ranking of Circumcenter of the Centroid of Pentagonal Fuzzy Number	16
A Fuzzy Multi-item Displayed Inventory Model with Generator Cost Using Trapezoidal and Pentagonal Fuzzy Number under Space Constraint	10
к. к. зититуи	+/

Page No-29

Ranking Fuzzy Numbers Using a Satellite Town Scheme and a Distance Method S. Priyanka and A. Leema Rose	48
Study a Degree Sum Condition with Connectivity for Relative Length of Shortest Paths and Cycles Arockia Jancy K.	49
Algebraic Structures and their Applications in Coding Theory and Cryptography C. Senthilnathan and S. Karunanithi	50
Optimal Results for Curse of Dimensionality Problems Using Orthogonal Transformation T. R. Thirumavalavan and S. Karunanithi	51
Towards Sustainable Development in the EGQ Model with Constraints under Demand Uncertainty P. Janavarthini and I. Antonitte Vinoline	52
Programme for Vertex Coloring for Complement Fuzzy Graph Using $lpha$ -Cut R. Parimaleswari and A. Leema Rose	53
Bond Additive Descriptors of Tetrapyrrolic Macrocyclic Nanostructures J. Singh Junias and Joseph Clement	54
Isometric Cycle Transversal in Graphs and Networks S. Arul Amirtha Raja and V. Santiagu Therasal	55
On Topological Entropies of Zeolite Frameworks and its QSPR Modelling with Molecular Interactions Kavin Jacob and Joseph Clement	56
Entropy Descriptors on Predictive QSAPR Modelling of Zeolite Structures Pancras Peter and Joseph Clement	57
QSPR Analysis and Complexity Studies of Carbon Allotropes Using Topological Indices and Information Entropies <i>M. P. Rahul and Joseph Clement</i>	58
Rough (1,2,3)* Pre Generalized Closed Sets in Rough Tritopological Space K. Shakilaja and K. Bhuvaneswari	59
Modelling with Fractional Order Continuous and Discrete (Delta) Integration M. Sebasti Jeya Pushpam	60
On Computation of Topological Descriptors for Symmetrically Configured Pentagonal Structure of Carbon Nanocones	61
2-Shortest and Strong Shortest Path Union Cover for Networks Santiagu Theresal and S. Arul Amirtha Raja	62
Queuing System Using Fuzzy Graph for Social Distance in a Hospital P. Ilakkiya	63
Applications to Differential Equations of Fixed Point Theorems in b-Multiplicative Metric Spaces A. Mary Priya Dharsini and J. Jarvisvivin	64

Double Domination Number of Some Graphs
W. Suktinunnuluksinni
P4 – Decomposition in Boolean Function Graph B(KP, L(G), INC) of a Graph S. Dhanalakshmi
Fundamental Theorems in Discrete Fractional Calculus Using Nabla Operator <i>M. Sebasti Jeya Pushpam</i>
An Induced P ₃ -packing k-partition Number for Benzenoid System Santiagu Theresal and S. Arul Amirtha Raja
A Mathematical Model Based on the Chi-Square Distribution for the Impact of Psychological Stress on Gene Activity of the Clock-related Genes hPER1 and hPER2 Among Humans C. Senthilkumar and A. Leema Rose
Pricing and Coordination Policy for a Three Layer Integrated Supply Chain with Linear Declining Demand and Suggested Selling Price by Manufacturer Atmaram Nigwal
A Note on Subgroups of the Symmetric Group S5 and Some Lower Intervals of L (S5) <i>R. Sathiyapriya and Sathesh Kumar N</i> 71
A Mathematical Model for Lowering Cortisol Rates in Affected Persons with Comorbid Post-traumatic Stress Syndrome Using Log-normal Distributions A. Leema Rose
A Mathematical Model of Naturalistic Light's Effect on Diurnal Plasma Melatonin and Serum Cortisol Concentration Using the Exponential Distribution A. Leema Rose and A. Manickam
Contra (Ti, Tj) Z α -continuous and Open Mappings via (Ti, Tj) Z α -open Sets in Bitopological Spaces O. Uma Maheswari and S. Yesurani
Fuzzy Logic in the Medical Field Using Decision Making Process <i>M. Vijaya and V. Arivazhagan</i> 75
Fuzzy AHP Approach Using Trapezoidal Fuzzy Number in Irrigation of Water M. Anusiya and G. Sivakumar76
Hop and Perfect Domination Numbers of Bloom Graph S. Kulandai Therese and Andrew Arokiaraj77

•

AIJR Abstracts

Abstracts of the International Conference on Recent Trends in Mathematics and Computer Science 2023

Editors

- V. Santiagu Theresal
- S. Arul Amirtha Raja
- Joseph Clement

Organized by Department of Mathematics and Computer Science, Auxilium College of Arts and Science for Women, Regunathapuram, Tamil Nadu, India



ICRTMCS-2023 19-21 October 2023

To learn more about AIJR Publisher Please visit us at: www.aijr.org





Study the Flow of a Newtonian Fluid in a Cylinder with a **Peripheral Layer Using Numerical Methods**

Ram Naresh Singh B. S. Sisodiya

Department of Mathematics, Sardar Patel Mahavidyalaya, Chandrapur, India *Corresponding author: drrbs79@gmail.com

ABSTRACT

Peristaltic flows are beneficial for resolving problems with physiological flows. In order to represent a variety of fluid motions involving peristaltic, a number of problems involving distinct fluid behavior assumptions and geometrical configurations have been solved. This work considers the passage of a Newtonian fluid through a cylindrical tube in the presence of a peripheral layer of a Newtonian fluid with a different viscosity. It is determined what the relationship between flow rate and pressure difference is. In this paper, the trapping and reflux limits, as well as the pumping efficiency, are determined. The problem is analyzed numerically, and the results for peripheral layer viscosity, pressure difference, and pumping efficiency are obtained.

Keywords: Blood Flow, Artery, Arteriosclerosis



©2023 Copyright held by the author(s). Published by AIJR Publisher in "Abstracts of the International Conference on Recent Trends in Mathematics and Computer Science 2023" (ICRTMCS-2023), 19-21 October 2023. Organized by the Department of Mathematics and Computer Science, Auxilium College of Arts and Science for Women, Tamil Nadu, India. DOI: 10.21467/abstracts.158

Exploring the Reliability of the System in Various Configurations with Different Components

Nikhil Vikas Jaipurkar^{1*} and Dr. Ram Naresh B. S. Sisodiya²

¹Department of Mathematics, Institutions of Higher Learning, Research and Specialised Studies,

Sardar Patel Mahavidyalaya, Chandrapur, M. S., India

²Department of Mathematics, Sardar Patel Mahavidyalaya, Chandrapur, M. S., India *Corresponding authors: nikhiljaipurkar9@yahoo.com, drrbs79@gmail.com

ABSTRACT

Our study is a comprehensive exploration of complex system reliability across various configurations, encompassing diverse sets of components. We sought to gauge the dependability of these systems, an essential facet in industries, infrastructure, and technology, where real-world conditions and configurations vary. Our approach involved meticulous analysis, breaking down the systems into their constituent components. We calculated individual reliabilities for components operating in both series and parallel configurations. This dual focus allows us to understand the critical interplay between component reliability and system design. In series, each component is vital; in parallel, redundancy offers a safety net. The paramount goal was to establish a benchmark for evaluating system reliability in diverse real-world scenarios, enabling engineers and designers to make informed decisions about system configuration and maintenance. We presented our findings graphically, providing a clear, comparative view of system reliability across different setups. In summary, our research contributes to the foundational understanding of complex system reliability, furnishing valuable insights for industries, infrastructure, and technology. By creating a visual roadmap for system performance under varying conditions, we equip decision-makers with the tools to build more dependable, resilient, and robust systems in an ever-evolving technological landscape.

Keywords: Reliability, Failure Rate, Reliability Block Diagram



©2023 Copyright held by the author(s). Published by AIJR Publisher in "Abstracts of the International Conference on Recent Trends in Mathematics and Computer Science 2023" (ICRTMCS-2023), 19-21 October 2023. Organized by the Department of Mathematics and Computer Science, Auxilium College of Arts and Science for Women, Tamil Nadu, India. DOI: 10.21467/abstracts.158



ISBN: 978-93-91535-91-9

GLOBAL CHALLENGES AND SOLUTIONS FOR SUSTAINABLE DEVELOPMENT CERTIFICATE

INTERNATIONAL CONFERENCE ON BASIC SCIENCES:

This is to certify that

Ram Naresh B.S. Sisodiya

has participated and presented his/her paper titled

Analytical study of Reliability in the system of various configurations with different components by using Reliability Block Diagram

during the INTERNATIONAL CONFERENCE GCSSD-2K23 held on 13-14 October 2023

Organised by

Chintamani Education Society, Ballarpur's

CHINTAMANI COLLEGE OF SCIENCE, POMBHURNA, DIST. CHANDRAPUR, M.S.

In Collaboration With Gondwana University, Gadchiroli

C4091

Venue: VASANT-VAIBHAV HALL

CHINTAMANI COLLEGE OF SCIENCE, POMBHURNA, DIST. CHANDRAPUR, M.S

likes

Dr. Rajiv G. Weginwar Principal and convener

Dr. Anil Chitade Convener

Dr. Sudhir S. Hunge Co-Convener

Page No-34

Dr. Anont S. Deshponde Organizing Secretary (IQAC - Coordinator)

Women in Indian Society

Edited By

Ms. Meghavee G. Meshram

Assistant Professor, Dr. Babasaheb Ambedkar College of Social Work, Morane Dhule, Maharashtra, India

Mr. Naresh W. Patil

Assistant Professor, Kumbhalkar College of Social Work Wardha, Maharashtra, India



H. S. R. A PUBLICATIONS

Published by

HSRA Publications 2023 #02, Sri Annapoorneshwari Nilaya, 1st Main, Byraveshwara Nagar, Laggere, Bangalore – 560058 Ph- 7892793054

Sales Headquarters - Bangalore

Copyright © AUTHORS 2023

This book has been published with all reasonable efforts taken to make the material error-free after the consent of the respective authors. No part of this book shall be used, reproduced in any manner whatsoever without written permission from the editors, except in the case of brief quotations embodied in critical articles and reviews. The Authors of the respective chapters of this book is solely responsible and liable for its content.

All rights reserved.

No part of this publication may be reproduced, transmitted, or stored any digital or electronic form. Also photocopying, recording or otherwise without the prior permission of the editor and publisher is strictly prohibited.

ii

ISBN : 978-93-5506-878-1 First Edition 2023 No. of Pages : 344

46	Status of Women in India Even in this Artificial Intelligence Era	Jisha Mol M.S.	289
47	Empowerment of women through sports and physical education	Dr. Jadhav Dharmsing Gemsing	293
48	Indian Social System And Status Of Women.	Dr. Ramesh Baburao Ghegadmal,	297
49	Women and Health-National Rural Health Mission (NRHM)	Satish Kumar, Dr. Akhilesh Kumar	303
50	Women and Problem of HIV/AIDS	Prof. Dr. Dattatray P. Mundhe	313
51	The Innate Genius of Anna Bhau Sathe	Dr. Dhananjay Trimukhe	321
52	Book Review on Manikarnika- The Queen of Ibanai (Author: Gargi Sarkar)	Ms. Takshashila Jadhav	325
53	Present Status of Women in India	Dr. Usha M. Khandale	323

xiv

ï
ISBN : 978-93-5506-878-1

Present Status of Women in India

Dr. Usha M. Khandale

Head of Department of Home Economics Sardar Patel College, Chandrapur

Introduction

In modern times, however, due to the low social, economic and political status of women and increased oppression of women, the condition of women who are considered goddesses has become very miserable. According to law, women have got rights and position equal to men. The new system gave equal rights to women. Despite much talk of women's participation, equality has not been achieved in reality.

Even though our constitution has advocated the principle of equality between men and women, the position of women in the society has always remained secondary compared to men. Therefore, male dominance is found in all fields in our society. Even in daily life, women get secondary treatment from men. Father's words are more important than mother's. A brother's life is considered more than a sister's. The husband enjoys more freedom than the wife. A common experience like this reveals the secondary position of women. Earlier there was a society where a woman could not do the work or role that a man does. Because women are weaker than men. But women are not weaker than us men in our abilities. This has been proven. Women are now doing jobs that used to be monopoly of men. There are also women in the police and army. Women are working shoulder with men in all fields. Today there is no sector in which women are not working. Be it politics or social causes, women are at the forefront in all fields. Every educated parent wants their daughter to get higher education and get a job. While getting married, every boy thinks that he should get a working wife. So much change is seen in people's thinking today.

Education and employment changed the family roles of women is Men in families where women are employed help their wives in housework do Earlier men did not help their wives in housework, by the husband doing housework was considered inappropriate. But in the mindset of men Parivarman has happened. Therefore, there is a change in the traditional role of women. As women are inferior their status is inferior to men. Come on there has been a change in the traditional thinking of man. Women with their skill the quality has changed. While studying the current status of women in India in ancient times it is necessary to think about

325 | Women in Indian Society

ISBN : 978-93-5506-878-1

to historical times of women is I. it

After independence, special efforts have been made in terms of law to give equal status to women in Indian society. The Constitution of India Provides equal rights to both men and women without any distinction. So women are not inferior to men. They too can play any role as well as men. They just have to get a chance. The idea that women should not be underestimated gradually started to take root

The human values of liberty, equality, fraternity and justice began to influence the generation born after independence. Therefore, giving education to girls like boys, giving them jobs, and not distinguishing between men and women became very important. People started educating their daughters. Educated women started working. Therefore, the traditional role and status of women began to change.

Today a women in not limited to 'Chul and Child' only. Earlier in Indian society, men and women were assigned different roles based on gender. Women should stay at home and do all the household chores. Her roles were to give birth and bring up children, to serve the people of the family, etc. The woman was not making money. Only the women of the lower classes were engaged in agricultural work and mercenary work. Overall the roles of women were traditional. But these roles of women have changed a lot in modern times. Like boys, girls have also started working with higher education. Education, employment and new responsibilities have changed the traditional roles of women. It is found that the proportion of working women is gradually increasing.

To study from a historical perspective, different eras have been categorized on historical basis. The status of women at that particular time.

1. Period of Indus Civilization (2000 to 1750 BC) : The history of India is ancient many thinks explained that the Indus civilization had a matriarchal family system. Tartaktirtha Lakhmanasastri joshi clearly mentioned in his book Vikas Vedic Sanskriti that before the time of Rigveda there was a matriarchal family system in India. The region from the Indus to the Nile was once under a matriarchal family system of society. This conclusion was presented by Marthal. Excavations at Mohanjodoro and Harappa have found numerous female figurines of red and soft stone with images of the seven matrikas carved in rows on a shiki. Therefore, it can be inferred that mother worship was performed in Indus culture women had an important role in the family and society and their status was good. "The idea of woman as the primordial nature of creation did not strike a chord in Dravidian culture."

326 | Women in Indian Society

ISBN: 978-93-5506-878-1

Conclusion:

If present times the status of women has been raised. The status of women has increased in various fields such as social job, political etc. Not only this, women have now got a position equal to men in the family as well from this it is clear that in present times the role and status of Indian women is equal to that of men . Now their status has not deteriorated as before.

References :

- 1) Dr. Damodar Dharmanand kosambi, Culture and Civilization of Ancient India, Rajkmal Prakashan, Delhi, 1993, Page No. 77
- 2) Tarkatirtha Laxmanshastri Joshi, Development of vedic Culture Pragyashala mandal, Y, 1996 (Tri. A) Page No. 233
- Stree Pranshanchi vatchal parivaratnachya dishene Vidut Bhagwat, Pramita Prakashan Bavdhan, Pune.





Statistical Techniques and Business Mathematics for B.com. 1st., Sem.-I

Exclusive Right By Preface Book & Co., Magpur For Manufacture and Marketing this and subsequent editions

1st Edition ISBN 978-81-963511-4-4

© All Rights Reserved : No part of this publication may be reproduced or distributed in any form or by means of stored in a data base of retrival system without the prior written permission of the author.

Published By :

Preface Book & Co.

13-A, Empress Mill Colony, Behind Medical College, Nagpur 440027 Ph. 9325230701,7822087478

Plot No. 200, Nandanwan Layout, NAGPUR - 400 009 (MS). Email : prefacebook22@gmail.com Ph. 9325230701,7822087478

INDEX

HNFT - I	Pg. No.
UNIT 1	1-5
Meaning of Statistics	1.5
Frequency Distribution	0-15
Classification And Tabulation	16-22
Measures of CentralTendency	23-123
UNIT - II	
Measures of Dispersion	133-190
UNIT - III	
Measures of Skewness	191-211
UNIT - IV	
Business Mathematics	212
	212-215
Simple Interest	010 010
Simple Interest Compound Interest	216-220
	Meaning of Statistics Frequency Distribution Classification And Tabulation Measures of CentralTendency UNIT - II Measures of Dispersion UNIT - III Measures of Skewness UNIT - IV Business Mathematics

4

a of

Preface

We take immense pleasure in presenting the Book statistical Techniques and Business Mathematics to the readers at Gondwana University, Gadchiroli.

This book aims to fulfill the comprehensive needs of students preparing for B.Com. Isemester - I. It includes numerous solved illustrations from various university examination. Additionally, each chapter provides a sufficient number of unsolved problemsfor self-practice. Gradtitude extends to my spous, wards, parents, friends and relatives for their support. Special Thanks to our publisher M/s Preface Book & Co., Nagpur for printing the swift printing of this excellent publication. Any suggestions for improvement in this book are most welcome.

30th October 2023

Email:drsrweginwar@gmail.com

Dr. Sapana R. Weginwar Mr. Suyash D. Bhogawar

About the Author



Dr. Sapana R. Weginwar

Holding M.Com, M.Phil., and Ph.D. degrees in Commerce and Management, currently serves as an Associate Professor at Sardar Patel College Chandrapur since 2007. With 17 years of dedicated teaching experience in Commerce and M.A. Economics, she has authored 10 books spanning subjects like cost accounting and statistics. Dr. Weginwar is not only a registered Ph.D. guide but also contributes to Gondwana University Gadchiroli as a Board of Studies member. Her extensive research includes numerous papers presented in national and international conferences, along with publications in reputable journals. Recognition of her scholarly work includes two prestigious awards from a renowned society. Quite an accomplished journey



Mr. Suyash D. Bhogawar

A holder of a Bachelors of Engineering and Masters of Science from the USA. Currently, I serve as a Data Science Architect at RACKSPACE TECHNOLOGIES California USA, specializing in healthcare, neuroscience, and medical informatics. In my role as a research engineer, I've led and contributed to various data science and Al projects, driven by a passion for the foundational principles of statistics. Beyond work, I am a lifelong learner, staying updated on the latest developments in engineering, healthcare, economics, and data science. I believe the convergence of these fields holds the key to driving impactful innovations for an improved quality of life. Thank you for taking the time to learn more about me. I look forward to continuing my journey of positive impact in Statistics and Informatics, contributing to the advancement of knowledge in these dynamic fields

Preface Book & Co.

Empress Mill Colony, Behind Medical College, Ort. No. 13-A, Nagpur - 27 Ph. 9325230701, 7822087478

Plot No 200, Nandanwan Layout, Nagpur - 09, Ph. 9325230701

21. Ale: 7 2410



About the Editors



Dr. Pramod Kumar Yadawa is the Head of Department of Physics & Director of Prof. Rajendra Singh (Rajju Bhaiya) Institute of Physical sciences for Study and Research, Veer Bahadur Singh Purvanchal University, Jaunpur (U.P), India. He obtained his Ph. D. degree in Physics from Allahabad University, Pryagraj, India. He is the life member of the Ultrasonic Society of India (USI), Indian Association of Physics Teachers (IAPT), Acoustical Society of India (ASI), Academy of Microscope Science and Technology (IMST), Materials Research Society of India (MRSI). He is a member of the International Association of Engineers. He has been awarded by Best Teacher

Award 2016 by AMITY SET, New Delhi. He has published 72 research papers in peer-reviewed journals. He is an Editorial Board Member of International Journal of Physics and Material Science, Journal of Materials Physics and Chemistry, Boson-Journal of Modern Physics, BMC Research Notes, American journal of Nano Research and Applications etc. He is a reviewer of Elsevier Science Journals, International Journal of Multi-discipline Modelling of Material Structures, Advanced Materials Letters, Material Science, etc. His current interests of research include the ultrasonic non-destructive testing (NDT) characterization of condensed materials, thermal properties of nanofluids, synthesis of metal oxides nanoparticles, metallopolymers, etc. their applications as engineering and biomedicals.



Dr. Nitesh Jaiswal is a Senior Assistant Professor in the Department of Chemistry, Prof. Rajendra Singh Rajju Bhaiya Institute of Physical Sciences for Study and Research, Veer Bahadur Singh Purvanchal University, Jaunpur, Uttar Pradesh. He is currently serving as Convenor of the Board of Studies, Department of Chemistry, Veer Bahadur Singh Purvanchal University, Jaunpur, He obtained his doctoral and master's degrees from the University of Allahabad after completing his bachelor's degree in Chemistry (Hons.) from Banaras Hindu University, Varanasi,

His expertise in teaching lies in the realm of Inorganic Chemistry, while his research pursuits encompass synthetic inorganic chemistry, coordination chemistry, and computational chemistry. At the forefront of scientific innovation, his research group is dedicated to the design, synthesis, and application of transition metal complexes, showcasing a commitment to addressing pressing environmental challenges through cutting-edge chemistry. He is also working on a startup project of UGC to design a molecular catalyst for CO2 capture and conversion. Dr. Jaiswal has been conferred with the Young Scientist Award by the International Academy of Physical Sciences in 2019. He is a Life Member of the Indian Science Congress Association.



Dr. Sujeet Kumar Chaurasia is presently working as Scientist/Assistant Professor in the Centre for Nanoscience & Technology, Prof. Rajendra Singh (Rajju Bhaiya) Institute of Physical sciences for Study and Research, Veer Bahadur Singh Purvanchal University, Jaunpur, Uttar Pradesh, India. He has more than 9 years of teaching and research experience to undergraduate and postgraduate students of Science. He has published 34 research papers in reputed international journals and 06 book chapter in the area of functional materials for energy

storage, nanotechnology, food packaging, smart materials and also in field of science and technologies. He has completed his Ph.D. from Banaras Hindu University, Varanasi, India and also awarded Dr. D. S. Kothari Postdoc from UGC, New Delhi, India to work at Department of Physics and Astrophysics, University of Delhi, Delhi. He also worked as a Postdoctoral Researcher at University of Warmia and Mazury in Olsztyn, POLAND in the field of NMR Relaxometry. Currently, he is guiding 03 Ph.D. students in the field of Material science and nanotechnology. His current research interests include polymer electrolytes, nanocomposites, supercapacitors, functional materials for energy storage and conversion devices, battery & supercapacitor electrodes, and EMI shielding applications. Moreover, he has been serving as a reviewer for several reputed international journals.



SHREE PUBLISHERS & DISTRIBUTORS 4637/20, Ansari Road, Daryaganj, NEW DELHI-110002 www.shreepublishers.com



Advancement in Functional Materials



HREE

Advancement in **Functional Materials**



Editors

Pramod Kumar Yadawa Nitesh Jaiswal Sujeet Kumar Chaurasia



27 ILLUMINATION OF MECHANICAL, THERMAL AND ELASTIC PROPERTIES OF TRIPLE SUPER PHOSPHATE (TSP) IN AQUEOUS MEDIA: AN ULTRASONIC TACTIC

Paritosh L. Mishra¹* and Urvashi P. Manik²

ABSTRACT -

The ultrasonic tactic/approach helps to quantity the speed of sound in pure as well as in different liquids mixture. By obtaining these experimental data of speed of sound and density, various properties (like, mechanical, thermal and elastic) of the liquid and their mixtures can be calculated. These properties are too useful in understating and gathering the knowledge of interaction between the solute and solvent components of liquid and their mixtures. In view from this scenario, present manuscript reports the investigation of TSP in aqueous media to explore the intermolecular interaction in the liquid system (TSP+H₂O) at fixed 2MHz frequency by varying the concentration and temperature. All mechanical, thermal and elastic parameter shows the positive values suggesting strong intermolecular interaction among ions of solute (TSP) and solvent (H₂O/NaCl/Na₂SO₄) through hydrogen bonding. This kind of data provides the information requires in many aspects and have applications in the field of agriculture sectors through which the quality of fertilizer (TSP) can be improved.

Keywords: Acoustic, elastic property, mechanical property, thermal property, ultrasonic

INTRODUCTION

The excess concentration of salts in agricultural land not only generates the problem of salinity but adverse effects in plants and possess risks to human health.¹ Due to this rapid salinization, the scarcity and limited food resources, have created burden and pressure for the survival of human needs unless certain measures are taken to, overcome this pressure. To achieve food security, the attempts need to focus on both area expansion under agriculture as well as a rise in crop productivity.²

Accordingly, it is needed to develop a simple and low-cost method for soil salinity management to enhance the productivity of crops to meet the ever-increasing requirement of food also to improve the high and regular income and employment generation to the farmers.

In view to resolve these problems, fertilizers are applied in agricultural

^{1.} Department of Physics, Sardar Patel Mahavidyalaya, Chandrapur, Maharashtra, India. 442402, *Corresponding author

Professor & Head Department of Physics, Sardar Patel Mahavidyalaya, Chandrapur, Maharashtra, India. 442402

land worldwide to supply the micro and macronutrients for bulk crop production, to provide better plant nutrition, and to increase the fertility of saline soil.³ Literature survey revealed that the efficiency of fertilizer is higher than salt affected soil as compared to non-saline salts. The capacity in the absorption of P type macronutrient may be decreased in existence of surplus Cl or SO₄. However, P type macronutrients not only fulfill their deficiency in soil but also reduce their adverse effects like Cl or SO₄.⁴

As ultrasonic is a versatile non-destructive technique and highly useful for investigation of various physicochemical and thermo-acoustical properties of pure liquid as well as liquid compositions.^{5,6} Recent developments have found the use of the ultrasonic technique in medicine, engineering, and also in the agriculture field. Thus, this kind of study is important for both human beings as well as plants. Because of the above circumstance, an effort was carried out to fulfill the stated circumstances and to bring out such studies on fertilizer: Triple Super Phosphate of different concentrations viz. 0.002– 0.02 mol.kg⁻¹ change by weight fraction in water and aqueous 0.2 mol.kg⁻¹ solutions of salts namely: Sodium Chloride and Sodium Sulfate at various temperature 288.15–303.15K with the help of Non-Destructive Technique (Ultrasonic Technique). The data obtained also sheds light on intermolecular interactions between fertilizer-water and fertilizer-water-saline salts in view to find a way to control or minimalize the stated problem of soil salinity. The ultrasonic sound velocity (U) and density (Q) measurements and their aligned properties (elastic, mechanical and thermal) find the wide applications in characterizing the physico-chemical behavior of liquid mixture.

EXPERIMENTAL DETAILS

Table 27.1: Shows the different chemicals used in view to perform the experiment.									
Chemical Name	Molecular Weight	CAS Number	Purchased From	Purity					
Sodium Chloride	58.44	7647–14-5	Himedia Lab.	>>					
Sodium Sulfate	142.04	7757–86-6	Pvt. Ltd.,	99.8%					
Double Distilled Water	18.02	NIL	Mumbai						
Triple Super Phosphate	370.11	65996–95-4							

1. Chemicals Information

2. Apparatus Information

Table 27.2: Shows the different apparatuses used in view to perform the experiment.						
Name of Apparatus	Purchased From	Uses				
2 MHz operating Digital Ultra- sonic Velocity Interferometer	Vi Microsystems Pvt Ltd, Chennai	For the measurement of Ultrasonic sound velocity in liquid mixture.				
10ml Specific Gravity Bottle	Enggific Engineering and Scientific, Hyderabad	For the determination of densities of the solutions.				
Automatic Thermo- static Water Bath	Lab Hosp Pvt Ltd, Nagpur	For maintaining the experimental temperature of circulating water constant.				

Defining Parameters

For calculated values of several elastic, mechanical and thermal properties the following defining relations reported in the literature are used:

- (I) Surface Tension (σ): Surface tension describes the tendency of liquid surface at rest to shrink into the least surface area possible = (6.3*10⁻⁴) $\rho U^{3/2}$
- (II) Internal Pressure (π i): Internal Pressure is a significant parameter which is used to understand structure and nature of intermolecular interaction in the liquid molecules = {T α/k_{τ} }
- (III) Isothermal Compressibility (k_T): Isothermal compressibility is used to determine the compressible properties of water supply. Pandey et al. Method (k_T) = $17.1*10^{-4}/(T^{4/9}U^2\rho^{1/3})$
- (IV) Bulk Modulus (K): Bulk modulus is the reciprocal of adiabatic compressibility; it is used to measure the ability of substance = $\frac{1}{8}$
- (V) Thermal Conductivity (k): Thermal conductivity is referring to the ability of material or substance to conduct or transfer heat = $\{3.0^*(QN_A/M)^{2/3}k_BU\}$
- (VI) Specific Heat Ratio (γ): The simplified relation for Specific Heat Capacity as = $\left\{\frac{17.1}{T^{4/9} * o^{1/3}}\right\}$

RESULT AND DISCUSSION

Surface tension was used to interpret the strength of intermolecular interaction that exists in the solution. Between the liquid molecules, cohesive forces exist which are responsible for the variations in the surface tension. The increasing trend of surface tension (σ) with the concentration of solute and temperature indicates that a significant association in the solution.⁷

270 Advancement in Functional Materials

Sound properties have the tendency to irradiate the strength as well as kind of the interaction taking place in the corresponding experimental solutions. In the present work, the internal pressure (π_i) increases with an increase in the concentration of fertilizer, and the temperature for all three solvents. Which confirms there is an attractive force between the constituent components of liquid. This growing ilk of the internal pressure in all the solutions designates a good agreement of interaction between fertilizer-soil salt solutions as compared to fertilizer-water.⁸

Isothermal compressibility is a measure of relative fluctuations in volume. An examination of Table 27.3 reveals the trends in the isothermal compressibility (k_T) is declining with an intensification in the concentration of triple superphosphate fertilizer in water also in the aqueous NaCl and Na₂SO₄ solution of 0.2 mol-kg⁻¹. The diminution in isothermal compressibility values with an increase in the concentration of fertilizer (TSP) seems to be the result of a corresponding decrease in free volume and available volume.⁹ The calculated values of isothermal compressibility's have been further used for the determination of specific heat ratio.

Conc.	σ						k _T		
(mol- kg ⁻¹)	(kg/m ³)			(Nm ⁻	²)		(m ² N ⁻¹)		
	H ₂ O	NaCl	Na ₂ SO ₄	H ₂ O	NaCl	Na ₂ SO ₄	H ₂ O	NaCl	Na ₂ SO ₄
			28	38.15K					
0.000	35331.82	35331.82	35331.82	4.90	4.90	4.90	6.42	6.42	6.42
0.002	35473.68	36407.59	37727.94	4.93	5.09	5.33	6.40	6.26	6.04
0.004	35563.86	36473.87	37807.41	4.94	5.11	5.34	6.39	6.24	6.03
0.006	35632.82	36541.78	37885.95	4.96	5.12	5.36	6.38	6.23	6.02
0.008	35753.23	36609.33	37964.30	4.98	5.13	5.37	6.35	6.22	6.01
0.010	35828.80	36697.29	38041.68	4.99	5.15	5.39	6.34	6.21	6.00
0.012	35922.62	36786.71	38119.23	5.01	5.16	5.40	6.33	6.19	5.99
0.014	35990.58	36854.03	38196.21	5.02	5.17	5.41	6.32	6.18	5.98
0.016	36068.63	36883.86	38272.98	5.03	5.18	5.43	6.30	6.18	5.97
0.018	36125.37	36948.45	38349.93	5.04	5.19	5.44	6.30	6.17	5.96
0.020	36222.07	37016.53	38404.45	5.06	5.20	5.45	6.28	6.16	5.95
									Contd

Table 27.3: The values of Surface Tension (σ), Internal Pressure (π_i), Isothermal Compressibility (k_T), Bulk Modulus (K) and Thermal Conductivity (k) of System TSP+ H₂O/NaCl/Na₂SO₄

Table 27.3: The values of Surface Tension (σ), Internal Pressure (π_i), Isothermal Compressibility (k_r), Bulk Modulus (K) and Thermal Conductivity (k) of System TSP+ H ₂ O/NaCl/Na ₂ SO ₂ (Contd.)									
293.15K									
0.000	35859.88	35859.88	35859.88	5.06	5.06	5.06	6.24	6.24	6.24
0.002	36030.14	36870.76	38094.98	5.10	5.25	5.47	6.22	6.09	5.91
0.004	36104.36	36927.56	38175.75	5.11	5.26	5.49	6.21	6.08	5.9
0.006	36171.88	37001.10	38255.58	5.12	5.27	5.50	6.20	6.07	5.89
0.008	36251.80	37066.05	38312.80	5.14	5.28	5.51	6.19	6.06	5.88
0.010	36330.82	37131.47	38391.67	5.15	5.29	5.53	6.17	6.05	5.87
0.012	36385.24	37196.15	38448.19	5.16	5.30	5.54	6.17	6.04	5.87
0.014	36460.69	37268.06	38525.94	5.17	5.32	5.55	6.16	6.02	5.86
0.016	36536.82	37326.67	38604.23	5.19	5.33	5.56	6.14	6.01	5.85
0.018	36586.62	37401.91	38704.70	5.20	5.34	5.58	6.14	6.00	5.83
0.020	36668.47	37480.07	38783.03	5.21	5.36	5.60	6.13	5.99	5.82
			29	8.15K					
0.000	36420.62	36420.62	36420.62	5.23	5.23	5.23	6.06	6.06	6.06
0.002	36529.49	37336.12	38604.90	5.25	5.40	5.64	6.05	5.93	5.76
0.004	36603.83	37404.09	38664.79	5.27	5.41	5.65	6.04	5.92	5.75
0.006	36653.00	37472.35	38723.57	5.28	5.43	5.66	6.03	5.91	5.75
0.008	36714.39	37542.33	38789.36	5.29	5.44	5.67	6.02	5.90	5.74
0.010	36791.00	37588.43	38868.14	5.30	5.45	5.69	6.01	5.89	5.73
0.012	36844.98	37638.43	38949.65	5.31	5.46	5.70	6.01	5.89	5.72
0.014	36900.55	37684.37	39029.89	5.32	5.47	5.72	6.00	5.88	5.71
0.016	36955.42	37752.57	39104.77	5.33	5.48	5.73	5.99	5.87	5.70
0.018	37010.53	37786.21	39167.24	5.34	5.48	5.74	5.99	5.87	5.69
0.020	37067.29	37848.01	39227.91	5.36	5.50	5.76	5.98	5.86	5.68
303.15K									
0.000	36704.40	36704.40	36704.40	5.36	5.36	5.36	5.95	5.95	5.95
0.002	36858.75	37592.17	38878.01	5.39	5.57	5.77	5.92	5.83	5.65
0.004	36910.39	37639.74	38960.83	5.40	5.57	5.78	5.92	5.82	5.64

Contd.

Table 27.3: The values of Surface Tension (σ), Internal Pressure (π_i), Isothermal Compressibility (k_T), Bulk Modulus (K) and Thermal Conductivity (k) of System TSP+ H ₂ O/NaCl/Na ₂ SO ₄ (Contd.)									
0.006	36958.70	37685.80	39042.63	5.41	5.58	5.80	5.91	5.81	5.63
0.008	37021.06	37734.99	39122.21	5.42	5.59	5.81	5.91	5.81	5.62
0.010	37072.90	37781.56	39203.60	5.43	5.60	5.83	5.90	5.80	5.61
0.012	37126.66	37828.59	39285.12	5.44	5.61	5.84	5.89	5.80	5.60
0.014	37181.35	37875.68	39365.15	5.45	5.61	5.86	5.89	5.79	5.59
0.016	37234.56	37917.35	39446.49	5.46	5.62	5.87	5.88	5.79	5.58
0.018	37286.25	37968.16	39527.01	5.47	5.63	5.89	5.88	5.78	5.57
0.020	37341.98	38014.64	39604.92	5.48	5.64	5.90	5.87	5.77	5.56

272 Advancement in Functional Materials

Elastic ilk of liquid can be best understood by Bulk Modulus (K) due to the hydrogen bonding form in the distinct components of the solutions intensifications with the bulk modulus. In the existing case, it is found that the bulk modulus increases with varying the concentration of fertilizer. As, water is a polar solvent and when salts and fertilizer are mixed, the well inter-Molecular interaction arose, follow-on in close packing of molecules. The calculated values of bulk modulus are listed in Table 27.4 which specifies the strong association amongst fertilizer and saline salts molecules as compared to distilled water and fertilizer. The bulk modulus of the solvent is lower than that of the solution and increases with an increase in the concentration of the fertilizer.¹⁰

In the present investigation, the increase in bulk modulus values with the temperature rise is observed. This supports the facts indicated by the variation in ultrasonic velocity with temperature. Ultrasonic velocity determination or measurement can be utilized to evaluate thermal conductivity theoretically. The theoretical value of thermal conductivity of solvent (water) shows good agreement with the literature data.^{11,12}

From Table 27.4 it is observed that the evaluated value of thermal conductivity for triple super phosphate mixed in 0.2 molal concentration of salt solutions is more than that of the 0.2 molal salt solutions at all concentrations and temperature. The observed trend of thermal conductivity is: Water > Fertilizer + Saline Salts > Saline Salts

In the current investigation, both the ultrasonic velocity and density values increase with the increase in temperature and as per Bridgeman's relation, thermal conductivity directly depends on these two factors. The increase in thermal conductivity with an increase in concentration and temperature clear that the flow of energy is possible when molecules get close to each other. This means in the present system intermolecular interaction taking place. It is confirmed with the rise of velocity, density, and drop of free length values due to the close packing structure.

Table 27.4: The values of Bulk Modulus (K), Thermal Conductivity (k) and SpecificHeat Ratio (γ) of System TSP+ H ₂ O/NaCl/Na ₂ SO ₄ .										
Conc.	К			k	k			γ		
$(mol-k\sigma^{-1})$	(kg/m^3)			(Wm ⁻¹	(Wm ⁻¹ K ⁻¹)			$(K^{4/9})^{-1}(kg^{1/3}m^{-1})^{-1}$		
	H ₂ O	NaCl	Na ₂ SO ₄	H ₂ O	NaCl	Na ₂ SO ₄	H ₂ O	NaCl	Na ₂ SO ₄	
				288.15	K					
0.000	2147315500	2147321948	2147321948	0.6292	0.6292	0.6292	0.138027	0.138027	0.138027	
0.002	2173480500	2205266719	2307871343	0.0841	0.0856	0.0876	0.137919	0.137396	0.136798	
0.004	2192106686	2224726721	2322802241	0.0843	0.0857	0.0878	0.137880	0.137364	0.136755	
0.006	2210703584	2229615318	2328579846	0.0844	0.0858	0.0879	0.137842	0.137331	0.136712	
0.008	2226847866	2234607409	2334305503	0.0846	0.0859	0.0880	0.137790	0.137298	0.136670	
0.010	2243489963	2239581280	2340023962	0.0847	0.0860	0.0881	0.137744	0.137266	0.136629	
0.012	2261606018	2246232135	2345686504	0.0848	0.0862	0.0882	0.137702	0.137232	0.136588	
0.014	2275534528	2252980305	2351364374	0.0849	0.0863	0.0884	0.137666	0.137200	0.136549	
0.016	2288598783	2257955952	2357011928	0.0851	0.0863	0.0885	0.137619	0.137169	0.136510	
0.018	2301729318	2259879398	2362651870	0.0851	0.0864	0.0886	0.137572	0.137135	0.136470	
0.020	2314127374	2264596678	2368307046	0.0853	0.0865	0.0887	0.137527	0.137103	0.136432	
				293.15	K					
0.000	2190879703	2190879703	2190879703	0.6354	0.6354	0.6354	0.137016	0.137016	0.137016	
0.002	2217324518	2249335165	2350047306	0.0850	0.0863	0.0882	0.136903	0.136472	0.135820	
0.004	2232868400	2264583030	2354083182	0.0851	0.0864	0.0883	0.136861	0.136454	0.135776	
0.006	2249002609	2268933827	2359980717	0.0852	0.0865	0.0885	0.136827	0.136416	0.135734	
0.008	2263521423	2274316915	2365825616	0.0853	0.0866	0.0885	0.136778	0.136388	0.135692	
0.010	2279526666	2279175561	2369813608	0.0855	0.0867	0.0887	0.136731	0.136359	0.135652	
0.012	2295525210	2284068257	2375611860	0.0856	0.0868	0.0888	0.136688	0.136332	0.135611	
0.014	2310305998	2288919012	2379567536	0.0857	0.0869	0.0889	0.136646	0.136297	0.135572	
0.016	2324108184	2294218528	2385298006	0.0858	0.0870	0.0890	0.136602	0.136277	0.135533	
0.018	2341527525	2298703842	2391067121	0.0859	0.0871	0.0892	0.136566	0.136238	0.135494	
0.020	2360186293	2304217788	2398678696	0.0860	0.0873	0.0893	0.136516	0.136195	0.135455	
				298.15	K					
0.000	2237573686	2237573686	2237573686	0.6420	0.6420	0.6420	0.136045	0.136045	0.136045	
0.002	2261473075	2292117579	2395933525	0.0858	0.0870	0.0890	0.135961	0.135543	0.134846	
0.004	2277231793	2304353571	2396963955	0.0859	0.0871	0.0891	0.135921	0.135513	0.134803	
0.006	2294021301	2309436561	2401148275	0.0860	0.0872	0.0892	0.135879	0.135483	0.134761	
0.008	2307371415	2314540456	2405266400	0.0861	0.0874	0.0893	0.135836	0.135451	0.134719	

Illumination Of Mechanical, Thermal 273

Contd.

Table 27.4: The values of Bulk Modulus (K), Thermal Conductivity (k) and Specific Heat Ratio (γ) of System TSP+ H ₂ O/NaCl/Na ₂ SO ₄									
0.010	2323501867	2319754422	2409972427	0.0862	0.0874	0.0894	0.135793	0.135422	0.134679
0.012	2340188477	2323049570	2415779639	0.0863	0.0875	0.0895	0.135753	0.135388	0.134639
0.014	2353136675	2326587689	2421817949	0.0864	0.0876	0.0896	0.135711	0.135359	0.134601
0.016	2366463651	2329876003	2427768544	0.0864	0.0877	0.0898	0.135669	0.135329	0.134561
0.018	2382465556	2334991639	2433271973	0.0865	0.0877	0.0899	0.135628	0.135315	0.134522
0.020	2400773042	2337522649	2437754681	0.0866	0.0878	0.0900	0.135585	0.135267	0.134484
	1			303.15k	C				
0.000	2261908674	2261454293	2261908674	0.6454	0.6454	0.6454	0.135106	0.135106	0.135106
0.002	2282222345	2315644749	2421700500	0.0863	0.0874	0.0894	0.135021	0.134603	0.133937
0.004	2295241030	2326451087	2421114637	0.0864	0.0875	0.0895	0.134984	0.134572	0.133894
0.006	2310241988	2329845570	2427217099	0.0864	0.0876	0.0897	0.134951	0.134544	0.133853
0.008	2322527068	2333147327	2433261141	0.0865	0.0877	0.0898	0.134903	0.134511	0.133812
0.010	2339177160	2336645090	2439137922	0.0866	0.0877	0.0899	0.134864	0.134482	0.133772
0.012	2354356910	2339982356	2445174480	0.0867	0.0878	0.0900	0.134825	0.134448	0.133733
0.014	2368417486	2343283047	2451231992	0.0868	0.0879	0.0902	0.134785	0.134419	0.133693
0.016	2384323584	2346666351	2457166012	0.0869	0.0879	0.0903	0.134746	0.134390	0.133655
0.018	2397016404	2349595007	2463227199	0.0870	0.0880	0.0904	0.134708	0.134359	0.133616
0.020	2415693884	2353253872	2469221816	0.0870	0.0881	0.0905	0.134669	0.134329	0.133578

274 Advancement in Functional Materials

The variation of specific heat ratio of changeable weight fraction (0.002–0.02 mol-kg⁻¹) of fertilizer: Triple Super Phosphate in water and 0.2 mol-kg⁻¹ aqueous solutions of NaCl and Na₂SO₄ at overall temperatures. The heat capacity ratio (γ) is decreasing, with the rise of temperature and fertilizer addition in the pure water and salt solutions. These results of specific heat ratios well support the increase of density with an increase in the concentration of solute fertilizer.¹³ The order of variation in water and salt solution is observed as Na₂SO₄<NaCl<H₂O

CONCLUSION

Numerous mechanical, thermal and elastic parameters were calculated for triple superphosphate fertilizer in both water and saline salt solutions (NaCl and Na₂SO₄) at 288.15K, 293.15K, 298.15K, and 303.15K temperature by measuring the values of density and speed of sound. All parameters help to investigate the inter-molecular connections present between the molecules of triple superphosphate fertilizer-water and triple superphosphate fertilizer-saline salts. The influence of concentration on the different properties was observed and explained with the help of physicochemical behavior.

On the basis of resultant discussion, it was concluded that:

- a. The variation in concentration, ilk of solute, ilk of the solvent, and its place actings a chief role in defining the kind of interactions going on in the solution.
- b. It is further concluded that at higher concentrations the H-bonding is strong.
- c. All the properties exhibit the maximum values for TSP fertilizer dissolved in Na₂SO₄ solution, coz it has a weak interaction with water molecules can bind with fertilizer molecules more excellently.
- d. The trend of interaction occurred as: (TSP+H₂O+Na₂SO₄) > (TSP+H₂O+NaCl) > (TSP+H₂O)

REFERENCES

- 1. A, G. Murugesan, S. Maheshwari and G. Bagirath, J. Environ. Resour. vol. 2(3), pp. 307–312, 2008.
- K. Pradeep and P. K. Sharma, Front. Sustain. Food Syst. pp. 1–15, 2020. DOI: 10.3389/ fsufs.2020.533781.
- L.C. Nunes, G.G.A. de Carvalho, D. Santos and F.J. Krug. Acta. Part B At. Spectrosc. vol. 97, pp. 42–48, 2014.
- 4. B. Naseem, and I. Arif. J. Mol. Liq. vol. 225, pp. 64-86, 2017.
- D. V. Jahagirdar, B. R. Arbad, S. R. Mirgane, M. K. Lande and A. G. Shankarwar, J. Mol. Liq., vol. 75(1), pp. 33–43, 1998.
- 6. A. Nain, R. Pal and R. Sharma, J. Mol. Liq., vol. 165, pp. 154–160, 2012.
- Sonune P.R., Manik U.P. and Mishra, P.L Materials Today Proceeding, 2023. DOI: 10.1016/j.matpr.2023.06.417.
- 8. C. V. Suryanarayana and J. Kuppusami, J. Acoust. Soc. Ind. vol. 4, p. 75, 1976.
- Mishra, P.L., Lad A.B., and Manik U.P., *Materials Today Proceeding*, vol. 60, pp. 681–685,2022. DOI: 10.1016/j.matpr.2022.02.316.
- Sandhu J.K.S., Manik U.P. and Mishra, P.L., *Acta Scientific Applied Physics*, vol. 3(5), pp. 02–09, 2023.
- V. D. Yusufova, R. I. Pepinov, V. A. Nekolev and G. M. Guseinov, *Inzh. Fiz. Zh.*, vol. 29(4), pp. 600–605, 1975.
- 12. N. B. Vargaftik, Os'mini, P. Yu, Teploenergetika, vol. 3(7), pp. 11–16, 1956.
- Khatun, R., Sultana R. and Nath R.K., Orient Journal of Chemistry, 34(4), pp. 1755–1764, 2018.
- 10. Maurya, V.N., Arora D.K., Mauruya A.K. and Gautam R. A. (2013): World of Science Journal, vol 02.
- 11. Mishra, P.L. and Manik U.P., International Journal of Scientific Research, pp. 215–217, 2015.
- 12. Suryanarayana, C.V. and Kuppusami J., *Journal of Acoustical Society of India*, vol 4, p. 75, 1976.
- Mishra, P.L., Lad A.B., and Manik U.P., *Macromolecular Symposia*, vol. 407, pp. 2100432(1–8), 2023. DOI: 10.1002/masy.202100432.



Rajiv M. Meshram | Dr. Amit A. Jagiya | Dr. Khushal M. Borkar





CHAPTER 21

PHOSPHATE COMPOUNDS DOPED WITH RARE EART IONS EXHIBITING LUMINESCENCE

¹Sayli Guru,²U. P. Manik, ³Amol Nande

^{1. 2}S P College, Chandrapur, India ³G N College of Science, Ballarpur, India

Abstract

This book chapter centers on the examination of luminescent materials compromising phot compounds as the host material and rare earth ions. The initial section commences with an overvithe luminescence phenomenon, providing a background study to establish context and understan Subsequently, we delve into a review of the commonly employed methodologies for synthesis context. As a result, this article primarily focuses on providing an overview of various phosphate – luminescent materials and explores contemporary developments and advancements in the figluminescence.

Keywords: luminescence, phosphate, rare earth

Introduction

Luminescence is the ability of a body to produce light when subjected to electromagnetic radiation other forms such as energy released from chemical reaction or electrons. Examples: When the mat is stimulated by low – energy photons frequently ultraviolet radiations (photoluminescence), by carays (cathodoluminescence), by the strength of an electric field (electroluminescence), by X-rays (luminescence), and similar scenarios. The emitted light from a luminescent material manifests is visible segment of the electromagnetic spectrum but may also extend into the infrared (IR) or ultrav (UV) regions. Luminescent materials are commonly referred to as "Phosphors" which comprise a lattice and a luminescent center often referred to as an "activator". Activators absorb incoming radia becoming elevated to an excited state. Subsequently, the excited state transitions back to the ground emitting radiation [1]. Thus, the host lattice is composed of at least one type of oxide, derived products such as sulphide, aluminate, alumino silicate, silicate, phosphate and others. Above al mentioned materials, Phosphate holds a crucial position among these materials, playing a significant in environmental systems. It is valued for its versatility, contributing to a variety of functions due notable attributes, including remarkable simplicity, low viscosity, high ultraviolet (UV) transmis excellent mechanical and thermal stability, isotropic refractive index, and simple synthesis[2,3].

Over the recent years, phosphate compound have garnered significant attention as materials for Pla Display Panels (PDPs). This is attributed to the fact that phosphors based on phosphates serves as criluminescent hosts, exhibiting robust absorption in the Vacuum Ultraviolet (VUV) region (100-200. Additionally these, compounds showcase high chemical stability and are cost effective, fur contributing to their appeal for use in such applications [4]. Various rare – earth activated phosp materials have been recently explored for their application in the field of light – emitting diodes (LE Among these, mineral based phosphate materials have particularly garnered significant attention as p - luminescent materials. Phosphate based phosphors are widely recognized for their suitability in sol state lighting applications. Phosphate, due to its medium energy level, high damage threshold, ro thermal and chemical stability, cost effective raw materials, and relatively simple preparation conditi stands out as an ideal host for white LEDs (W-LEDs) [5].

Ent

Methodorous Various synthesis methods have been reported to date for the production of phosphate based materials. Methodology Each method are described below:

1. Melt Quenching Technique 1. Melt Quenching method is one of the oldest techniques used for producing phosphors, including The melt quenching mothod for glass or white LEDs (w- LEDs) preparation. In this the melt quenching incomes for glass or white LEDs (w- LEDs) preparation. In this method, starting phosphate based phosphate are carefully mixed and melted in a furnace. The resulting multiplication of the starting phosphate and minerals are carefully mixed and melted in a furnace. phosphate based phosphate carefully mixed and melted in a furnace. The resulting melt solution is then materials and minerals are carefully pressed with another plate to create a thin. poured onto a brass plate and promptly pressed with another plate to create a thin, transparent glass. The poured onto a brass plate is influenced by the starting materials and the dopant used in the preparation color of the obtained glass is crucial factor affecting glass formation; a first used in the preparation color of the obtained guenching is crucial factor affecting glass formation; a faster rate of melt quenching process. The rate of melt quenching formation [6, 7]. process. In the a more efficient glass formation [6, 7].

2 Wet Chemical method 2. Wet Chemical method is a chemical based - approach that utilizes materials typically in the form of the wet chemical dissolve in water. This method encomposition and the state of the s The wet chemical dissolve in water. This method encompasses various techniques such as sol - gel nethod and co precipitation method, all falling under the broader category of "Wet Chemical Synthesis". method and to perform the considered a primary wet chemical method, involves the transformation of a the sol- gel method, involves the transformation of a The sol- 5ct agel, allowing for the controlled synthesis of materials. This approach offers versatility in solution into a gel, allowing for the controlled synthesis of materials. This approach offers versatility in solution material properties. Overall, the wet chemical method represents a diverse set of techniques, each contributing to the synthesis of material with specific characteristics [8].

The role of Phosphates in various luminescent domains

Extensive research has been conducted in research years on luminescent materials based on phosphates, driven by their exceptional spectroscopic characteristics and crystalline structure. In recent studies, Lanthanides (Ln) - activated rare - earth phosphate has emerged as a noteworthy and practical optical fuorescent probe. This is attributed to its unique luminescent behavior, particularly in suitable activators like Eu3+, Tb3+, Gd3+, etc. Consequently, it serves as an effective alternative to conventional semiconductor quantum dots containing cadmium. Moreover, Ln- activated large band gap nanocrystals also provide a robust crystal environment for the dopant ions, leading to an elevated photoluminescence (PL) quantum yield (QY). Among numerous phosphate based phosphors, LaPo4 (LAP) has gained paramount importance in various fields, serving as laser materials, catalysts, and versatile biological labels, heat resistant materials, catalysts and photon up conversion materials [9].

Conclusion

The proposed review article comprehensively examines the recent advancements of phosphate - based materials in various luminescent applications. The detailed discussion encompasses different synthesis methods for preparing phosphate based phosphor materials, including the wet chemical method, combustion method and sol gel method. The research emphasizes that enhancing the photoluminescence (PL) of phosphate phosphors requires materials with properties suitable for solid state lighting applications. This phosphors hold potential for applications in workforce, clinical dosimetry and environmental dosimetry, with a particular focus on achieving extremely low dose dosimetry. The investigation underscores the necessity for materials highly sensitive to ionizing radiations in this context.

References:

- Ł Lingeshwar Reddy K, Balaji R, Kumar A and Krishnan V 2018 Lanthanide Doped Near Infrared Active Upconversion
- 2 Nanophosphors: Fundamental Concepts, Synthesis Strategies, and Technological Applications Small 14 1-27
- 3 Saad M, Stambouli W, Mohamed S A and Elhouichet H 2017 Ag nanoparticles induced luminescence enhancement of
- 4 Eu3+doped phosphate glasses J. Alloys Compd. 705 550-8

- Phalak G, Patil D, Patil A and Mhaske S 2019 Synthesis of acrylated cardanol diphenyl phosphate for UV carable 5. coating application Eur. Polym. J. 121 109320
- coating application Eur. Polym. J. 121 105521 Mukhopadhyay L and Rai V K 2020 Thermally stable red emitting xenotime phosp hate nanophosphors for displays Mate 6.
- 7. Res. Bull. 121 110628
- Res. Bull. 121 110628 Salman S M, Salama S N, Darwish H and Abo-Mosallam H A 2009 In vitro bioactivity of glass -ceramics of the 8. 9.
- CaSiO3-Ca5(PO4)3F-Na2SiO3 system with TiO2 or ZnO additives Ceram. Int. 35 1083-93
- CaSiO3-Ca5(PO4)3F-Na2SiO5 of the second state of Dy 3+ doped
 Kaewkhao J, Wantana N, Kaewjaeng S, Kothan S and Kim H J 2016 Luminescence characteristics of Dy 3+ doped
- 12. CaO-SiO2-B2O3 scintillating glasses J. Rare Earths 34 583-9
- Ullah I, Khan I, Shah S K, Khattak S A, Shoaib M, Kaewkhao J, Ahmad T, Ahmed E, Rooh G and Khan A 2021 13.
- 14. Luminescence properties of Sm3+ doped Na2B4O7 glasses for lighting application J. Lumin. 230 117700
- Oguz K F, Ekdal E, Aslani M A A, Canimoglu A, Garcia Guinea J, Can N and Karali T 2016 Study of luminescence of 15.
- Runowski M, Shyichuk A, Tymiński A, Grzyb T, Lavin V and Lis S 2018 Multifunctional Optical Sensors for 16.
- Nanomanometry and Nanothermometry: High-Pressure and High-Temperature Upconversion Luminescence of 17.
- Lanthanide-Doped Phosphates LaPO4/YPO4:Yb3+-Tm3+ ACS Appl. Mater. Interfaces 10 17269-79

International Books

1. Ulhaskumar Patki, Ajay Kurhe, and S. B. Kishor, Text Document Clustering published by International publisher LAMBERT, Germany, in Feb. 2022 (ISBN-13: 978-620-4-74538-1, ISBN-10:6204745387) Also published in 6 different languages (Russian, German, Spanish, French, Italian, Portuguese)

<u>M. Urkunde, Dr. S. B. Kishor, Dr. Narjane, Face Recognition Technique: A Literature Survey on Face Recognition and Insight on Machine Recognition Using</u>

National Books

BCA/BBA/BBM/BMS Common Entrance Examination (CET)

an o Dr.S.

22.5

Ren Ling

P









Das Ganu Prakashan Pratibha Book Distributors Above Maratha Mandir, Rani Zanel Square, Skabuka, Nagpur-440 012 Mo. 9822201952

A COMPLETE **COMPREHENSIVE GUIDE FOR**

> Dr. S. K. Lokhande Tr. S. Madhavi Edited by:

COMMON ENTRANCE TEST

Co-Authors: Tr. Nigar M. Pathan Tr. Goldy S. Thadwani Tr. Apeksha N. Pimpale Dr. Bipin G. Pardeshi Tr. Swapnil M. Bhagat

E F

Dr. S. B. Kishor

Tr. Shailendra S. Gupta



DAS GANU PRAKASHAN NAGPUR (INDIA)

Python



Dr. S. B. Kinker, Haad of Computer Science Dipt. Working Lince Sep 4, 2000, In Presh Makeretryslags, Characterian III is a Chairman of Computer Science Band, leaves University, Cashberd, and a manifest of Academic Council Member, Dr. Kikler is leaves University Computer Science/Application board to Fire MAAC A and above Universities and Autocounces unlines. He is shading his Mean and superfise on many start councilies that include KAAC, MAAC, Research, etc. He is the Head of PL/RSS tart itute of Higher Learning, Research in Specialized Studies) in Computer Science in the

During Credit-2019, Dr. Kinker has uploaded more than 225 videos on Teaching Aids and other Utility videos for Students, Faculty, Academician, and Researcher, on his You Tube channel. Nove than 1.74 labb Vignets viewed his videos across 38 countries. So far, 14 research achieve here been anarded a Ph.D. under his guidence and 96 scholars have been awarded NLPhil. He has been cracilied with 05 Patents including Australian Patents. He has published 65 papers in a reputed Journal including Scopus and published 57 papers in various Rational and International conferences.

Dr. S. B. Kishor is a double Ph.D. holder in the Professional subjects that includes Computer Science and Management, has written 75 books (including 17 International Publication) and with re-edition, this figure goes to 128 books. Many of his books are included as a Text Book and Reference book in various courses across the Universities and Autonomous colleges

Dr. Kishor has conducted more than 75 sessions as a resource person in various UGC-HRDC Academic Staff College, Ph.D. Course Work at University center, and Teaching Learning Center, etc. He has trained more than Nine thousand Ph.D. Research Scholars and faculty members on the lopic of ICT, Plagtarism, etc.

As a social responsibility, Or, Kishos has so far donated blood 17 times and has registered a posthumous tyre donation at a government medical hospital, Charderapur, He has donated 531 unique books on Compute Science and Motivational books to Lale Ramadevi Ookarneth Sharma, Library He has published 54 articles in the Hitavada orwspaper (A Central India largest Circulation daily English Newspaper), mostly on Research topics for making awareness of Research among the reader.

Research topics for making awareness of Research among Elst nader. Nost Important, to honor his dedication and determination for his work, Dr. S. B. Kisher has been awarded many awards that include some of the prestigious like 1deal Teacher Award-2012 and Test Writze Award-2013' by RTMHU, Nagpur. A holder of "India Book of Records-2014' for writing Maximum Books is Computer Science and Recipients of "Ideal Teacher Award-2016" of Gondesna University, Gadchiroli and "Distinguish HOD-2017", of Computer Society of India, Mambal and "ShikshaRatan-2019" of Zero Mile, Nagpar, "Best Faculty Development for Online Teaching During COVID-16" of IESA, "Excellence Teacher Award - 2021" from SSN Mandal's Chandraper, MASTER Award from IIT Secter Tudorial, Mumbal in 2022.

Ex. Vice-Chancellor, Prof. Dr. Srinivasa Varkhedi of Gondwana University, Gadchiroli sent him a letter of appreciation for his valuable contribution and guidance for successful conduction of the 1° online examination of university and similarly, IIT Mumbal sent an appreciation letter for his contribution to IIT Spoken Tutortal awareness and conducting a program.



ISBN 978-81-864733-5-8



Pratibha Book Distributors Above Maretha Mandir, Rani Zansi Square, Stabuldi, Nagpur-440 012 Ph.:0712-2547129



Computer Security and Law



Dr. Masthery Macham Bokary is a Post Oroduste and Ph.D. is Computer Science & Engineering, also complete MBA (Markeling) & Currently Working as Associate Piofessor & HOD, at Department of Computer Science, Shi Sharda Braves Education Society's, Institute of Technology and Management, Nanded. Its has more than 20 years of teaching orderionce. He has brieght Gladente & Post Gradente

Ha has more than 20 years or recording or the form of a paper of a paper of S.R.T.M. students. He is working as BCIS & DCE member Computer Science & Application of S.R.T.M. University, Nanded, He organizes different conferences & Seminars. He has published

meanth papers at National & International level and also working as a Supervisor for Ph.D. in Computer Science.



Dr. S. S. History, Head of Computer Science Dept. working tince Sep 4, 2009, in Santar Pater Mathemidyalaya, Chandraper, He was a Chairman of, Computer Science Board, Condwana University, Celebrate, and a member of Academic Council Member. Dr. Nisternia working as a member of the Computer Science (Application housdto Five MAAC X and atoms Grade Universities and Autonomous colloges. He is sharing his ideas and entry is a program important committees that include IDAC, NAAC, Research, etc. He is the Head of PILRSS (Institute of Higher Learning, Research in Specialized Studies) in Computer Science in the college During Cond-2019, Dr. Historich as uploaded more than 225 videos on Teaching Aide

and other Utility videos for Students. Faculty Academidian, and Researcher, on his YouTube channel. More than 1.74 lack Viewers viewed his videos across 38 countries. So far, 54 research scholars have been owerried a Ph.D. under his guidance and bit scholars have been awarded M.Phil. His has been profiled with 65 Patients including Australian Patients. He has published 55 papers in a reputed Journal including Scoous and published 57 papers in various National and International contentions.

Gr. 5. B. Kitcher is a double Ph.D. holder in the Professional audjocts that includes Computer Science and Management, Nas written 74 books (Including 17 International Publication) and with re-edition, this figure goes to 127 books. Many of his books are included as a Text Book and Reference book in venous consets adross the Universities and Autonomous colleges. Dr. Kishor has conducted more than 75 sessions as a resource person in venous. USC-HEDC Academic Staff College, Ph.D. Course Work at University center, and TeachingLearning Center, etc. He has trained more than Nine Thousand Ph.D. Research Scholers and Bookly members on the topic of ICT, Plagaetim, etc.

As a social responsibility, Dr. Klicher has so far deniated bloed 17 times and has registered a positive or nye donation at a government medical hospital. Chardrague: He has donated 531 unique books on Computer Science and Nutrivisional books to Late Ramadew Omkarrath Sharma, Library, He has published 54 anticles in the Hawada mempaper (A. Gestral India largest Cerculation daily English Newspaper), mostly on Research spice for making assertess of Research and a second many awards that include sorte of the prostigious like. Total the has not far Award-2012 and Basel Winter America 2011 by RTMNJ, hageur. A holder of multiplote of Research 2014, for writing Maximum Books in Computer Science and Residents of Takes and Award-2018 of Gordeana University. Galobiana and "Dastroguest HOD-2011", of Computer Society of India Dook and "Britishistica" and 2018. Neglect "Dastroguest HOD-2011", of Computer Society of India. New and 2018 of Gordeana University. Galobiana and "Dastroguest HOD-2011", of Computer Society of India. New and 2018" of Gordeana University. Galobiana and "Dastroguest HOD-2011", of Computer Society of India. New and 2018" of Gordeana University. Galobiana and "Dastroguest HOD-2011", of Computer Society of India. New and 2018" of Issael and Taketo Hotel. Neglect "Dastroguest HOD-2011", of Computer Society of India. Neurola and "Bhitahalistatur-2019" of Zero Nike. Neglect Taketo Ferreror, Start Society of India. Neuron GOVID-19" of IESA. "Excellance Teacher Award 2011" from SSM. Mandells Chandrigue. Mathetic Award "Bhitahalistature-2019" of 2010 Excellance Teacher Award 2011" from SSM. Mandells Chandrigue. Mathetic Award The India Books Teacher and patient of the chancels on With Son and guicements for Society of India Taketo form IIT Spoken Tutorial, Mumbel in 2022, Ex. Vice-Chancels on With Son and guicements for society for ordination of the "forme examination of Interestly and tainely, IT Mumbe part in appreciation letter for his conduction to IIT Spoken Tutorial avameness and cooduc

> Das Ganu Prakashan Pratibha Book Distributors Above Maratha Mandir, Rani Zanat Square, Sitatudo, Nagour - 440 012 Mo. 9622201952

CYBER SECURITY & LAW

applements) at

Dr. Madhav M. Bokare Dr. S. B. Kishor

DAS GANU PRAKASHAN, NAGPUR (INDIA)

DATABASE MANAGEMENT SYSTEM & MS-ACCESS

Database Management System & MS-Access O Dr. S. B. Kishor

🕦 🛃 Das Gann Prakashan, Nagpur



Dr. S. EL Klatere, Hand of Computer Science Dayl, working since Sap 4, 2006, In Senter Paire Hadrondynings, Charodropor, He was a Charosen of Computer Science Reard, Constrained University, Cactobrick, and a member of Academic Council Member, D. Kolstein in unrefered as a member of the Computer Science/Application based to Fire MAC A and interm Grade Universities and Autoromous colleges. No is sharing his kines and separate to a many separaterial committees that include KOAC, NAAC, Research, etc. He is the Table of BRIESS (Includes of Higher Learning, Research in Specialized Statise) in Computer Science in the college.

Computer Science in the college. During Covid-2019, Dr Kishor has uploaded more than 225 videos on Teaching Aids and other Utility videos for Stadents, Faculty, Academician, and Researchie, on his You Tube channel. Bare than 1.74 lash Vanuers viewed his videos scross 36 countries, So far, 14 research scholars have been meaded a Ph.D. ander his glasses and bit scholars have been metaded Bit Phil. He has been credited with 05 Patents including Australian Patents. He has published to papers in a regulated Journal Including Scopus and published Sy papers in warrows Material and international conferences. Dr. S. B. Kinston is a double Ph.D. holder in the Professional subjects that international Publication) and with re-edition, this figure goes to 126 books. Many of his books are included as a Text Book and Hamagement, has writen 74 books. Many of his books are included as a Text Book and Hamagement bas figure goes to 126 books. Many of his books are included as a Text Book and Hamagement.

Universities and Automoreous colleges. Dr. Klaber has conducted more than 75 sessions as a resource person in various UGC-IRTDC Academic Staff College, Ph.D. Course Work at University conter, and traching Learning Center, etc. He has trained more than Nine thousand Ph.D. Research Scholars and faculty members on the topic of RCT, Pleglarian, etc. As a social responsibility, Dr. Kisher has so far denoted blood 17 times and has registered a posthumous eye donation at a government medical hospital, Chandraphy, He has donated SAT unique becks on Computer Science and Motivatorial books to Late Remadered Omkarnath Sharma, Library, He has published S4 articles in the Historida newspaper (A Central India largest Circulation daily English Newspaper), meetly on Research topics for making awareness of Research and get termong the reader.

Research topics for making avareness of Research among the reader. Most Important, in honor his dedication and determination for his work, Dr. S. B. Kishot has been amonded many awards that include some of the prestigious like Tokal Teacher Award-2012' and 'Best Writer Award-2011' by RTMINU, Nappur, Aholder of 'India Book of Records-2014' for writing Machnum Docks in Computer Science and Receivers of 'Ideal Teacher Award-2016' of Gondwara University, Gadchirobi and 'Dettinguish HOO 2017', of Computer Society of India, Number and 'ShikshaRatan-2019' of Zero Nile Nagpur, "Rest Faculty Development for Online Teaching During COVID-15' of USA Nagpur, "Rest Faculty Development for SSM Nanda's Chandrapor, MASTER Award from W Develop Totols! Mumbal in 2022' IT Spoken Tutorial, Mumbai in 2022.

Ex. Vice-Chancellor, Prof. Dr. Srielwasa Varkhedi of Gondwana University, Gadchiroll aant him a letter of appreciation for his valuable contribution and guidance for successful conduction of the 1⁻ online examination of university and similarly, IT Numbel sent an appreciation letter for his contribution to IIT Spoken Tatorial awareness and conducting a program.



ISBN 978-81-964733-0-3

Das Ganu Prakashan Pratibha Book Distributors

Above Maratha Mandir, Rani Zanti Square, Sitabuldi, Nagpur - 440 012 Mo. 9822201957

Database **Management System** & MS-Access

Dr. S. B. Kishor

DAS GANU PRAKASHAN, NAGPUR (INDIA)

FRONT END DEVELOPMENT USING VISUAL BASIC

About Author



MASTER Dr. S. B. Kisher, Head of Computer Science Dept. working since Sep 4, 2000, in Sandar Patel Mahavidyalaya, Chandrupur, Maharashera, India, He was a Chairman of the Computer Science Buard, at Goodwana University, Gadehiroli, He is a double Ph.D. holder in Built at Gondwarra University, Galchiroli, He is a dealate PALO, noncor in Professional subjects that includes Computer Science and Management. So far, he has been credited with 05 Patents including Australian Patents. 14 researchers have been awarded a Ph.D. under his guidance and 05 scholars have been awarded M.Phil. He has written 73 books (including 17 International publications), and with re-edition, this figure goes to 125 books. Many of his books are included as a Text Book and Reference books in various courses and universities. He has publiced 116 papers in a reported loweral including Science and Australian Universities and an including science and an environment of the papers in eral including Scopes and various national and international

tes having good Citations. He has been a resource person more than 125 times at many bacmatismal Conferences and UGC-HRDC Academic Staff College, Ph.D. Course Work, Learning Contex sie. And minor more than Fearteen theseand.

canning Center, etc. And instead more than Femriteen theoreane. are 2013, under his guidance. 31 students/acholars (UG, PG, PPG Group) represented the University, Gudehrolis in Science and Technology faculty at State level Avishkar-ing Obthershins State Level fatter-University Research. Convention. And Sime Covid-2019, websided more than 240 videos on teaching ands for faculty. Academics, Research, and other cas on his YouTube channel. More than 2.51 laids Viewers viewed his videos across 40.

social responsibility. Dr. Kishar has so far diseated blood 17 times and registered a ye donation at a government endical hospital. Chandrapur. He has published 55 articles in a newspager (Central Induite Ingent circulation daily English Newspaper), mostly on its to make antereness of Research arrong the reader. He has donated 531 unique books on inverse and Metivational books to Law Ramadevi Omkarnah Sharma Library.

at importantly, to honor his dedication and determination for his work. Dr. Rishor has been by avants including some prestigious ones like the "ident Teacher Award – 2012 and "Best rel – 2013" from RTM Napper University, Napper. A holder of "Iedla Book of Records iting the Maximum Books in Computer Science and Recipient of the "Iedla Book of Records indexes Lawrence, Gadehiroli and "Bistingsish BOD-2017", of Computer Society of India, "Shilash Rates – 2019" of Zero Mile, Napper, "Best Faculty Development for Online wring COVID-19" of IESA, "Excellence Teacher Award - 2021" from SSM Mandal's "MASTER Award – 2023" by UT, Bornhay.

First, Dr. Schultus Variabedi, during his senare as a Vice-Chanteller of Gondwana rely in Cadebiral, sent him a letter of appreciation for his indepictful draft and contribution to received administration of the antiversity's first online examination in 2020. If T Mumban also investigate administration to the sentence appreciation formers, Since the year 2012, under his guidance, more than 4500 analouts and 500 analouts analouts and 500 analouts analouts and 500 analouts analouts and 500 an

In 2020, He has trained Management Council Members, Senate Members, and Academic Council members of Couds and University Godehiroli, for the online voting system, the first of its kind in entire holis for the Online selection of a member in the Panel for Vice-Chancellor.



VISUAL BASIC PROGRAMMING 🗢 Dr. S. B. Kister 🔹 🔂 Das Gamu Frakasban, Negnur

Event Driven rogramming Front End Development Using **ISUAL** BASIC Programming With Gaming Application

Dr. S. B. Kishor



pg Das Ganu Prakashan, Nagpur

Information & Communication Technology

Information & Communication Technology

•

Dr. S. B. Kishor

Das Ganu Prakashan.



Dr. S. B. Kishor, Head of Computer Science Dept. working since Sep 4, 2000, in Point Bisharridyskeys, Charaktepur. He was a Chairman of, Computer Science Council Newber, Dr. Gendennes University, Calcheroli, and a member of Academic Council Newber, Dr. a working as a weeter of the Computer Science/Application board to Five NAAC showe Gaste Universities and Automonous scilleges. He is Sharing he ideas and a or many important committees that include KAAC, NAAC, Research, etc. He is a diff.RSS (Institute of Higher Learning, Research in Specialized Studies) in a science in the college.

Computer Science in the college. During Covid-3019, Dr. Kishor has upleaded more than 225 videos on Teaching Auto and other Unity videos for Stockards, Faculty, Academics, and Hasearcher, on his for Tole channel. How than 1.74 bath Venezs viewed his videos across 30 countries. So the, 14 research scheder, have been awarded a Ph.D. under his guidance and 05 scholars have been awarded M.Phil. He has been credited with 65 Patents including Australian Patents. He has publicated to papers in a reputed Journal including Scopes and published S7 papers in waters. National and International continuous. Dr. 55, B. Wisher is a cloude Ph.D. holder in the Professional subjects that includes Computer Science and Management, has written 74 books (including 17 international Academic Tools and with meetition, this Spare goes to 127 books. Many of his lends and included at a fact Scok and Reference book in warkees course, across the Uniternation and Indenness colleges.

Dr. Ministry has constructed more than 75 sessions as a recourse person in various MICA is always. Staff College, Ph.D. Course Work at University certain, and Toaching leng Certain, etc. He has trained more than Nine Micasand Ph.D. Research Scholars will senadow a use the legist of KT, Plagaeiran, atc.

and a social response of it. 1, Polytement, etc. a social responsibility, Dr. Kilder has so be denoted blood 17 times and has profilements eve documents at a generative medical hospital, Charolingtor, He (ST) unique books on Computer Schence and Methodomic books to Late behavior for Starma, Litzwy, He has published 34 articles in the Hillerate (A Constant tooks target), and Computer Schence and public tooks to the pice for making assertaints of Research among the reader.

In the second parameters of Research among the reader.
Ind Important, to honor his dedication and determination for his work, Dr. S. B. based associated many awards that include poole of the prestigious like ideal word. 2012 and Tiost White Awards 2013 by R 1MUU. Naguer: A holder of "India Conde 2014" for writing Maximum Books in Computer Science and Recipients the Award 2018 " of Condenant University, Galachiroli and Tostinguish HOD Computer Science and Recipients to Computer Science and Recipients to the Recipient Science and Recipients (Science and Recipients Books in Computer Science and Recipients HOD Computer Science and Recipients HOD Computer Science (Science Science) for Office Tosticiting COND-10" of JEBA 1 Test Netword 2021 Three SSM Manda's Chandraper, MASTER Award from Interfal, Woodeal in 2022.

New Indextal, Hundras et clear. EX. Vice-Chancellor, Prof. Dr. Scientwara Varkhedi of Gonderena University, traff user bies a letter of appreciation for his valuable confollotion and guidance for safet conductive of the 1° online examination of university and similarity. If Mambail at appreciation letter for his contribution to 8T Spoken Televial serametes and ad a bundle was

ISBN 978-81-964733-2-7



Das Ganu Prakashan Pratibha Book Distributors Above Marstta Mantir, Rani Zanei Squara, Silabuidi, Nagpur - 440 012 Ma. 9822201952

INFORMATION COMMUNICATION TECHNOLOGY

Dr. S. B. Kishor

DAS GANU PRAKASHAN, NAGPUR (INDIA)

DG

Research and Publication Ethics

About Bool

s is the "Open Access" and which products free er prise tang ann r work for walk by changed by unservice

y USC for Ph.D.







Das Ganu Prakashan Pratibha Book Distributors Adaptive Secrets, Plant Zierre Stations A.R. Naggest 440 012 Mail: MT22201100

Price Rs. 300/-

SBN 978-83-84338-71-4

Cover the Complete Syllabus prescribed by UGC for Ph.D. Course work & including MCQ and Practical List

RESEARCH AND PUBLICATION

Dr. S. B. Kishor Dr. Ajay S. Kushwaha Dr. Gitanjali J.

2nd Edition



Programming in JAVA



About Authors

Dr. S. B. Kishor has been working as Head of, the Department of Computer Sciences since Sep. 4, 2000 in Sercar Patal Mahavidiyalaya, Chandrapur, He is a Chalman of, the Computer Science Board, a' Condwand University, Gadehiroli, and a member of the Academic Council Nember, Se fir, he has been credited with five pretents, Six M. Phil & 14 research acholars have been awarded a Ph.D. under his guidance. Dr. Kishor a double Ph.D. holder has discipline of Computer Science and Management. He has conducted more than 125 sessions in UCC-HRDCASC & others and guided more than 13000 "Bast Teecher Award-2012" A "Bast Writer Award-2013" by RTMNU, Nagpur, and the deal Teacher Award-2018" of Gondwana University, Gadehiroli. Dr. Raiani D. Sinoh here here



Dr. Radani D. Singh has been awarded a doctorate in Computer Science at R.T.M. Nagpur University, Nagpur, She has more than 13 years of Teaching Experience and She is currently working as an Assistant Professor in the Department of Computer Studies and Research, Serder Patel Mahavidyalaya, Chandrapur, Earlier she was associated with Shivaji Science College, Nagpur in the MCA Department and worked in the Computer Science Department of the institute of Science College, Negpur as well. Dr. Rajani Singh worked as an Organizer and Head of various Programs and Committees in the College. She has presented and published several papers in National and International Journals and Conferences.



She has proficiency in Web Technology, Deta Security Java, and Database Management and is a co-sultior of more than 10 popular Computer textbooks that Include Web Technology, E-Commerce and Web Designing, Nedia Management, 'Oxacle' Practical Guide, and Programming in Java. She has been swended 'Charles Babbage Young Women Beat Writer Award-2016' by Nathatma Fule Talent, Research Academy for her overall contribution to academic activities.

Miss. Goldy Suresh Thadwani worked in the Department of Computer Studies and Research at Sandar Patel Mahavidyalaya, Chandrapur, She holds a Master's Degree in Computer Application. Her area of Interest includes ODP Technology and .NET Framework. She is a keen observer and a

Miss. Mamta M. Rathod worked in the Department of Computer Science at Sardar Patel Mahavidyalaya, Chandrapur, She holds a Master's degree in Computer Science. During her PG. She represented Computer University, Gadchirell at the Inter-State level Research Competition Avishikar twice under the category "Engineering & Technology. Her area of Interest includes Web Designing, Data Structures, and OCP5 Technology

ISBN 978-93-84336-49-3





Price : Rs. 300/-

PROGRAMMING IN JAWA O Dr. S. B. Kater, Dr. Rajavi Sirgh, Mas. Soldy Thadwark Mes. Hands Rott of 🗨 🛃 Las Gazzi Philadeau

-

Programming in

Das Ganu Prakashan, Nageur

Dr. S. B. Kishor Dr. Rajani D. Singh Miss. Goldy Suresh Thadwani Miss. Mamta M. Rathod

Chapter in eBook

Dr. S. B. Kishor, "Fostering Ethical AI Adoption for Education" under ebook AI for Everyone: Applications published by Dnyanopasak Shikshan Mandal's, Arts, Commerce & Science College, Jintur ISBN: 978-81-957387-48 DOI https://zenodo.org/record/8286729 Aug. 2023 ISBN: 978-81-957387-4-8

Mohan, B. K. Madhavi, S. B. Kishor, "Detection of UDP SYN Flood DDoS Attack Using Random Forest Machine Learning Algorithm in a Simulated Software Defined Network" published in International Publication Springer eBook 2023, Page 513-523 Aug. 2023 <u>https://link.springer.com/chapter/10.1007/978-981-99-3982-4_44</u>, <u>https://doi.org/10.1007/978-981-99-3982-4_44</u> ISSN 2190-3026 (electronic)